

Introduction to Al Techniques for Renewable Energy Systems

Edited by Suman Lata Tripathi Mithilesh Kumar Dubey Vinay Rishiwal Sanjeevikumar Padmanaban



Introduction to AI Techniques for Renewable Energy Systems



Introduction to AI Techniques for Renewable Energy Systems

Edited by

Suman Lata Tripathi Mithilesh Kumar Dubey Vinay Rishiwal Sanjeevikumar Padmanaban



CRC Press is an imprint of the Taylor & Francis Group, an **informa** business First edition published 2021 by CRC Press 6000 Broken Sound Parkway NW, Suite 300, Boca Raton, FL 33487-2742

and by CRC Press 2 Park Square, Milton Park, Abingdon, Oxon, OX14 4RN

© 2021 selection and editorial matter, Suman Lata Tripathi, Mithilesh Kumar Dubey, Vinay Rishiwal and Sanjeevikumar Padmanaban; individual chapters, the contributors

CRC Press is an imprint of Taylor & Francis Group, LLC

The right of Suman Lata Tripathi, Mithilesh Kumar Dubey, Vinay Rishiwal and Sanjeevikumar Padmanaban to be identified as the authors of the editorial material, and of the authors for their individual chapters, has been asserted in accordance with sections 77 and 78 of the Copyright, Designs and Patents Act 1988.

Reasonable efforts have been made to publish reliable data and information, but the author and publisher cannot assume responsibility for the validity of all materials or the consequences of their use. The authors and publishers have attempted to trace the copyright holders of all material reproduced in this publication and apologize to copyright holders if permission to publish in this form has not been obtained. If any copyright material has not been acknowledged please write and let us know so we may rectify in any future reprint.

Except as permitted under U.S. Copyright Law, no part of this book may be reprinted, reproduced, transmitted, or utilized in any form by any electronic, mechanical, or other means, now known or hereafter invented, including photocopying, microfilming, and recording, or in any information storage or retrieval system, without written permission from the publishers.

For permission to photocopy or use material electronically from this work, access www.copyright.com or contact the Copyright Clearance Center, Inc. (CCC), 222 Rosewood Drive, Danvers, MA 01923, 978-750-8400. For works that are not available on CCC please contact mpkbookspermissions@tandf.co.uk

Trademark Notice: Product or corporate names may be trademarks or registered trademarks and are used only for identification and explanation without intent to infringe.

ISBN: 978-0-367-61092-0 (hbk) ISBN: 978-0-367-61167-5 (pbk) ISBN: 978-1-003-10444-5 (ebk)

Typeset in Times by KnowledgeWorks Global Ltd.

Contents

Preface About the E	ditors	ix xi
Chapter 1	Artificial Intelligence: A New Era in Renewable Energy Systems	1
	Kandra Prameela, Challa Lahari, Grandhi Sai Kishore, Kandula Venkata Nikhil, and Pavuluri Hemanth	
Chapter 2	Role of AI in Renewable Energy Management	15
	Anupama Sharma, Sanjeev Kumar Prasad, and Rashmi Chaudhary	
Chapter 3	AI-Based Renewable Energy with Emerging Applications: Issues and Challenges	29
	Omkar Singh, Mano Yadav, Preeti Yadav, and Vinay Rishiwal	
Chapter 4	Foundations of Machine Learning	43
	Neeta Nathani and Abhishek Singh	
Chapter 5	Introduction of AI Techniques and Approaches	65
	Namrata Dhanda and Rajat Verma	
Chapter 6	A Comprehensive Overview of Hybrid Renewable Energy Systems	85
	Amit Kumer Podder, Muhammed Zubair Rahman, Sujon Mia, and S M Fuad Hossain Fahim	
Chapter 7	Dynamic Modeling and Performance Analysis of Switched- Mode Controller for Hybrid Energy Systems	107
	Linnet Jaya Savarimuthu and Kirubakaran Victor	
Chapter 8	Artificial Intelligence and Machine Learning Methods for Renewable Energy	123
	Sushila Palwe and Prerna Lahane	

Chapter 9	Artificial Neural Network-Based Power Optimizer for Solar Photovoltaic System: An Integrated Approach with Genetic Algorithm	137
	Revathy Subbiah Rajaram and Kirubakaran Victor	
Chapter 10	Predictive Maintenance: AI Behind Equipment Failure Prediction	155
	S. Sharanya, Revathi Venkataraman, and G. Murali	
Chapter 11	AI Techniques for the Challenges in Smart Energy Systems	171
	S. Dwivedi	
Chapter 12	Energy Efficiency	185
	Har Lal Singh, Sarita Khaturia, and Mamta Chahar	
Chapter 13	Renewable (Bio-Based) Energy from Natural Resources (Plant Biomass Matters)	201
	Rajesh K. Srivastava	
Chapter 14	Evolving Trends for Smart Grid Using Artificial Intelligent Techniques	215
	Pooja Yadav, Prakhar Chaudhary, and Hemant Yadav	
Chapter 15	Introduction to AI Techniques for Photovoltaic Energy Conversion System	229
	Siddharth Joshi and Nirav Karelia	
Chapter 16	Deep Learning-Based Fault Identification of Microgrid Transformers	251
	S. Poornima	
Chapter 17	Power Quality Improvement for Grid-Integrated Renewable Energy Sources: A Comparative Analysis of UPQC Topologies	269
	Nirav Karelia, Amit San, and Vivek Pandya	
Chapter 18	AI-Based Energy-Efficient Fault Mitigation Technique for Reliability Enhancement of Wireless Sensor Network	287
	Syed Mufassir Yaseen, Mithilesh Kumar Dubey, and Majid Char	00

Contents

Chapter 19	AI Techniques Applied to Wind Energy	. 309
	Swagat Kumar Samantaray and Shasanka Sekhar Rout	
Chapter 20	Comparative Performance Analysis of Multi-Objective Metaheuristic Approaches for Parameter Identification of Three-Diode-Modeled Photovoltaic Cells	. 325
	Saumyadip Hazra and Souvik Ganguli	
Chapter 21	Artificial Intelligence Techniques in Smart Grid Irtiqa Amin and Mithilesh Kumar Dubey	. 347
Chapter 22	Parameter Identification of a New Reverse Two-Diode Model by Moth Flame Optimizer	. 367
Chapter 23	Time Series Energy Prediction and Improved Decision-Making Iram Naim and Tripti Mahara	. 383
Chapter 24	Machine Learning-Enabled Cyber Security in Smart Grids Anand Sharma, Manish Kumar, and Nitai Pal	. 397
Index		.409



Preface

Artificial intelligence (AI) approaches are being developed to produce more accurate predictions of *renewable energy* using AI techniques, including their generation and impacts on the electric grid such as net load forecasting, line loss predictions, maintaining system reliability, integrating hybrid solar and battery storage systems. The book mainly deals with AI techniques for modeling, analysis, prediction of the performance, and control of renewable energy systems. This book will provide conceptual as well as practical knowledge about AI techniques used in renewable energy systems to the students, researchers, and academicians.

CHAPTER ORGANIZATION

This book is organized into 24 chapters in total.

Chapter 1 intends to study the brief introduction and past history relevant to artificial intelligence (AI) in fields of solar, wind, and other renewable energy sources.

Chapter 2 presents a study on issues and challenges of renewable energy and AI applications to resolve these issues.

Chapter 3 demonstrates the key challenges faced in renewable energy using machine learning and various AI applications in real-life scenarios.

Chapter 4 deals with enhanced computational power and hybrid transaction or analytical processing systems (HTAPS) enabling ML algorithms to optimize the energy and also the power sector on an outsized scale.

Chapter 5 highlights the importance of machine learning in research and development, and also focuses on its types and techniques.

Chapter 6 is designed to provide the ins and outs of hybrid renewable energy systems (HRESs) such as the configuration architecture, stability issues, maintenance, available optimization techniques, and performance predicting simulation software for HRESs.

Chapter 7 focuses mainly on the integration of solar photovoltaic (PV) and electricity board (EB) source for sustaining the electric power system to satisfy the energy demand throughout the adequate insolation.

Chapter 8 describes the various AI techniques and machine learning algorithms and how these approaches have been applied to different renewable energy (RE) types, especially solar energy.

Chapter 9 demonstrates the accuracy and effectiveness of the proposed optimizer; a 400-W domestic standalone SPV system and its specifications are used as a model for simulation in MATLAB.

Chapter 10 throws light on the various AI-based failure prediction strategies that are deployed on renewable source infrastructure.

Chapter 11 deals with the application of AI techniques in smart energy systems and is an effort to concise all the information on the optimization of solar energy systems for common-life applications.

Chapter 12 gives an idea about AI-enabled energy forecasting systems to handle fluctuations that may adversely affect the planning, designing, and operations according to the requirements.

Chapter 13 focuses on biomass resources and their products with their utilization for renewable energy based on the current trend of research and development in these sectors.

Chapter 14 discusses the integration of renewable energy-based smart grid that comes up with some new challenges which enhance the research in the field of AI.

Chapter 15 deals with the modeling, simulation, and analysis of small-scale photovoltaic energy system (PVES).

Chapter 16 describes about the deep learning-based fault identification of microgrid transformers.

Chapter 17 aims to compare and analyze different configurations of UPQC, namely, UPQC-I (interline), UPQC-MC (multiconverter), UPQC-DG (distributed generation) supplying loads, which are very sensitive toward voltage variations and critical in operation.

Chapter 18 discusses concepts of reliability and maintainability theory, which are concerned with our work for calculating the maintainability.

Chapter 19 focuses on different machine learning models and it's achievements for damage estimation capability to design and develop efficient wind turbines for wind energy systems

Chapter 20 utilizes the data from the two commercial solar cells and for the purpose of the validation of results, the IV and PV curves of each cell are presented along with the other statistical analyses, which have been performed at various temperature and irradiance levels.

Chapter 21 deals with the modern wind generation system and its monitoring of health in operating condition, recognition of smart gird (SG) subsystem's fault pattern, and control of SG based on real-time simulator.

Chapter 22 presents the estimation of parameters of thin-film solar panels using the newly developed reverse two diode model (RTDM) applying moth flame optimizer (MFO). RTDM being not a regular model used for parameter estimation, increases the complexity of the equations.

Chapter 23 describes the traditional time series forecasting to achieve effective energy resource planning that ensures the availability of the desired amount of natural gas for various manufacturing operations at appropriate point of time.

Chapter 24 addresses the security issue in the smart grid due to large data processing and proposed machine learning-based cyber security solutions for the intelligent grid.

About the Editors

Suman Lata Tripathi is working as Professor, School of Electronics and Electrical Engineering, Lovely Professional University, India. She has over 17 years of experience in academics and has published over 35 research papers in refereed journals and conferences. She has organized several workshops, summer internships, and expert lectures for students. She has worked as a session chair, conference steering committee member, editorial board member, and reviewer in international/national *IEEE Journal* and conferences. She has been nominated for the "Research Excellence Award" in 2019 at Lovely professional University. She has received the best paper at IEEE ICICS-2018. Her areas of expertise include microelectronics device modeling and characterization, low power VLSI circuit design, VLSI design of testing, and advance FET design for IoT, embedded system design, and biomedical applications, etc.

Mithilesh Kumar Dubey is working as Associate Professor, School of Computer Science and engineering, Lovely Professional University, India. He has over 12 years of teaching experience and industry experience. He has worked with Tata Consultancy Services (TCS) as Software Engineer. He has filed and published three patents in the government of India in "E-health care system." He has published 21 research papers in various reputed journals like *Scopus/UGC*. He has also published one book entitled "*Core python programming*" in Lambert Germany (LAP publisher) in the year of 2018.

Vinay Rishiwal is working as Professor, Department of Computer Science and Information Technology, Faculty of Engineering and Technology, MJP Rohilkhand University, Bareilly, Uttar Pradesh, India. He received his PhD in Computer Science and Engineering from Gautam Buddha Technical University, Lucknow, India. He has 20 years of experience in academics. He is a senior member of IEEE. He has published 85+ research papers in various journals and conferences of international repute. He is a general chair of three International Conferences, namely, ICACCA, IoT-SIU, and ICAREMIT. He has received many awards for the best paper and best orator in various conferences.

Sanjeevikumar Padmanaban (Member'12–Senior Member'15, IEEE) received a Ph.D. degree in electrical engineering from the University of Bologna, Bologna, Italy 2012. He was an Associate Professor at VIT University from 2012 to 2013. In 2013, he joined the National Institute of Technology, India, as a Faculty Member. In 2014, he was invited as a Visiting Researcher at the Department of Electrical Engineering, Qatar University, Doha, Qatar, funded by the Qatar National Research Foundation (Government of Qatar). He continued his research activities with the Dublin Institute of Technology, Dublin, Ireland, in 2014.

Further, he served as an Associate Professor with the Department of Electrical and Electronics Engineering, University of Johannesburg, Johannesburg, South Africa, from 2016 to 2018. From March 2018 to February 2021, he has been a Faculty Member with the Department of Energy Technology, Aalborg University, Esbjerg, Denmark. Since March 2021, he has been with the CTIF Global Capsule (CGC) Laboratory, Department of Business Development and Technology, Aarhus University, Herning, Denmark.

S. Padmanaban has authored over 300 scientific papers and was the recipient of the Best Paper cum Most Excellence Research Paper Award from IET-SEISCON'13, IET-CEAT'16, IEEE-EECSI'19, IEEE-CENCON'19 and five best paper awards from ETAEERE'16 sponsored Lecture Notes in Electrical Engineering, Springer book. He is a Fellow of the Institution of Engineers, India, the Institution of Electronics and Telecommunication Engineers, India, and the Institution of Engineering and Technology, U.K. He is an Editor/Associate Editor/Editorial Board for refereed journals, in particular the IEEE SYSTEMS JOURNAL, IEEE Transaction on Industry Applications, IEEE ACCESS, *IET Power Electronics, IET Electronics Letters*, and Wiley-*International Transactions on Electrical Energy Systems*, Subject Editorial Board Member—*Energy Sources*—*Energies Journal*, MDPI, and the Subject Editor for the *IET Renewable Power Generation, IET Generation, Transmission and Distribution*, and *FACETS* journal (Canada).

1 Artificial Intelligence A New Era in Renewable Energy Systems

Kandra Prameela¹, Challa Lahari², Grandhi Sai Kishore³, Kandula Venkata Nikhil³, and Pavuluri Hemanth³ ¹Department of Biotechnology, Gitam Institute of Technology, GITAM (deemed to be University), Visakhapatnam, Andhra Pradesh, India ²Department of Computer Science and Engineering, National Institute of Technology (NIT), Tadepalligudem, Andhra Pradesh, India ³Department of Computer science, Gitam Institute of Technology, GITAM (deemed to be University), Visakhapatnam, Andhra Pradesh, India

CONTENTS

1.1	Introduction	1
1.2	Historical Perspectives of Artificial Intelligence	2
1.3	Applications of AI	4
1.4	Applications of AI in Wind Renewable Energy Systems	5
1.5	Modeling and Simulation Studies of AI in Wind Renewable Energy System	6
1.6	Applications of AI in Solar Renewable Energy Systems	7
1.7	Modeling and Simulation of Solar Renewable Energy Systems	8
1.8	Current Trends and Future Prospects of AI	9
1.9	Conclusion	11
Refe	rences	11

1.1 INTRODUCTION

The emergence of technology has helps us to generate solutions to minimize the risk of human errors in the complex systems. Most of the technologies developed by scientists are inspired by natural system. Artificial intelligence (AI) is an interdisciplinary promising approach for prompting the human brain functions and incorporating them into the machine to make it human smart. In the modern world, science and technology is a remarkably strong venture with development of quantified self to empower the challenges faced by humans. AI is the mimic of human brain and programmed to the computer to use for several computer systems which have been

generated, for example, to analyze electric circuits, to solve differential equations, to design for synthesis of complex chemical molecules, and to detect and treat diseases. Due to AI's systems degree, it is possible to understand and build a neural network to challenges, for human benefits [1]. In general, AI includes fuzzy logic, expert systems, and artificial neural networks (ANNs). These are very powerful tools of renewable energy and smart grid systems for design, stimulation, estimation, control, fault diagnostics, and fault-tolerant control [2]. AI is a creation of a system with human brain intelligence to find solutions of narrow specific manually unsolved tasks. For example, in a supervised learning process, the system will learn the mapping based on their input and output programms.

The pervasiveness of technology allows spreading it across all fields in the world, including energy systems, health, agriculture economics, engineering, marine, and others. In every field, consumption of energy is increasing drastically and is causing decline to fossil fuels. In such situation, many countries have shifted their focus to renewable energy resources such as wind and solar energy sources. The main purpose of AI is to understand the problems of human cognition, analyze circuits, limit the human speech, train the system and store the information to retrieve it. It includes powerful tools such as fuzzy logic, expert systems, and ANNs that are used for designing, simulating, controlling, and diagnosing faults in renewable energy systems and others. Genetic algorithm is also one of the optimum AI techniques that imitates the evolution and is based on natural selection which utilizes some of the genetic actions such as selection, crossover, and variation. Similarly, data mining is another potential technique used to predict information from large databases. Most of these techniques are used in the field of production control, customer behavior, and stock prices. Data mining is also applicable in astronomical and medical data. They are complex and require precise analysis to set up global technological systems. AI programs require a clear-cut explanation since detailed reasoning can be drifted out so as to reach adequate solutions. AI research in the energy field is majorly focusing on solving several technical issues. The most promising development in AI is artificial neural techniques, which can be used for wind energy source assessment and apex power point tracking in photovoltaic (PV) generators. Expert systems have helped in building efficient fault-tolerant control in modern grid systems. Currently, many countries are working in R&D of smart grid to replace and develop advanced, efficient smart grid, as older versions are insufficient to resolve some of the faults.

1.2 HISTORICAL PERSPECTIVES OF ARTIFICIAL INTELLIGENCE

The great contribution of some researchers in the development of AI gains momentum toward the change of the world's scenario. AI concept can be traced back to the ancient time when people used to mention machine learning in their mythical stories [3]. This belief has made to found an expression in many forms of AI to explore the origin and the reasons for our existence and to create workable technologies for improvement of quality of human life. Initially, people used to mention this concept as "mythical thinking machines." Later, in 10 AD–70 AD, Heron of Alexandria had written in his famous work *Pneumatica* that these mythical thinking machines were built to exhibit some form of tricked intelligence [4]. However, during that

3

time, it was not possible to develop mechanical machines having all intelligencebased functionalities. In the 17th century, scientists worked on the development of complex machines with intelligence, where this concept was compared with animals (Descartes), which is the basis of mind, body, and soul for the treatment of animals [5]. This idea of comparing the animal body with machines and mind with the intelligence for operation of machines was the novel approach for understanding of AI [6]. There was a great furor against this concept by philosophers and mathematicians. The first consequences of mathematical logic to computers came in 1642 with the invention of the first mathematical calculator [7, 8]. It took almost 30 years to make a mechanical calculator perform functions such as division and multiplication.

The most striking attributes came at the end of the 19th century with a notable contribution from George Boole who invented Boolean algebra, which is the major component of modern computers. Almost at the end of the 19th century, Samuel Butler noted that the machines might be evolved like Darwinian evolution [9]. Samuel has correctly interrelated his concepts since nowadays, researchers and scientists are building a foundation for the early 20th century. It is important to note that in 1942 Isaac Asimov published the concepts that he contrived with allusion to his laws of responsible robotics pertaining to his belief in growing the AI [10]. He proposed three concepts with two laws. They are: robots are not harmful to humans; it obeys human orders (first law); and robots must protect its own existence (second law). Another professor from the University of Connecticut focused most of his research work on machine ethics [11]. In 1950, a conference was conducted by one of the youngest renowned computer scientists John McCarty to bring many computer scientists on a single platform to discuss the creation of machine intelligence. He is considered the "father of artificial intelligence." After 2 years, there was a drastic improvement in AI. In 1961, Marvin Minsky mentioned in his published paper the steps toward AI and he summarized much of his work during the first decade of AI [12, 13]. He was able to represent in frames what he called "society of mind." In a week-long Dartmouth Summer Research Project on AI (DSRPAI) conducted by Marvin and John McCarthy in New Hampshire, the word "artificial intelligence" was coined. The main objective of this project was to reunite scientists in different fields in order to build a novel research platform on machines that are able to simulate human intelligence. Newell and Simon's work on GPS was developed based on a psychological query and empirical methods [14]. It is important to note that contributions from the end of the 19th century help in understanding computers and their languages to store and retrieve information in a massive way. In 1977, Goldstein and Papert demonstrated and published the Dendral program [15, 16]. However, the probability of expanding AI started in the 1960s, and it took a huge leap forward in the 20th century. Initially, two laboratories were established and the burden was taken by MIT (Massachusetts Institute of Technology) and Carnegie Tech University (CMU), with the Rand Corporation establishing AI laboratories later in Stanford and Edinburg. Edinburg was the place where the first conference of AI was conducted in 1965. In 1969, International Joint Conferences on Artificial Intelligence (IJCAI) started biannual series on AI. In 1980, the American Association of Artificial Intelligence (AAAI) was established [17]. It has expanded by conducting conferences and a series of workshops, sponsoring scholarships, encouraging journals, and maintaining digital libraries.

1.3 APPLICATIONS OF AI

AI has unique applications where complicated languages allow to make quicker, better, accurate predictions than manual methods. It has many branches, such as fuzzy logic, ANN, adaptive network-based fuzzy inference system (ANFIS), and data mining. ANN is inspired by the nervous system of humans, i.e., connections between elements to establish a specific function by adjusting the connections between elements [18, 19]. Mapping multiple inputs to get a single output, pattern classification, recognition, association and generation of meaningful patterns are some of the applications of AI. AI applications have also been successfully extended to other fields such as engineering, mathematics, meteorology, medicine, neurology, economics, and psychology. One of the interesting applications of AI is the recognition of sound and speech. In 1960, the adaptive linear combiner was developed, which is a very useful law [20]. Most of the neural networks depend on algorithms, mathematical models, and nonlinear operations, which include maps of noise. AI is also applicable in market trends, weather forecasting, thermal and electrical load prediction, exploration of mineral sites, and prediction of data collected from sensors [18, 19]. Analysis of medical signatures, electromyography, and identification of explosives and military targets are other applications. One of the interesting applications of AI is in PV electricity production. The main objective of AI application is to produce viable solar emission info for real-time designing and forecasting of solar radiation and steam generators. Generating PV modules, designing solar collectors, heating controllers, sun tracking and building controlled solar air conditioning systems, predicting temperature and radiation, and developing PV-diesel and solar-wind hybrid systems are other encouraged applications [21, 22]. Another interesting domain of AI is molecular biology. AI can be applied to design and analyze scientific data, solving wide problems, hierarchical patterns, and knowledge-based maintenance and obtaining knowledge acquisition technologies. AI has momentum in the field of molecular biology for the last few years. Nutritional meal planning is another important issue for cancer patients whenever they undergo chemotherapy. Through AI, the meal components with nutrition were designed in Mary to greater variety. Now, this Mary is used as a public tool; similarly, other additional nutritional elements such as carbohydrates and cholesterol are also added to the design of the foods. AI has large applications for cancer patients such as resolving drug and dosage parameters. One of the designs for drugs and doses is quadratic phenotypic optimization platform (QPOP). This model will help in reducing the toxicity of drugs with maximum output in preclinical studies. With the QPOP platform, nearly 14 chemotherapeutic drugs were screened and are used for several cancer treatments such as multiple myeloma [23]. The combination of drugs, such as decitabine and mitomycin C, markedly shown improvement in outcomes in mouse models [24]. Designing of biomarkers is another advantage of AI [25]. The step forward in treatment of cancer patients is targeted therapies genomic alterations and treatment responses [26, 27]. Therefore, AI analysis of electronic health records and patient biomarker data may affect trial outcomes [28]. Some of the applications of AI in renewable energy with AI apps are shown in Figure 1.1.



FIGURE 1.1 Some applications of AI in renewable energy with AI apps.

1.4 APPLICATIONS OF AI IN WIND RENEWABLE ENERGY SYSTEMS

Fossil fuel consumption all over the world is increasing day by day; therefore, the need for electric energy is also increasing. Effective ways of electric power generation are inherently dependent on the world economy. One of the sources for substitute renewable energy is wind. In large geographical areas such as Mexico, it was determined that the renewable energy wind is a periodical phenomenon [29]. Countries such as Mexico have a high demand for renewable energy sources because of the increase in prices of fossil fuels and high demand to minimize environmental contaminants that fossil fuels generate [30]. For better optimal designs and operations of wind energy systems, new strategies are continuously developed by many researchers [31]. Because of low-speed winds, this system does not have the same viability in all locations and is more uncertain than solar energy systems. Preferred locations of wind farms are offshore and high-altitude sites. In 2014, Norway's contribution toward renewable energy was 98%. Brazil has contributed 73.4%, New Zealand 79%, Venezuela 62.8%, and Colombia 70%. The traditional method that is used in generating wind energy is, first, installation of windmills in the farm; the wind will rotate the turbine and due to this the kinetic energy is changed to mechanical energy, which then transformed into electric energy. The first wind turbine with a capacity of 12 kW was installed in Ohio in 1887-1888. An overall growth rate of 9.96% is observed in installing wind energy scope when correlated with the previous year's scope scope in the European Union. The annual installation increased from 48 GW in 2006 to 141 GW in 2015, where the annual growth was more than 9% [32]. From 2010 to 2015, China was leading in installing the wind power units. In 2006 and 2007, Germany has greater statistics in the installation of wind power systems. In 2008 and 2009, the United States has a greater number of installed wind power units, but in 2008, both the United States and Germany contributed similar statistics in wind power systems that were installed. The efficiency of power generation using wind turbine can be increased by gathering the past wind data at turbine location. Accurate forecasting of wind speed cannot be done but this can be predicted using the past data of the wind speed. Using data analytics and AI, there are two different approaches to prediction of wind speed.

1.5 MODELING AND SIMULATION STUDIES OF AI IN WIND RENEWABLE ENERGY SYSTEM

Most of the wind designs can be recognized with the mathematical formula V(k + l) of the wind vector, where V(k) is based on the previous *m* measurements V(k), V(k - 1), ..., V(k - m + 1).

For a better wind speed forecast, *l* is chosen to be small; therefore, it is called short-term wind speed prediction. Wang and his co-workers have produced a general procedure to produce fuzzy rule [33, 34]. The fuzzy model will predict the wind speed and these are used to produce electrical power, especially at the wind units. ANN is a favorable technology for wind speed forecasting [35]. The proposed ANN will have fewer neuron numbers and it will have accurate wind prediction results in less time in different fields such as transient detection, approximation, pattern recognition, and time series prediction. V(k), V(k-1), ..., V(k-m+1) are the measured wind speeds in time series, and based on this $V(k + l), \ldots, V(k + 2), V(k + 1)$ are predicted or estimated wind speeds, and some properties of time series such as standard deviation, slope, and average are calculated and these values are given as inputs to the fuzzy or neural network predictor. The fuzzy rule base will be reduced if we reduce the inputs to the fuzzy interface system. This will give all the speeds of wind without sacrificing the estimated wind speed. Accuracy of the prediction and the learning process speed will be increased by reducing the ANN size for a neural network. In the period 2002-2005, many investigations were carried out using real wind speed data in Rostamabad in northern Iran. For every 30-minute interval, these measured data were averaged and the data for the months of February, May, August, and November were averaged for the period 2002-2004 [36]. The data are used as train data for the neural network and measurements for the year 2005 are used as test data. The absolute error between predicted and real values is called RMSE - lesser the value, the better the wind speed estimation. On the other hand, the neural method gives very little error in both the COD and RMSE values, where the absolute change will be less the 20%. To determine the certainty, one's prediction from a certain model is done by COD where its value varies from 0 to 1, 0 for the menial speed forecast and 1 for best accurate speed prediction [36].

In order to see the improved performance of ANN method, a hybrid method is adopted which is a combination of neural and fuzzy methods. Data from Tasmania, Australia, are used to measure wind energy using the ANFIS method on a very short-term basis [37]. Even though it is a neural network, functionally, it is equivalent to the fuzzy inference model and thus it is called a neuro-fuzzy system [37]. The prediction is done by ANFIS at the height of 40 m, which has only 3% of the mean absolute percentage error when compared with the actual wind speed at the same height of 40 m. The RMSE between the predicted values of ANFIS and the actual values of wind is 0.230. In general, every input is presented by two fuzzy sets only, and sometimes three sets are also considered as common. With the increase of extra membership functions, the accuracy of the result will increase but it will take more time to train the model or the neural network. The architecture of the ANFIS consists of six layers, where each layer has a different application. Layer 1 is an input layer where it just sends the crisp external signals to layer 2. Layer 2 is called fuzzification layer where fuzzification is performed by each neuron. Layer 3 is called a rule layer where each neuron corresponds to a single fuzzy rule. Layer 4 is known as the normalization layer where normalized firing strength of the given rules is being calculated by taking the inputs from all neurons in the rule layer. Layer 5 is called a defuzzification layer. Every neuron in this layer is connected to the corresponding normalized neuron, and input of $\times 1$ and $\times 2$ is also received. Layer 6 is a single summation neuron where it sums all the defuzzification neurons and produces the output *y* for the ANFIS model [37].

1.6 APPLICATIONS OF AI IN SOLAR RENEWABLE ENERGY SYSTEMS

In today's world, climate change is one of the major threats currently faced by the people. As these challenges are considered by the government and various energy solution providers, it has become important for them to provide a sustainable mode of renewable energy. This made the renewable energy an alternate source to fossil fuels which we are currently using [38]. Due to climatic conditions and depletion of nonrenewable energy rapidly, research and development on sustainable energy have increased, and it affects all countries and enterprises [39]. When compared to other areas, solar energy applications are very expensive in remote areas since devices such as pyranometers which are used to measure solar radiation perfectly, are either present in only some locations or absent [40]. The increase in energy consumption from several decades has risen fright of using up reserves of resources in coming future [41]. With the tough year ahead, production and exhaustion of distinct types of fossil fuels to give energy constitute notable changes in environment [42]. According to the World Energy Forum prediction, the energy fuels such as oil, coal, and gas reserves will be drained in the coming 10 decades. Normally, the primary consumption of fossil fuel in the world is over 79% and of that over 57.7% is consumed by the transport and is drained rapidly [43]. India needs to improve energy equipping for population, and speedy improvement in economy poses a discouraging challenge.

1.7 MODELING AND SIMULATION OF SOLAR RENEWABLE ENERGY SYSTEMS

The solar energy application has become an interesting research topic. Advancement in AI has brought about more significant results in weather forecasting and henceforth in upgrading the renewable energy and more secure access. Solar energy is considered as the better common and famous source of sustainable energy, as the sunlight is abundant in nature and access to this is simple when compared to others [39]. The solar power plants convert the energy from sunlight by using either intensive solar power plants, known as concentrating solar power (CSP) plants, or solar PV plants [44]. The CSP plants generate electricity from solar without emitting any harmful gases. This results in not polluting the environment [39]. Coming to solar PV plants, PV effects are used in order to generate electricity [44]. The CSP plants use steam turbines in order to convert sunlight into electrical energy. These plants can also run with fossil fuels when there are cases like blocking of the sun or at night times [45]. Despite all these, there are some disadvantages in solar energy. Till now there is no best tech in order to transform the electrical energy into storage medium. But there are three technologies currently in use and up to some extent they provide a feasible storage solution. They are thermal energy, smart power unit depository, and hydrogen fuel cells [39]. These three provide a reliable storage medium for the generated renewable electricity. The PV storage systems are used in generating electricity from the solar energy. These systems are divided into on-grid, off-grid, and hybrid systems. The on-grid systems continuously fluctuate and modify the state of the battery according to alterations in the potential of the sun and the charge status. The on-grid systems consist of a group of inverters that convert DC power to AC and store it in the battery pack. On the other hand, the off-grid systems convert the DC power to AC power and supply to the systems or to battery units. The hybrid solar PV systems unite the on-grid systems with additional battery modules. These systems are more complex and very cost-effective [39]. PV systems encounter a problem that is nothing but the most appropriate resolution of their dimensions. The optimization problem was dealt with by using hybrid energy storage technology which was proposed by Li et al. [46].

The main component of the solar energy generation is sunlight. There are many methods and predictions in order to forecast the solar radiance of sun, by forecasting weather and other climatic conditions. To approximate the solar radiance, few of the procedures are utilized but they are efficient when there is less spatial resolution than complex ones such as mountainous areas [47]. Due to this an artificial-based technique using ANN was put forward by Bosch et al. [48] that predicts the solar radiance even on terrains. Several other algorithms to predict the solar radiance include ANN and neuro-fuzzy inference systems [21]. ANN is used for identification of optimal operating point of PV systems [49]. To teach the ANN controller in order to identify the maximal operating point, a gradient sinking algorithm is used [19]. Using ANN, the solar radiation was predicted where there is no coverage of direct measurement instrumentation [40]. This prediction has a 93% accuracy of solar radiation with a mean absolute percentage error of 7.3%. Another method, back-propagation neural network (BPNN), is also used for forecasting the solar radiation. In the years

1988–2002, global solar radiations were predicted using BPNN; for the year 2002, the RMSE value obtained by using this method was 2.823×10^4 [50]. Using BPNN, daily solar radiation was estimated with an RMSE of 5.0–7.5% [48].

Predictions of solar radiation using ANN models with distant back-propagation algorithms were carried out by Premalatha and Valan Arasu [51]. In this, they considered two ANN models, and nine inputs were given to ANN model as trained data. The first model considered four stations, and the second model consisted of five stations in India. The two models are tested using the trained data and the algorithms. Among the two models, the model with five stations (model-2) depicted greater accuracy when calculated with LM (Levenberg-Marquardt Back Propagation) algorithm since it has shown lesser RMSE with a value of 1.0416 and a maximum linear correlation (R) of 0.9545. ANFIS is a hybrid method that combines both neuro-fuzzy and fuzzy logic. It is used to suspect clarity index and daily solar emission forecasting with RMSE 0.0215–0.0235, modeling PV power supply with 98% accuracy [52]. Therefore, by using AI we can predict the factors required to generate solar energy accurately with least error, which reduces the operational cost and human error. By using solar energy, the emission of greenhouse gases can be decreased, which encourages us to produce more renewable energy. It has very less impact on producing carbon dioxide and global warming emissions. In India, it is expected that nearly 60,000 MW of power will be produced from renewable energy sources during 2031–2032. The investment required to build a solar power plant is very high. It requires more land for installing more panels. It works only during daytime (i.e., in sunlight) and thus the energy cannot be collected during the night. In order to process the equipment using solar energy at night, the energy needs to be stored in batteries and supplied to the components.

1.8 CURRENT TRENDS AND FUTURE PROSPECTS OF AI

Currently, AI is present in many areas and is much developed in the area of renewable energy. AI is being used in predicting, designing, controlling, monitoring, and optimizing renewable energy models. Compared to the first quarter of 2019, this year the global usage of renewable energy has risen to 1.5%. After the completion of more than 100 GW of solar PV and 60 GW of wind power projects in 2019, the renewable energy generation has increased to 3% [53]. As per the renewable energy policy network (REN), at least 10 GW of energy each was produced from 32 countries globally in 2019. For most of the countries, producing energy from solar and wind is cost effective than from other sources such as coal and thermal [54]. India is producing 75 W of electricity from solar and wind energy. It may increase up to 175 W by 2022 and to 500 W by 2030 [55]. Physical methods are also used for predicting the renewable energy. These methods are efficient in forecasting atmospheric movements but they require large computation resources since a lot of data are required to assess. Physical methods are suitable only for short-term forecasting [56]. The main aim in wind energy is to predict the wind power ahead and we can plan accordingly with the predicted data. ANN methods are considered as the best estimation models in to predict wind power. The major drawback in ANN methods is that their performance varies with the location. For example, BPNN methods may provide an accurate estimation of wind power in one site; in another site, the radial basis function neural network (RBFNN) can be used as the best estimation model [57. Fuzzy logic can also be used in predicting, but the major problem in this is that it should generate a large number of fuzzy rule bases. In order to generate less rule base for fuzzy logic, a proposed model is defined based on ANN and fuzzy logic, which helps in generating less rule [57]. By using some algorithms, we can decide which places are suitable to build wind farms; this may result in less operating costs [58]. The numerical weather prediction method is efficient in short-term forecasting, while ANN and statistical methods are best in very short-term forecasting [37]. Hybrid model ANFIS is also the best model in estimating the very short-term forecasting but the training of ANFIS may fail if the set consists of insufficient variation to properly model the characteristic data [37]. Germany has developed a deep learning neural network model which helps in saving the lives of birds by not being killed by the wind turbine blades. A camera is attached near the turbine and predicts the movement using image processing, so when a bird approaches the blades, the turbine is automatically switched off [59]. Although there are models to forecast the wind speed such as physical, conventional statistical, and spatial correlation models, there are some disadvantages with these models, e.g., a statistical model cannot forecast high noise, fluctuation, and irregular nonlinear trends, spatial correlational model finds it difficult to predict the wind speed since the model needs more information like wind speed values of spatial correlated sites, etc. [60]. Now, considering the solar energy, the main usage of AI is in predicting the solar radiation. ANN in line with solar energy is used for monitoring the thermal ratings and rise of temperature of overhead power lines, depending on the meteorological conditions. It is also helpful in determining the solar irradiance [19]. The hybrid model ANFIS is used for determining the frequencies from the temperature as well as the duration of sunshine [61] and the modeling of the PV power supply system with 98% accuracy [57]. BPNN is used to measure the global solar radiation. Forecasting solar radiation by BPNN consists of higher value coefficients [57]. Another method that is used for predicting the short-term solar power is the support vector machine (SVM). Compared to other methods such as RBFNN, the SVM methods perform better, having a mean absolute error of 62 W/m² [57]. The solar radiation is higher in India when compared to countries like Germany, where the solar radiation ranges from 800 kWh/m² to 1200 kWh/m² [62]. Some of the hybrid AI methods are used in solar energy systems, e.g., in estimating the PV power using the hybrid evolutionary optimization of ANN with particle swarm optimization and genetic algorithm [63]. Genetic swarm optimization of BPNN [64] predicts the solar radiation using a combination of autoregressive and moving average (ARMA) and time delay neural network [65]. By using AI, we can improve safety, integration of micro grids, etc., in solar energy. The world is running on fossil fuels currently and for sure they are running out and none will be left over after. This may lead the world to begin changing its course, relying on the renewable energy completely. This takes much more advancement in the area of AI being used in the renewable energy. Everything currently is being automated and we may not be surprised if the world is completely automated in the near future. All the renewable energy monitoring may be automated and checked by the machines. New methods and algorithms may predict all the conditions perfectly without any errors and make human lives easier and the environment less polluted. An estimation given by the German Advisory Council on global change states that 50% of energy will be accounted for renewable energy by 2050 [66]. The International Renewable Energy Agency suggests that renewable energy can account for 60% or more of many countries' total final energy consumption and all countries may consider the use of renewable energy in their total energy use by 2050. Hence, progress in conventional applications of AI provides the opportunities to design and develop new mechanisms for specific applications in the near future.

1.9 CONCLUSION

AI is an emerging technology in renewable energy systems. Many researchers have made efforts to integrate the applications by innovative approaches for a sustainable environment. Wind and solar energy have enabled us to assess the potentiality through neural-based approaches. Computer science and analytical techniques in the field of biology have been applied to understand basic machine learning to application challenges in medicine and agriculture. In order to advance this technology, multiple strategies were used to pace the progress for human challenges. Hence, this chapter focused on applications of AI in the fields of solar, wind, and other renewable energy sources. Moreover, it emphasizes the concepts of different AI technologies with few examples such as solar and wind renewable energy systems. It deals with a major breakthrough in acceptance, modeling, and simulation studies with current trends and future prospects of AI technologies. Hence, progress in conventional applications of AI provides opportunities to design and develop new mechanisms for specific applications in the near future.

REFERENCES

- Nilsson N. (1980). Principles of Artificial Intelligence. Springer-Verlag, New York, NY. ISBN 3-540-11340-1.
- Bose K. B. (2017). Artificial intelligence techniques in smart grid and renewable energy systems—Some example applications. Proceedings of the IEEE, 105(11), 2263–2273.
- 3. McCorduck P. (2004). Machines Who Think. https://monoskop.org/images/1/1e/ McCorduck_Pamela_Machines_Who_Think_2nd_ed.pdf
- 4. Heron of Alexandria. Encyclopedia Britannica. Encyclopedia Britannica, 1-24-017. https://www.britannica.com/biography/Heron-of-Alexandria
- 5. Cottingham J. (1978). 'A Brute to the Brutes?': Descartes' treatment of animals. Philosophy, 53(206), 551–559.
- Laland K. N. & Hoppitt W. (2003). Do animals have culture? Evolutionary Anthropology, 12(3), 150–159.
- Karwatka P. & Dennis M. (2004). Blaise Pascal and the first calculator. Tech Directions, 64(4), 10–15. 1p. 2 Black and White Photographs. https://search.proquest.com/openview/bc3f2d92bff9d978494763814084ad0c/1.pdf?pq-origsite=gscholar&cbl=182
- 8. Solla Price D. D. (1984). A history of calculating machines. IEEE Micro, 4(1), 22–52.
- 9. Samuel B. (1873). The Book of the Machines: Chapters in *Erewhon*. The Floating Press. https://oheg.org/hyr.pdf
- 10. Asimov I. (2004). I Robot. (1). Spectra, New York, NY.
- 11. Anderson S. L. (2008). Asimov's "three laws of robotics" and machine metaethics. AI & Society, 22(4), 477–493.

- 12. Minsky M. (1968). Semantic Information Processing. MIT Press, Cambridge, MA.
- Minsky M. (1961). Steps toward artificial intelligence. Proceedings of the Institute of Radio Engineers, 49(1), 8–30.
- 14. Newell A. & Simon H. (1972). Human Problem Solving. Prentice-Hall, Englewood Cliffs, NJ.
- 15. Goldstein I. & Papert S. (1977). Artificial intelligence, language and the study of knowledge. Cognitive Science, 1(1), 84–123.
- Lindsay R. K., Buchanan B. G., Feigenbaum E. A., & Lederberg J. (1980). Applications of Artificial Intelligence for Chemical Inference: The DENDRAL Project. McGraw-Hill, New York, NY.
- 17. American Association for Artificial Intelligence (AAAI). (2005). Menlo Park, CA. www.aaai.org/aitopics/history
- Kalogirou S. A. (2000). Applications of artificial neural-networks for energy systems. Applied Energy, 67(1–2), 17–35.
- 19. Kalogirou S. A. (2001). Artificial neural networks in renewable energy systems applications: A review. Renewable and Sustainable Energy Reviews, 5(4), 373–401.
- Chien J. T. (2005). Predictive hidden Markov model selection for speech recognition. IEEE Transaction on Speech and Audio Processing, 13(3), 377–387.
- Mellit A., Kalogirou S. A., Shaari S., Salhi H., & Hadj Arab A. (2008). Methodology for predicting sequences of mean monthly clearness index and daily solar radiation data in remote areas: Application for sizing a stand-alone PV system. Renewable Energy, 33(7), 1570–1590.
- 22. Mellit A. & Pavan A. M. (2010). A 24-h forecast of solar irradiance using artificial neural network: Application for performance prediction of a grid-connected PV plant at Trieste, Italy. Solar Energy, 84(5), 807–821.
- Palmer A. C. & Sorger P. K. (2017). Combination cancer therapy can confer benefit via patient-to-patient variability without drug additivity or synergy. Cell, 171(7), 1678–1691.
- Rashid M. B. M. A. et al. (2018). Optimizing drug combinations against multiple myeloma using a quadratic phenotypic optimization platform. Science Translation Medicine, 10(453), eaan0941.
- 25. Lin A. et al. (2019). Off-target toxicity is a common mechanism of action of cancer drugs undergoing clinical trials. Science Translation Medicine, 11(509), eaaw8412.
- Wong C. H., Kein W. S., & Andrew W. L. (2019). Estimation of clinical trial success rates and related parameters. Biostatistics, 20(2), 273–286.
- Harrer S., Shah P., Antony B., & Hu J. (2019). Artificial intelligence for clinical trial design. Trends in Pharmacological Sciences, 40(8), 577–591.
- Rajkomar A. et al. (2018). Scalable and accurate deep learning with electronic health records. NPJ Digital Medicine, 1(18). https://doi.org/10.1038/s41746-018-0029-1
- Hernández-Escobedo Q., Manzano-Agugliaro F., Gazquez-Parra J. A., & Zapata-Sierra A. (2011). Is the wind a periodical phenomenon? The case of Mexico. Renewable and Sustainable Energy Reviews, 15(1), 721–728.
- Hernández-Escobedo Q., Manzano-Agugliaro F., & Zapata-Sierra A. (2010). The wind power of Mexico. Renewable and Sustainable Energy Reviews, 14(9), 2830–2840.
- Henriksen L. C. (2010). Wind energy literature survey no 16. Wind Energy, 13(4), 524–526.
- EWEA. (2006–2015). Wind in power European statistics. http://www.ewea.org/ statistics
- Wang L. X. & Mendel M. (1992). Generating fuzzy rules by learning from examples. IEEE Transactions on Systems, Man and Cybernetics, 22(6), 1414–1426.
- Wang L. X. (1996). A course in fuzzy systems and control. Prentice-Hall Inc, Upper Saddle River, NJ.

- 35. Lapedes A. & Farber R. (1987). Nonlinear Signal Processing Using Neural Networks: Prediction and System Modeling. Technical report LA-UR-87-2662. Los Alamos National Laboratory, Los Alamos, NM.
- Monfared M., Rastegar H., & Kojabadi H. M. (2009). A new strategy for wind speed forecasting using artificial intelligent methods. Renewable Energy, 34(3), 845–848.
- 37. Potter C. W. & Negnevitsky M. (2006). Very short-term wind forecasting for Tasmanian power generation. IEEE Transactions on Power Systems, 21(2), 965–972.
- Saurenergy.com. https://www.saurenergy.com/solar-energy-articles/the-role-of-ai-andml-in-solar-energy
- 39. Trappey A. M. C., Chen P. P. J., Trappey C. V., & Ma L. (2019). A machine learning approach for solar power technology review and patent evolution analysis. Applied Sciences, 9(7), 1478.
- AI-Alawi S. M. & AI-Hinai H. A. (1998). An ANN-based approach for predicting global radiation in locations with no direct measurement instrumentation. Renewable Energy, 14(1–4) 199–204.
- Sharma N. K., Tiwari P. K., & Sood Y. R. (2012). Solar energy in India: Strategies, policies, perspectives and future potential. Renewable and Sustainable Energy Reviews, 16(1), 933–941.
- 42. Ramedani Z., Omid M., Keyhani A., Shamshirband S., & Khoshnevisan B. (2014). Potential of radial basis function based support vector regression for global solar radiation prediction. Renewable and Sustainable Energy Reviews, 39, 1005–1011.
- Kumar A., Kumar K., Kaushik N., Sharma S., & Mishra S. (2010). Renewable energy in India: Current status and future potentials. Renewable and Sustainable Energy Reviews, 14(8), 2434–2442.
- 44. Dawn S., Tiwari P. K., Goswami A. K., & Mishra M. K. (2016). Recent developments of solar energy in India: Perspectives, strategies and future goals. Renewable and Sustainable Energy Reviews, 62, 215–235.
- 45. Philibert C. et al. (2014). Technology Roadmap: Solar Thermal Electricity. International Energy Agency, Paris, France.
- 46. Li C.-H., Zhu X.-J., Cao G.-Y., Sui S., & Hu M.-R. (2009). Dynamic modeling and sizing optimization of stand-alone photovoltaic power systems using hybrid energy storage technology. Renewable Energy, 34(3), 815–826.
- Baños R., Manzano-Agugliaro F., Montoya F. G., Gil C., Alcayde A., & Gómez J. (2010). Optimization methods applied to renewable and sustainable energy: A review. Renewable and Sustainable Energy Reviews, 15(4), 1753–1766.
- Bosch J. L., López G., & Batlles F. J. (2008). Daily solar irradiation estimation over a mountainous area using artificial neural networks. Renewable Energy, 33(7), 1622–1628.
- 49. Veerachary M. & Narri Y. (2000). ANN based peak power tracking for PV supplied DC motors. Solar Energy, 69(4), 343–350.
- Rehman S. & Mohandes M. (2008). Artificial neural network estimation of global solar radiation using air temperature and relative humidity. Energy Policy, 36(2), 571–576.
- 51. Premalatha N. & Valan Arasu A. (2016). Prediction of solar radiation for solar systems by using ANN models with different back propagation algorithms. Journal of Applied Research and Technology, 14(3), 206–214.
- 52. Mellit A. & Kalogirou A. S. (2011). ANFIS-based modelling for photovoltaic power supply system: a case study. Renewable Energy, 36(1), 250–258.
- 53. IEA.org. https://www.iea.org/reports/global-energy-review-2020/renewables
- 54. Ren21.net. https://www.ren21.net/what-are-the-current-trends-in-renewable-energy/
- CSiS.org. https://www.csis.org/analysis/optimizing-indias-electricity-grid-renewablesusing-ai-and-machine-learning-applications

- Wang H., Lei Z., Zhang X., Zhou B., & Peng J. (2019). A review of deep learning for renewable energy forecasting. Energy Conversion and Management, 198, 111799. DOI: 10.1016/j.enconman.2019.111799.
- Jhaa S. Kr., Jasmin B., Anju J., Nilesh P., & Han Z. (2017). Renewable energy: Present research and future scope of artificial intelligence. Renewable and Sustainable Energy Reviews, 77(C), 297–317.
- Zhao X., Wang C., Su J., & Wang J. (2019). Research and application based on the swarm intelligence algorithm and artificial intelligence for wind farm decision system. Renewable Energy, 134, 681–697.
- RESET.org. https://en.reset.org/blog/germany-artificial-intelligence-making-wind-turbines-more-bird-friendly-02162020
- 60. Fu T. & Wang C. (2018). A hybrid wind speed forecasting method and wind energy resource analysis based on a swarm intelligence optimization algorithm and an artificial intelligence model. Sustainability, 10(3913), 1–24.
- Mellita A. & Kalogirou A. S. (2008). Artificial intelligence techniques for photovoltaic applications: A review. Progress in Energy and Combustion Science, 34(5), 574–632.
- Pillai I. R. & Banerjee R. (2009). Renewable energy in India: Status and potential. Energy, 34(8), 970–980.
- Caputo D., Grimaccia F., Mussetta M., & Zich R. E. (2010). Photovoltaic plants predictive model by means of ANN trained by a hybrid evolutionary algorithm. In: Proceedings IEEE JCNN Barcelona, 1–6.
- 64. Ji W. & Chee K.-C. (2011). Prediction of hourly solar radiation using a novel hybrid model of ARMA and TDNN. Solar Energy, 85(5), 808–817.
- Ogliari E., Grimaccia F., Leva S., & Mussetta M. (2013). Hybrid predictive models for accurate forecasting in PV systems. Energies, 6(4), 1918–1929.
- 66. Goswami D. Y. (2008). A review and future prospects of renewable energy in the global energy system. In: Goswami D. Y. & Zhao Y. (eds.), Proceedings of ISES World Congress 2007 (Vol. I–Vol. V), Springer, Berlin, Heidelberg.

Artificial Intelligence

Nilsson N. (1980). Principles of Artificial Intelligence. Springer-Verlag, New York, NY. ISBN 3-540-11340-1.

Bose K. B. (2017). Artificial intelligence techniques in smart grid and renewable energy systems—Some example applications. Proceedings of the IEEE, 105(11), 2263–2273. McCorduck P. (2004). Machines Who Think.

https://monoskop.org/images/1/1e/McCorduck_Pamela_Machines_Who_Think_2nd_ed.pdf Heron of Alexandria. Encyclopedia Britannica. Encyclopedia Britannica, 1-24-017. https://www.britannica.com/biography/Heron-of-Alexandria

Cottingham J. (1978). 'A Brute to the Brutes?': Descartes' treatment of animals. Philosophy, 53(206), 551–559.

Laland K. N. & Hoppitt W. (2003). Do animals have culture? Evolutionary Anthropology, 12(3), 150–159.

Karwatka P. & Dennis M. (2004). Blaise Pascal and the first calculator. Tech Directions, 64(4), 10–15. 1p. 2 Black and White Photographs.

https://search.proquest.com/openview/bc3f2d92bff9d978494763814084ad0c/1.pdf?pq-origsite=gscholar&cbl=182

Solla Price D. D. (1984). A history of calculating machines. IEEE Micro, 4(1), 22–52. Samuel B. (1873). The Book of the Machines: Chapters in *Erewhon*. The Floating Press. https://oheg.org/hyr.pdf

Asimov I. (2004). I Robot. (1). Spectra, New York, NY.

Anderson S. L. (2008). Asimov's "three laws of robotics" and machine metaethics. AI & Society, 22(4), 477–493.

Minsky M. (1968). Semantic Information Processing. MIT Press, Cambridge, MA. Minsky M. (1961). Steps toward artificial intelligence. Proceedings of the Institute of Radio Engineers, 49(1), 8–30.

Newell A. & Simon H. (1972). Human Problem Solving. Prentice-Hall, Englewood Cliffs, NJ. Goldstein I. & Papert S. (1977). Artificial intelligence, language and the study of knowledge. Cognitive Science, 1(1), 84–123.

Lindsay R. K., Buchanan B. G., Feigenbaum E. A., & Lederberg J. (1980). Applications of Artificial Intelligence for Chemical Inference: The DENDRAL Project. McGraw-Hill, New York, NY.

American Association for Artificial Intelligence (AAAI). (2005). Menlo Park, CA. www.aaai.org/aitopics/history

Kalogirou S. A. (2000). Applications of artificial neural-networks for energy systems. Applied Energy, 67(1–2), 17–35.

Kalogirou S. A. (2001). Artificial neural networks in renewable energy systems applications: A review. Renewable and Sustainable Energy Reviews, 5(4), 373–401.

Chien J. T. (2005). Predictive hidden Markov model selection for speech recognition. IEEE Transaction on Speech and Audio Processing, 13(3), 377–387.

Mellit A., Kalogirou S. A., Shaari S., Salhi H., & Hadj Arab A. (2008). Methodology for predicting sequences of mean monthly clearness index and daily solar radiation data in remote areas: Application for sizing a stand-alone PV system. Renewable Energy, 33(7), 1570–1590.

Mellit A. & Pavan A. M. (2010). A 24-h forecast of solar irradiance using artificial neural network: Application for performance prediction of a grid-connected PV plant at Trieste, Italy. Solar Energy, 84(5), 807–821.

Palmer A. C. & Sorger P. K. (2017). Combination cancer therapy can confer benefit via patientto-patient variability without drug additivity or synergy. Cell, 171(7), 1678–1691.

Rashid M. B. M. A. et al. (2018). Optimizing drug combinations against multiple myeloma using a quadratic phenotypic optimization platform. Science Translation Medicine, 10(453), eaan0941. Lin A. et al. (2019). Off-target toxicity is a common mechanism of action of cancer drugs undergoing clinical trials. Science Translation Medicine, 11(509), eaaw8412.

Wong C. H., Kein W. S., & Andrew W. L. (2019). Estimation of clinical trial success rates and related parameters. Biostatistics, 20(2), 273–286.

Harrer S., Shah P., Antony B., & Hu J. (2019). Artificial intelligence for clinical trial design. Trends in Pharmacological Sciences, 40(8), 577–591.

Rajkomar A. et al. (2018). Scalable and accurate deep learning with electronic health records. NPJ Digital Medicine, 1(18). https://doi.org/10.1038/s41746-018-0029-1

Hernández-Escobedo Q., Manzano-Agugliaro F., Gazquez-Parra J. A., & Zapata-Sierra A. (2011). Is the wind a periodical phenomenon? The case of Mexico. Renewable and Sustainable Energy Reviews, 15(1), 721–728.

Hernández-Escobedo Q., Manzano-Agugliaro F., & Zapata-Sierra A. (2010). The wind power of Mexico. Renewable and Sustainable Energy Reviews, 14(9), 2830–2840.

Henriksen L. C. (2010). Wind energy literature survey no 16. Wind Energy, 13(4), 524–526. EWEA. (2006–2015). Wind in power European statistics. http://www.ewea.org/statistics

Wang L. X. & Mendel M. (1992). Generating fuzzy rules by learning from examples. IEEE Transactions on Systems, Man and Cybernetics, 22(6), 1414–1426.

Wang L. X. (1996). A course in fuzzy systems and control. Prentice-Hall Inc, Upper Saddle River, NJ.

Lapedes A. & Farber R. (1987). Nonlinear Signal Processing Using Neural Networks: Prediction and System Modeling. Technical report LA-UR-87-2662. Los Alamos National Laboratory, Los Alamos, NM.

Monfared M., Rastegar H., & Kojabadi H. M. (2009). A new strategy for wind speed forecasting using artificial intelligent methods. Renewable Energy, 34(3), 845–848.

Potter C. W. & Negnevitsky M. (2006). Very short-term wind forecasting for Tasmanian power generation. IEEE Transactions on Power Systems, 21(2), 965–972.

Saurenergy.com. https://www.saurenergy.com/solar-energy-articles/the-role-of-ai-and-ml-in-solar-energy

Trappey A. M. C., Chen P. P. J., Trappey C. V., & Ma L. (2019). A machine learning approach for solar power technology review and patent evolution analysis. Applied Sciences, 9(7), 1478. AI-Alawi S. M. & AI-Hinai H. A. (1998). An ANN-based approach for predicting global radiation in locations with no direct measurement instrumentation. Renewable Energy, 14(1–4) 199–204. Sharma N. K., Tiwari P. K., & Sood Y. R. (2012). Solar energy in India: Strategies, policies, perspectives and future potential. Renewable and Sustainable Energy Reviews, 16(1), 933–941.

Ramedani Z., Omid M., Keyhani A., Shamshirband S., & Khoshnevisan B. (2014). Potential of radial basis function based support vector regression for global solar radiation prediction. Renewable and Sustainable Energy Reviews, 39, 1005–1011.

Kumar A., Kumar K., Kaushik N., Sharma S., & Mishra S. (2010). Renewable energy in India: Current status and future potentials. Renewable and Sustainable Energy Reviews, 14(8), 2434–2442.

Dawn S., Tiwari P. K., Goswami A. K., & Mishra M. K. (2016). Recent developments of solar energy in India: Perspectives, strategies and future goals. Renewable and Sustainable Energy Reviews, 62, 215–235.

Philibert C. et al. (2014). Technology Roadmap: Solar Thermal Electricity. International Energy Agency, Paris, France.

Li C.-H., Zhu X.-J., Cao G.-Y., Sui S., & Hu M.-R. (2009). Dynamic modeling and sizing optimization of stand-alone photovoltaic power systems using hybrid energy storage technology. Renewable Energy, 34(3), 815–826.

Baños R., Manzano-Agugliaro F., Montoya F. G., Gil C., Alcayde A., & Gómez J. (2010). Optimization methods applied to renewable and sustainable energy: A review. Renewable and Sustainable Energy Reviews, 15(4), 1753–1766.

Bosch J. L., López G., & Batlles F. J. (2008). Daily solar irradiation estimation over a mountainous area using artificial neural networks. Renewable Energy, 33(7), 1622–1628. Veerachary M. & Narri Y. (2000). ANN based peak power tracking for PV supplied DC motors. Solar Energy, 69(4), 343–350.

Rehman S. & Mohandes M. (2008). Artificial neural network estimation of global solar radiation using air temperature and relative humidity. Energy Policy, 36(2), 571–576.

Premalatha N. & Valan Arasu A. (2016). Prediction of solar radiation for solar systems by using ANN models with different back propagation algorithms. Journal of Applied Research and Technology, 14(3), 206–214.

Mellit A. & Kalogirou A. S. (2011). ANFIS-based modelling for photovoltaic power supply system: a case study. Renewable Energy, 36(1), 250–258.

IEA.org. https://www.iea.org/reports/global-energy-review-2020/renewables

Ren21.net. https://www.ren21.net/what-are-the-current-trends-in-renewable-energy/ CSiS.org. https://www.csis.org/analysis/optimizing-indias-electricity-grid-renewables-using-aiand-machine-learning-applications

Wang H., Lei Z., Zhang X., Zhou B., & Peng J. (2019). A review of deep learning for renewable energy forecasting. Energy Conversion and Management, 198, 111799. DOI: 10.1016/j.enconman.2019.111799.

Jhaa S. Kr., Jasmin B., Anju J., Nilesh P., & Han Z. (2017). Renewable energy: Present research and future scope of artificial intelligence. Renewable and Sustainable Energy Reviews, 77(C), 297–317.

Zhao X., Wang C., Su J., & Wang J. (2019). Research and application based on the swarm intelligence algorithm and artificial intelligence for wind farm decision system. Renewable Energy, 134, 681–697.

RESET.org. https://en.reset.org/blog/germany-artificial-intelligence-making-wind-turbines-more-bird-friendly-02162020

Fu T. & Wang C. (2018). A hybrid wind speed forecasting method and wind energy resource analysis based on a swarm intelligence optimization algorithm and an artificial intelligence model. Sustainability, 10(3913), 1–24.

Mellita A. & Kalogirou A. S. (2008). Artificial intelligence techniques for photovoltaic applications: A review. Progress in Energy and Combustion Science, 34(5), 574–632. Pillai I. R. & Banerjee R. (2009). Renewable energy in India: Status and potential. Energy, 34(8), 970–980.

Caputo D., Grimaccia F., Mussetta M., & Zich R. E. (2010). Photovoltaic plants predictive model by means of ANN trained by a hybrid evolutionary algorithm. In: Proceedings IEEE JCNN Barcelona, 1–6.

Ji W. & Chee K.-C. (2011). Prediction of hourly solar radiation using a novel hybrid model of ARMA and TDNN. Solar Energy, 85(5), 808–817.

Ogliari E., Grimaccia F., Leva S., & Mussetta M. (2013). Hybrid predictive models for accurate forecasting in PV systems. Energies, 6(4), 1918–1929.

Goswami D. Y. (2008). A review and future prospects of renewable energy in the global energy system. In: Goswami D. Y. & Zhao Y. (eds.), Proceedings of ISES World Congress 2007 (Vol. I–Vol. V), Springer, Berlin, Heidelberg.

Role of AI in Renewable Energy Management

Elsheikh, A. H., & Elaziz, M. A. (2019). Review on applications of particle swarm optimization in solar energy systems. International Journal of Environmental Science and Technology, 16(2), 1159–1170.

Chavan, S., & Chavan, M. (2020). Recent trends in ICT-enabled renewable energy systems. In: Tuba, M., Akashe, S., & Joshi A. (eds), Information and Communication Technology for Sustainable Development. Advances in Intelligent Systems and Computing, Vol. 933 (pp. 327–332). Springer, Singapore.

Andoni, M., Robu, V., Flynn, D., Abram, S., Geach, D., Jenkins, D., & Peacock, A. (2019). Blockchain technology in the energy sector: A systematic review of challenges and opportunities. Renewable and Sustainable Energy Reviews, 100, 143–174.

Cali, U., & Fifield, A. (2019). Towards the decentralized revolution in energy systems using blockchain technology. International Journal of Smart Grid and Clean Energy, 8(3), 245–256. Ji, L., Wang, Z., Chen, M., Fan, S., Wang, Y., & Shen, Z. (2019). How much can AI techniques improve surface air temperature forecast?—A report from AI Challenger 2018 Global Weather Forecast Contest. Journal of Meteorological Research, 33(5), 989–992. doi: 10.1007/s13351-019-9601-0

Notton, G., Voyant, C., Fouilloy, A., Duchaud, J. L., & Nivet, M. L. (2019). Some applications of ANN to solar radiation estimation and forecasting for energy applications. Applied Sciences, 9(1), 209.

Mohanty, S., Patra, P. K., Mohanty, A., Viswavandya, M., & Ray, P. K. (2019). Artificial intelligence based forecasting & optimization of solar cell model. Optik, 181, 842–852.

Bermejo, J. F., Fernández, J. F. G., Polo, F. O., & Márquez, A. C. (2019). A review of the use of artificial neural networks models for energy and reliability prediction. A study for the solar PV, hydraulic and wind energy sources. Applied Sciences, 9, 1844.

Zhao, X., Wang, C., Su, J., & Wang, J. (2019). Research and application based on the swarm intelligence algorithm and artificial intelligence for wind farm decision systems. Renewable Energy, 134, 681–697.

Kosovic, B., Haupt, S. E., Adriaansen, D., Alessandrini, S., Wiener, G., Delle Monache, L., & Politovich, M. (2020). A comprehensive wind power forecasting system integrating artificial intelligence and numerical weather prediction. Energies, 13(6), 1372.

Çevik, H. H., Çunkaş, M., & Polat, K. (2019). A new multistage short-term wind power forecast model using decomposition and artificial intelligence methods. Physica A: Statistical Mechanics and Its Applications, 534, 122177.

Jafarian-Namin, S., Goli, A., Qolipour, M., Mostafaeipour, A., & Golmohammadi, A. M. (2019). Forecasting the wind power generation using Box–Jenkins and hybrid artificial intelligence: A case study. International Journal of Energy Sector Management, 13(4), 1038–1062.

Mabel, M. C., & Fernandez, E. (2008). Analysis of wind power generation and prediction using ANN: A case study. Renewable Energy, 33(5), 986–992.

Li, G., & Shi, J. (2010). On comparing three artificial neural networks for wind speed forecasting. Applied Energy, 87(7), 2313–2320.

Mohandes, M., Rehman, S., & Rahman, S. M. (2011). Estimation of wind speed profile using an adaptive neuro-fuzzy inference system (ANFIS). Applied Energy, 88(11), 4024–4032.

Yang, Z., Liu, Y., & Li, C. (2011). Interpolation of missing wind data based on ANFIS. Renewable Energy, 36(3), 993–998.

Malik, H., & Garg, S. (2019). Long-term solar irradiance forecast using artificial neural network: Application for performance prediction of Indian cities. In: Applications of Artificial Intelligence Techniques in Engineering (pp. 285–293). Springer, Singapore.

Benali, L., Notton, G., Fouilloy, A., Voyant, C., & Dizene, R. (2019). Solar radiation forecasting using artificial neural network and random forest methods: Application to normal beam, horizontal diffuse and global components. Renewable Energy, 132, 871–884.

AI-Based Renewable Energy with Emerging Applications

M. N. Akhter, S. Mekhilef, H. Mokhlis and N.M. Shah. "Review on forecasting of photovoltaic power generation based on machine learning and metaheuristic techniques," IET Renewable Power Generation, vol. 13, pp. 1009–1023, 2019.

N. Phuangpornpitak and S. Tia, "Opportunities and challenges of integrating renewable energy in smart grid system," Energy Procedia, vol. 34, pp. 282–290, 2013.

J. Heinermann and O. Kramer, "Machine learning ensembles for wind power prediction," Renewable Energy, vol. 89, pp. 671–679, 2016.

M. A. F. B. Lima, P. C. M. Carvalho, L. M. Fernández-Ramírez and A. P. S. Braga. "Improving solar forecasting using deep learning and portfolio theory integration," Energy, vol. 195, pp. 1–25, 2020.

R. B. Ammar, M. B. Ammar and A. Oualha, "Photovoltaic power forecast using empirical models and artificial intelligence approaches for water pumping systems," Renewable Energy, vol. 153, pp. 1016–1028, 2020.

T. Ahmad, H. Zhang and B. Yan, "A review on renewable energy and electricity requirement forecasting models for smart grid and buildings," Sustainable Cities and Society, vol. 55, pp. 1–101, 2020.

S. M. Dawaoud, X. Lin and M. I. Okba, "Hybrid renewable microgrid optimization techniques: A review," Renewable and Sustainable Energy Reviews, vol. 82, pp. 2039–2052, 2018. E. M. Sandhu and T. Thakur, "Issues, challenges, causes, impacts and utilization of renewable

energy sources - grid integration," International Journal of Engineering Research and Applications, vol. 4, pp. 636–643, 2014.

A. Essl, A. Ortner and P. Hetteger, "Machine learning analysis for a flexibility energy approach towards renewable energy integration with dynamic forecasting of electricity balancing power", in *14th International Conference on the European Energy Market (EEM)*, Dresden, 2017, pp.

1–6.

A. Gligor, C. D. Dumitru and H. S. Grif, "Artificial intelligence solution for managing a photovoltaic energy production unit," Procedia Manufacturing, vol. 22, pp. 626–633, 2018.
M. Gautam, S. Raviteja and R. Mahalakshmi, "Household energy management model to maximize solar power utilization using machine learning," Procedia Computer Science, vol. 165, pp. 90–96, 2019.

P. Sharmila, J. Baskaran, C. Nayanatara and R. Maheswari. "A hybrid technique of machine learning and data analytics for optimized distribution of renewable energy resources targeting smart energy management," Procedia Computer Science, vol. 165, pp. 278–284, 2019.
M. Alkandari and I. Ahmad, "Solar power generation forecasting using ensemble approach based on deep learning and statistical methods," Applied Computing and Informatics, vol. 19, pp. 1–26, 2019.

M. Z. Sušac, S. Mitrović and A. Has, "Machine learning based system for managing energy efficiency of public sector as an approach towards smart cities," International Journal of Information Management, vol. 20, pp. 1–12, 2020.

T. Som, M. Dwivedi, C. Dubey and A. Sharma. "Parametric studies on artificial intelligence techniques for battery SOC management and optimization of renewable power," Procedia Computer Science, vol. 167, pp. 353–362, 2020.

R. Ak, O. Fink and E. Zio. "Two machine learning approaches for sort-term wind speed timeseries prediction," IEEE Transactions on Neural Networks and Learning Systems, vol. 27, pp. 1734–1747, 2016.

M. S. Al-Musaylh, R. C. Deo, J. F. Adamowski and Y. Li. "Short-term electricity demand forecasting using machine learning methods enriched with ground-based climate and ECMWF reanalysis atmospheric predictors in southeast Queensland, Australia," Renewable and Sustainable Energy Reviews, vol. 113, pp. 3–22, 2019.

K. N. Prakash and P. D. Vadana. "Machine learning based residential energy management system," in *IEEE International Conference on Computational Intelligence and Computing Research (ICCIC)*, Coimbatore, pp. 1–4, 2017.

A. T. D. Perera, P. U. Wickramsinghe, V. M. Nik and J.-L. Scartezzini. "Machine learning methods to assist energy system optimization," Applied Energy, vol. 243, pp. 191–205, 2019.
S. S. Sanz, L. C. Bueno, L. Prieto et al. "Feature selection in machine learning prediction systems for renewable energy applications", Renewable and Sustainable Energy Reviews, vol. 90, pp. 728–741, 2018.

B. Zhang, W. Hu, D. Cao et al. "Deep reinforcement learning–based approach for optimizing energy conversion in integrated electrical and heating system with renewable energy", Energy Conversion and Management, vol. 202, pp. 1–13, 2019.

T. Ahmad, H. Chen, W. A. Shah "Effective bulk energy consumption control and management for power utilities using artificial intelligence techniques under conventional and renewable energy resources," EPES, vol. 109, pp. 242–258, 2019.

E. Hossain, I. Khan, F. U. Noor et al. "Application of big data and machine learning in smart grid, and associated security concerns: A review", 10.1109/ACCESS.2019.2894819, *IEEE Access*, pp. 1–40, 2017.

S. K. Jha, J. Bilalovic, A. Jha et al. "Renewable energy: Present research and future scope of Artificial Intelligence", RSER, vol. 77, pp. 297–317, 2017.

K. W. Kow, Y. W. Wong, R. K. Rajkumar et al. "A review on performance of artificial intelligence and conventional method in mitigating PV grid-tied related power quality events", RSER, vol. 56, pp. 334–346, 2016.

K. R. Kumar and M. S. Kalavathi "Artificial intelligence based forecast models for predicting solar power generation," PMME, vol. 5, pp. 796–802, 2018.

M. Borunda, O. A. Jaramillo, A. Reyes et al. "Bayesian network in renewable energy systems: A bibliographical survey", RSER, vol. 62, pp. 32–45, 2016.

M. A. M. Daut, M. Y. Hassan, H. Abdullah et al. "Building electrical energy consumption forecasting analysis using conventional and artificial intelligence methods: A review", RSER, vol. 16, pp. 1–11, 2016.

M. Ramezanizadeh, M. H. Ahmadi, M. A. Nazari et al. "A review on the utilized machine learning approaches for modeling the dynamic viscosity of nanofluids", RSER, vol. 114, pp. 1–15, 2019.

M. Sharifzadeh, A. S. Lock and N. Shah, "Machine-learning methods for integrated renewable power generation: A comparative study of artificial neural networks, support vector regression, and Gaussian process regression," Renewable and Sustainable Energy Reviews, vol. 108, pp. 513–538, 2018.

G. D. Sharma, A. Yadav and R. Chopra, "Artificial Intelligence and Effective Governance: A Review, Critique and Research Agenda," AJTES, vol. 20, pp. 1–16, 2020.

S. Sinha and S. S. Chandel, "Review of recent trends in optimization techniques for solar photovoltaic–wind-based hybrid energy systems," RSER, vol. 50, pp. 755–769, 2015.

A. Stetco, F. Dimohammadi, X. Zhao et al. "Machine learning methods for wind turbine condition monitoring: A review", IJRE, vol. 18, pp. 1–23, 2018.

C. Voyant, G. Notton, S. Kalogirou et al. "Machine learning methods for solar radiation forecasting: A review", Renewable Energy, vol. 105, pp. 569–582, 2017.

S. Walker, W. Khan, K. Katic et al. "Accuracy of different machine learning algorithms and added-value of predicting aggregated-level energy performance of commercial buildings", Energy & Buildings, vol. 209, pp. 1–14, 2020.

Z. Wang and R. S. Srinivasan, "A review of artificial intelligence-based building energy use prediction: Contrasting the capabilities of single and ensemble prediction models", Renewable and Sustainable Energy Reviews, vol. 16, pp. 1–13, 2016.

H. Wang, Z. Lei, X. Zhang et al. "A review of deep learning for renewable energy forecasting", Energy Conversion and Management, vol. 198, pp. 1–16, 2019.

H. Wang, Y. Liu, B. Zhou et al. "Taxonomy research of artificial intelligence for deterministic solar power forecasting", ECM, vol. 214, pp. 1–17, 2020.

J. H. Yousif, H. A. Kazem, N. N. Alattar et al. "A comparison study based on artificial neural network for assessing PV/T solar energy production", Case Studies in Thermal Engineering, vol. 13, pp. 1–13, 2019.

S. M. Zahraee, M. K. Assadi and R. Saidur, "Application of artificial intelligence methods for hybrid energy system optimization", RSER, vol. 66, pp. 617–630, 2016.

Foundations of Machine Learning

Kaplan, A., and Haenlein, M. (2019). "Siri, Siri, in my hand: Who's the fairest in the land? On the interpretations, illustrations, and implications of artificial intelligence." Business Horizons. 62:15–25.

Nilsson, N. (1980). Principles of Artificial Intelligence. Palo Alto, CA: Tioga Press.

Singh, A. (2020). Python Programming Universe v 3.8: With Walrus Operator, Positional Argument Operator and Many More... (Vol. 1). Independently Published.

Scudder, H. (1965). "Probability of error of some adaptive pattern-recognition machines." IEEE Transactions on Information Theory. 11(3):363–371.

https://en.wikipedia.org/wiki/Semi-supervised_learning#cite_note-survey-7.

Zhu, X. (2008). "Semi-supervised learning literature survey." Technical Report 1530. University of Wisconsin-Madison.

Lagoudakis, M. G., and Parr, R. (2003). "Reinforcement learning as classification: Leveraging modern classifiers." In: Proceedings of the Twentieth International Conference on Machine Learning (ICML-2003), Washington, DC.

Roelfsema, P. R., and van Ooyen, A. (2005). "Attention-gated reinforcement learning of internal representations for classification", Neural Computation. 17:2176–2214.

Kaelbling, L. P., Littman, M. L., and Moore, A. W. (1996). "Reinforcement learning: A survey." Journal of Artificial Intelligence Research. 4:237–285.

Sutton, R. S., and Barto, A. G. (1998) Reinforcement Learning: An Introduction. Cambridge MA: The MIT Press.

Wang, Y., Xie, Q., Ammari, A., and Pedram, M. (2011). "Deriving a near-optimal power management policy using model-free reinforcement learning and Bayesian classification." In: Proceedings of the 48th Design Automation Conference (DAC 2011), San Diego, CA.

Benini, L., Bogliolo, A., and De Micheli, G. (2000). "A survey of design techniques for system level dynamic power management." IEEE Transactions on VLSI Systems. 8(3):299–316.

https://energyfactor.exxonmobil.com/news/mit-collaboration.

https://deepmind.com/blog/article/deepmind-ai-reduces-google-data-centre-cooling-bill-40. https://javatpoint.com/applications-of-machine-learning.

Introduction of AI Techniques and Approaches

Lei, Y., He, Z., & Zi, Y. (2008). A new approach to intelligent fault diagnosis of rotating machinery. Expert Systems with Applications, *35*(4), 1593–1600.

Dilek, S., Çakır, H., & Aydın, M. (2015). Applications of artificial intelligence techniques to combating cyber crimes: A review. Preprint arXiv:1502.03552.

Yampolskiy, R. V. (Ed.). (2018). Artificial intelligence safety and security. Boca Raton, FL: CRC Press.

Demertzis, K., & Iliadis, L. (2015). A bio-inspired hybrid artificial intelligence framework for cyber security. In Computation, Cryptography, and Network Security (pp. 161–193). Cham: Springer. Brefeld, U., & Zimmermann, A. (2017). Guest editorial: Special issue on sports analytics. Data Mining and Knowledge Discovery, *31*(6), 1577–1579.

Yu, Y., Choi, T. M., & Hui, C. L. (2011). An intelligent fast sales forecasting model for fashion products. Expert Systems with Applications, *38*(6), 7373–7379.

Kim, H. S., & Cho, S. B. (2000). Application of interactive genetic algorithm to fashion design. Engineering Applications of Artificial Intelligence, *13*(6), 635–644.

Semenov, V. P., Chernokulsky, V. V., & Razmochaeva, N. V. (2017, October). Research of artificial intelligence in the retail management problems. In *2017 IEEE II International Conference on Control in Technical Systems (CTS)* (pp. 333–336).

Wong, C., Guo, Z. X., & Leung, S. Y. S. (2013). Optimizing decision making in the apparel supply chain using artificial intelligence (AI): From production to retail. Cambridge: Elsevier. Fethi, M. D., & Pasiouras, F. (2010). Assessing bank efficiency and performance with operational research and artificial intelligence techniques: A survey. European Journal of Operational Research, *204*(2), 189–198.

Pomerleau, D. A. (1989). ALVINN: An autonomous land vehicle in a neural network. In Advances in neural information processing systems *1* (pp. 305–313). San Francisco, CA: Morgan Kaufmann.

Campbell, M., Hoane Jr, A. J., & Hsu, F. H. (2002). Deep blue. Artificial Intelligence, *134*(1–2), 57–83.

Bennett, J. S., & Hollander, C. R. (1981, August). DART: An Expert System for Computer Fault Diagnosis. In *Proceedings of the VIIth IJCAI*, Vancouver (pp. 843–845).

Littman, M. L., Keim, G. A., & Shazeer, N. M. (1999, July). Solving crosswords with PROVERB. In *Proceedings of the Sixteenth National Conference on Artificial Intelligence (AAAI-99)*, Orlando, FL (pp. 914–915).

Pomerleau, D. A. (1993). Knowledge-based training of artificial neural networks for autonomous robot driving. In Robot learning (pp. 19–43). Boston, MA: Springer.

Li, Z., Wang, J., Li, B., Gao, J., & Tan, X. (2014). GPS/INS/Odometer integrated system using fuzzy neural network for land vehicle navigation applications. The Journal of Navigation, *67*(6), 967–983.

Newborn, M. (2012). Kasparov versus Deep Blue: Computer chess comes of age. Berlin: Springer Science & Business Media.

Seirawan, Y., Simon, H. A., & Munakata, T. (1997). The implications of Kasparov vs. Deep Blue. Communications of the ACM, *40*(8), 21–25.

Rashmi, K. V., & Gilad-Bachrach, R. (2015, May). DART: Dropouts meet multiple additive regression trees. In *Proceedings of the 18th International Conference on Artificial Intelligence and Statistics (AISTATS)*, San Diego, CA (pp. 489–497).

Hedberg, S. R. (2002). DART: revolutionizing logistics planning. IEEE Intelligent Systems, *17* (3), 81–83.

Borrajo, D., Ríos, J., Pérez, M. A., & Pazos, J. (1990). Dominoes as a domain where to use proverbs as heuristics. Data & Knowledge Engineering, *5*(2), 129–137.

Aggarwal, M. (2011). Information retrieval and question answering NLP approach: An artificial intelligence application. International Journal of Soft Computing and Engineering (IJSCE), *1* (NCAI2011).

Jacobs, P. S. (1992, March). Joining statistics with NLP for text categorization. In *Proceedings* of the Third Conference on Applied Natural Language Processing (pp. 178–185).

Searle, J. R. (1980). Minds, brains, and programs. Behavioral and Brain Sciences, 3(3), 417–424.

Omohundro, S. M. (2008, February). The basic AI drives. In *Proceedings of the 2008 Conference on Artificial General Intelligence* (Vol. 171, pp. 483–492).

Bostrom, N., Dafoe, A., & Flynn, C. (2016). Policy desiderata in the development of machine superintelligence. Future of Humanity Institute, University of Oxford. Retrieved June 8, 2018. Gill, K. S. (2016). Artificial super intelligence: beyond rhetoric. AI and Society, *31*, 137–143.

Moor, J. (Ed.). (2003). The Turing test: The elusive standard of artificial intelligence (Vol. 30). Berlin: Springer Science & Business Media.

Saygin, A. P., Cicekli, I., & Akman, V. (2000). Turing test: 50 years later. Minds and Machines, *10*(4), 463–518.

Hodges, A. (2012). Alan Turing: The Enigma. London: Random House.

Friedman, C., & Hripcsak, G. (1999). Natural language processing and its future in medicine. Academic Medicine, *74*(8), 890–895.

Brachman, R. J., & Levesque, H. J. (1985). Readings in knowledge representation. Los Altos, CA: Morgan Kaufmann Publishers Inc.

Wos, L., Overbeck, R., Lusk, E., & Boyle, J. (1984). Automated reasoning: introduction and applications. Englewood Cliffs, NJ: Prentice-Hall.

Bratko, I. (1993). Machine learning in artificial intelligence. Artificial Intelligence in Engineering, 8(3), 159–164.

Tan, U. (2007). The psychomotor theory of human mind. International Journal of Neuroscience, *117*(8), 1109–1148.

Macaulay, M., & Arjoon, S. (2013). An Aristotelian-Thomistic approach to professional ethics. Journal of Markets & Morality, *16*(2), 507–527.

Miiller, Y. (1990). Decentralized artificial intelligence. In: Decentralised A.I.(pp. 3–13). Amsterdam: North Holland.

Yu, K. H., Beam, A. L., & Kohane, I. S. (2018). Artificial intelligence in healthcare. Nature Biomedical Engineering, *2*(10), 719–731.

Hamscher, W. (1994). AI in business-process reengineering. AI Magazine, *15*(4), 71. Beck, J., Stern, M., & Haugsjaa, E. (1996). Applications of AI in education. XRDS: Crossroads, The ACM Magazine for Students, *3*(1), 11–15.

Rychtyckyj, N. (1999, July). DLMS: Ten Years of AI for Vehicle Assembly Process Planning. In *Proceedings of the* AAAI/IAAI (pp. 821–828).

Li, W., Logenthiran, T., Phan, V. T., & Woo, W. L. (2016, November). Intelligent multi-agent system for power grid communication. In *2016 IEEE Region 10 Conference (TENCON)* (pp. 3386–3389).

Sozontov, A., Ivanova, M., & Gibadullin, A. (2019). Implementation of artificial intelligence in the electric power industry. In E3S Web of Conferences (Vol. 114, p. 01009).

Hernández, L., et al. (2013). A multi-agent system architecture for smart grid management and forecasting of energy demand in virtual power plants. IEEE Communications Magazine, *51*(1), 106–113.

Szkuta, B. R., Sanabria, L. A., & Dillon, T. S. (1999). Electricity price short-term forecasting using artificial neural networks. IEEE Transactions on Power Systems, *14*(3), 851–857. Alpaydin, E. (2020). Introduction to machine learning. Cambridge, MA: MIT Press.

Miyato, T., Maeda, S. I., Koyama, M., & Ishii, S. (2018). Virtual adversarial training: a regularization method for supervised and semi-supervised learning. IEEE Transactions on Pattern Analysis and Machine Intelligence, *41*(8), 1979–1993.

Baldi, P. (2012, June). Autoencoders, unsupervised learning, and deep architectures. In *Proceedings of ICML Workshop on Unsupervised and Transfer Learning* (pp. 37–49). Srivastava, N., Mansimov, E., & Salakhudinov, R. (2015, June). Unsupervised learning of video representations using LSTMs. In *Proceedings of the 32nd International Conference on Machine*

Learning (pp. 843–852).

Niebles, J. C., Wang, H., & Fei-Fei, L. (2008). Unsupervised learning of human action categories using spatial-temporal words. International Journal of Computer Vision, *79*(3), 299–318.

Lee, H., Grosse, R., Ranganath, R., & Ng, A. Y. (2011). Unsupervised learning of hierarchical representations with convolutional deep belief networks. Communications of the ACM, *54*(10), 95–103.

Memisevic, R., & Hinton, G. (2007, June). Unsupervised learning of image transformations. In 2007 IEEE Conference on Computer Vision and Pattern Recognition (pp. 1–8).

Andreae, J. H. (1995, November). The future of associative learning. In *Proceedings of the* 1995 Second New Zealand International Two-Stream Conference on Artificial Neural Networks and Expert Systems (pp. 194–197).

Hugh, A. J., & Lawrence, J. P. (1967). U.S. Patent No. 3,355,713. Washington, DC: U.S. Patent and Trademark Office.

Abbeel, P., & Ng, A. Y. (2004, July). Apprenticeship learning via inverse reinforcement learning. In *Proceedings of the Twenty-First International Conference on Machine Learning* (p. 1). Wiering, M., & Van Otterlo, M. (2012). Reinforcement learning. Adaptation, Learning, and Optimization. *12*, 3.

Ziebart, B. D., Maas, A. L., Bagnell, J. A., & Dey, A. K. (2008, July). Maximum entropy inverse reinforcement learning. In *Proceedings of the Twenty-third AAAI Conference on Artificial Intelligence* (Vol. 8, pp. 1433–1438).

Rothkopf, C. A., & Dimitrakakis, C. (2011, September). Preference elicitation and inverse reinforcement learning. In *Joint European Conference on Machine Learning and Knowledge Discovery in Databases* (pp. 34–48). Berlin, Heidelberg: Springer.

Anderson, M. J. (2009). Carl Linnaeus: Father of classification. Berkeley Heights, NJ: Enslow Publishing, LLC.

Ye, Q., Zhang, Z., & Law, R. (2009). Sentiment classification of online reviews to travel destinations by supervised machine learning approaches. Expert Systems with Applications, *36* (3), 6527–6535.

Jain, P., Garibaldi, J. M., & Hirst, J. D. (2009). Supervised machine learning algorithms for protein structure classification. Computational Biology and Chemistry, *33*(3), 216–223. Unler, A., & Murat, A. (2010). A discrete particle swarm optimization method for feature selection in binary classification problems. European Journal of Operational Research, *206*(3), 528–539.

Ball, C. A., & Tschoegl, A. E. (1982). The decision to establish a foreign bank branch or subsidiary: An application of binary classification procedures. Journal of Financial and Quantitative Analysis, *17*(3), 411–424.

Ghamrawi, N., & McCallum, A. (2005, October). Collective multi-label classification. In *Proceedings of the 14th ACM International Conference on Information and Knowledge Management* (pp. 195–200).

Trohidis, K., Tsoumakas, G., Kalliris, G., & Vlahavas, I. P. (2008, September). Multi-label classification of music into emotions. In ISMIR *2008* (Vol. 8, pp. 325–330).

Tsoumakas, G., & Vlahavas, I. (2007, September). Random k-labelsets: An ensemble method for multilabel classification. In *European Conference on Machine Learning* (pp. 406–417). Berlin, Heidelberg: Springer.

Hastie, T., Rosset, S., Zhu, J., & Zou, H. (2009). Multi-class AdaBoost. Statistics and Its Interface, *2*(3), 349–360.

Ou, G., & Murphey, Y. L. (2007). Multi-class pattern classification using neural networks. Pattern Recognition, 40(1), 4-18.

Amit, Y., Fink, M., Srebro, N., & Ullman, S. (2007, June). Uncovering shared structures in multiclass classification. In *Proceedings of the 24th International Conference on Machine Learning* (pp. 17–24).

Huang, C., Li, Y., Change Loy, C., & Tang, X. (2016). Learning deep representation for imbalanced classification. In *Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition* (pp. 5375–5384).

Garcı, S., Triguero, I., Carmona, C. J., & Herrera, F. (2012). Evolutionary-based selection of generalized instances for imbalanced classification. Knowledge-Based Systems, *25*(1), 3–12. Sáez, J. A., Luengo, J., Stefanowski, J., & Herrera, F. (2015). SMOTE–IPF: Addressing the noisy and borderline examples problem in imbalanced classification by a re-sampling method

with filtering. Information Sciences, 291, 184–203.

Criminisi, A., Shotton, J., & Konukoglu, E. (2011). *Decision forests for classification, regression, density estimation, manifold learning and semi-supervised learning.* Microsoft Research Technical Report TR-2011-114, *5*(6), 12.

Amini, M. R., & Gallinari, P. (2002, July). Semi-supervised logistic regression. In *Proceedings of the 15th European Conference on Artificial Intelligence* (pp. 390–394).

Ritter, M. A., Harty, L. D., Davis, K. E., Meding, J. B., & Berend, M. E. (2003). Predicting range of motion after total knee arthroplasty: clustering, log-linear regression, and regression tree analysis. The Journal of Bone and Joint Surgery, *85*(7), 1278–1285.

Breiman, L., & Friedman, J. H. (1997). Predicting multivariate responses in multiple linear regression. Journal of the Royal Statistical Society: Series B (Statistical Methodology), *59*(1), 3–54.

Zou, K. H., Tuncali, K., & Silverman, S. G. (2003). Correlation and simple linear regression. Radiology, *227*(3), 617–628.

Kleinbaum, D. G., Dietz, K., Gail, M., Klein, M., & Klein, M. (2002). Logistic regression. New York, NY: Springer-Verlag.

Menard, S. (2002). Applied logistic regression analysis (Vol. 106). Thousand Oaks, CA: Sage. Demartines, P., & Hérault, J. (1997). Curvilinear component analysis: A self-organizing neural network for nonlinear mapping of data sets. IEEE Transactions on Neural Networks, *8*(1), 148–154.

Max, T. A., & Burkhart, H. E. (1976). Segmented polynomial regression applied to taper equations. Forest Science, *22*(3), 283–289.

Bendel, R. B., & Afifi, A. A. (1977). Comparison of stopping rules in forward "stepwise" regression. Journal of the American Statistical Association, *72*(357), 46–53.

Mahmood, Z., & Khan, S. (2009). On the use of k-fold cross-validation to choose cutoff values and assess the performance of predictive models in stepwise regression. The International Journal of Biostatistics, *5*(1).

Hoerl, A. E., Kannard, R. W., & Baldwin, K. F. (1975). Ridge regression: some simulations. Communications in Statistics-Theory and Methods, *4*(2), 105–123.

Fearn, T. (1983). A misuse of ridge regression in the calibration of a near infrared reflectance instrument. Journal of the Royal Statistical Society: Series C (Applied Statistics), *32*(1), 73–79. Hans, C. (2009). Bayesian lasso regression. Biometrika, *96*(4), 835–845.

Zou, H., & Hastie, T. (2005). Regularization and variable selection via the elastic net. Journal of the Royal Statistical Society: Series B (Statistical Methodology), *67*(2), 301–320.

Ogutu, J. O., Schulz-Streeck, T., & Piepho, H. P. (2012, December). Genomic selection using regularized linear regression models: Ridge regression, lasso, elastic net and their extensions. BMC *Proceedings*, *6*(S2), S10).

Preacher, K. J., Curran, P. J., & Bauer, D. J. (2006). Computational tools for probing interactions in multiple linear regression, multilevel modeling, and latent curve analysis. Journal of Educational and Behavioral Statistics, *31*(4), 437–448.

Preacher, K. J., & Rucker, D. (2003). A primer on interaction effects in multiple linear regression. Retrieved November 10, 2003, http://www.quantpsy.org/interact/interactions.htm. Zahir, N., & Mahdi, H. (2015). Snow depth estimation using time series passive microwave imagery via genetically support vector regression (case study Urmia Lake Basin). The International Archives of Photogrammetry, Remote Sensing and Spatial Information Sciences, *40*(1), 555.

Maroco, J., Silva, D., Rodrigues, A., Guerreiro, M., Santana, I., & de Mendonça, A. (2011). Data mining methods in the prediction of dementia: A real-data comparison of the accuracy, sensitivity and specificity of linear discriminant analysis, logistic regression, neural networks, support vector machines, classification trees and random forests. BMC Research Notes, *4*(1), 299.

Rossi, F., & Villa, N. (2006). Support vector machine for functional data classification. Neurocomputing, *69*(7–9), 730–742.

Chang, C. C., & Lin, C. J. (2011). LIBSVM: A library for support vector machines. ACM Transactions on Intelligent Systems and Technology (TIST), *2*(3), 1–27.

Hsu, C. W., & Lin, C. J. (2002). A comparison of methods for multiclass support vector machines. IEEE Transactions on Neural Networks, *13*(2), 415–425.

Shawe-Taylor, J., & Cristianini, N. (2000). Support vector machines (Vol. 2). Cambridge: Cambridge University Press.

Huang, G. B. (2015). What are extreme learning machines? Filling the gap between Frank Rosenblatt's dream and John von Neumann's puzzle. Cognitive Computation, *7*(3), 263–278. Dimla Sr, D. E., & Lister, P. M. (2000). On-line metal cutting tool condition monitoring.: II: toolstate classification using multi-layer perceptron neural networks. International Journal of Machine Tools and Manufacture, *40*(5), 769–781.

Lawrence, S., Giles, C. L., Tsoi, A. C., & Back, A. D. (1997). Face recognition: A convolutional neural-network approach. IEEE Transactions on Neural Networks, *8*(1), 98–113.

Socher, R., Lin, C. C., Manning, C., & Ng, A. Y. (2011). Parsing natural scenes and natural language with recursive neural networks. In *Proceedings of the 28th International Conference on Machine Learning (ICML-11)* (pp. 129–136).

Mikolov, T., Karafiát, M., Burget, L., Černocký, J., & Khudanpur, S. (2010). Recurrent neural network based language model. In *Eleventh Annual Conference of the International Speech Communication Association*.

Ma, X., Tao, Z., Wang, Y., Yu, H., & Wang, Y. (2015). Long short-term memory neural network for traffic speed prediction using remote microwave sensor data. Transportation Research Part C: Emerging Technologies, *54*, 187–197.

Sak, H., Shannon, M., Rao, K., & Beaufays, F. (2017, August). Recurrent neural aligner: An encoder-decoder neural network model for sequence to sequence mapping. In Interspeech 2017 (Vol. 8, pp. 1298–1302).

Soltanolkotabi, M., Javanmard, A., & Lee, J. D. (2018). Theoretical insights into the optimization landscape of over-parameterized shallow neural networks. IEEE Transactions on Information Theory, *65*(2), 742–769.

Chopra, A., Prashar, A., & Sain, C. (2013). Natural language processing. International Journal of Technology Enhancements and Emerging Engineering Research, *1*(4), 131–134.

Manning, C. D., Manning, C. D., & Schütze, H. (1999). Foundations of statistical natural language processing. Cambridge, MA: MIT press.

Collobert, R., Weston, J., Bottou, L., Karlen, M., Kavukcuoglu, K., & Kuksa, P. (2011). Natural language processing (almost) from scratch. Journal of Machine Learning Research, *12*, 2493–2537.

Aone, C., Okurowski, M. E., Gorlinsky, J., & Larsen, B. (1999). A trainable summarizer with knowledge acquired from robust NLP techniques. In Advances in automatic text summarization (71–80). Cambridge, MA: MIT Press.

Cambria, E., Poria, S., Gelbukh, A., & Thelwall, M. (2017). Sentiment analysis is a big suitcase. IEEE Intelligent Systems, *32*(6), 74–80.

Khanna, A., Pandey, B., Vashishta, K., Kalia, K., Pradeepkumar, B., & Das, T. (2015). A study of today's AI through chatbots and rediscovery of machine intelligence. International Journal of u- and e-Service, Science and Technology, $\mathcal{B}(7)$, 277–284.

Hulth, A. (2004). Combining machine learning and natural language processing for automatic keyword extraction. Doctoral dissertation, Institutionen för data-och systemvetenskap (tills m KTH).

Pustejovsky, J., & Boguraev, B. (1993). Lexical knowledge representation and natural language processing. Artificial Intelligence, *63*(1–2), 193–223.

Gotti, F., & Langlais, P. (2016, May). Harnessing open information extraction for entity classification in a French corpus. In Canadian Conference on Artificial Intelligence (pp. 150–161). Cham: Springer.

Johansson, F., Brynielsson, J., & Quijano, M. N. (2012, August). Estimating citizen alertness in crises using social media monitoring and analysis. In *2012 European Intelligence and Security Informatics Conference* (pp. 189–196).

Alam, F., Ofli, F., & Imran, M. (2020). Descriptive and visual summaries of disaster events using artificial intelligence techniques: case studies of Hurricanes Harvey, Irma, and Maria. Behaviour & Information Technology, *39*(3), 288–318.

A Comprehensive Overview of Hybrid Renewable Energy Systems

ADB. (2017). Deployment of hybrid renewable energy systems in mini-grids. http://dx.doi.org/10.22617/TIM178889-2

Podder, A. K., Roy, N. K., & Pota, H. R. (2019). MPPT methods for solar PV systems: A critical review based on tracking nature. IET Renewable Power Generation, 13(10), 1615–1632.

Podder, A. K., Hasan, M. R., Roy, N. K., & Komol, M. M. R. (2019). Economic analysis of a gridconnected PV system: A case study in Khulna. European Journal of Engineering Research and Science, 3(7), 16–21.

Ibrahim, M., Khair, A., & Ansari, S. (2015, August). A review of hybrid renewable energy systems for electric power generation. International Journal of Engineering Research and Applications, 5(8), 42–48.

Podder, A. K., Ahmed, K., Roy, N. K., & Habibullah, M. (2019). Design and simulation of a photovoltaic and fuel cell-based micro-grid system. In: International Conference on Energy and Power Engineering: Power for Progress, ICEPE 2019, 1–6.

Lazarov, V. D., Notton, G., Zarkov, Z., & Bochev, I. (2005). Hybrid power systems with renewable energy sources types, structures, trends for research, and development. In: Eleventh International Conference on Electrical Machines, Drives and Power Systems ELMA 2005, 515–520.

Negi, S., & Mathew, L. (2014). Hybrid renewable energy system: A review. International Journal of Electronic and Electrical Engineering, 7, 535–542.

Bappy, F. I., Islam, M. J., Podder, A. K., Dipta, D. R., Faruque, H. M. R., & Hossain, E. (2019). Comparison of different hybrid renewable energy systems with optimized PV configuration to realize the effects of multiple schemes. In: First International Conference on Advances in Science, Engineering and Robotics Technology (ICASERT), 1–6.

Rekioua, D. (2020). Hybrid renewable energy system overview. In: Rekioua, D. (Ed.), Hybrid Renewable Energy Systems: Optimization and Power Management Control (pp. 1–37). Springer, New York.

Swarnakar, S. C., Podder, A. K., & Tariquzzaman, M. (2020). Solar, fuel cell and battery based hybrid energy solution for residential appliances. In: Fourth International Conference on Electrical Information and Communication Technology (EICT), 1–6.

Reddy, Y. J., Kumar, Y. V. P., Raju, K. P., & Ramsesh, A. K. (2012, December). Hybrid power system design with renewable energy sources for buildings. IEEE Transactions on Smart Grid, 3(4), 2174–2187.

Reddy, Y. J., Kumar, Y.V. P., & Raju, K. P. (2011, September). Real time and high fidelity simulation of hybrid power system dynamics. In: 2011 IEEE Recent Advances in Intelligent Computational Systems, 890–895.

Solanki, C. S. (2011). Solar Photovoltaics: Fundamentals, technologies and Applications. Second edition. Prentice Hall of India, Delhi.

Podder, A. K., & Habibullah, M. (2018). Model predictive based energy efficient control of gridconnected PV systems. In: 10th International Conference on Electrical and Computer Engineering (ICECE), 413–416.

Podder, A. K., Habibullah, M., & Roy, N. K. (2019). Current THD analysis of model predictive control based grid-connected PV inverter. In: International Conference on Electrical, Computer and Communication Engineering (ECCE), 1–6.

Brooks, W., & Dunlop, J. (2013). PV Installation Professional Resource Guide. NABCEP, Clifton Park, NY.

United States Agency for International Development. (2013). Solar PV System Maintenance Guide. Washington, DC.

Podder, A. K., Ahmed, K., Roy, N. K., & Biswas, P. C. (2017). Design and simulation of an independent solar home system with battery backup. In: 4th International Conference on Advances in Electrical Engineering (ICAEE), 427–431.

Clarke, S. (2003). Electricity generation using small wind turbines at your home or farm. Rural Environment/OMAFRA, 1–16.

Bhandari, B., Lee, K. T., Lee, G. Y., Cho, Y. M., & Ahn, S. H. (2015, January) Optimization of hybrid renewable energy power systems: A review. International Journal of Precision Engineering and Manufacturing-Green Technology, 2(1), 99–112.

Ghofrani, M., & Hosseini, N. N., (2016). Optimizing hybrid renewable energy systems: A review. IntechOpen.

Borowy, B. S., & Salameh, Z. M. (1996). Methodology for optimally sizing the combination of a battery bank and PV array in a wind/PV hybrid system. IEEE Transactions on Energy Conversion, 11(2), 367–375.

Markvart, T. (1996). Sizing of hybrid photovoltaic-wind energy systems. Solar Energy, 57(4), 277–281.

Markvart, T., Fragaki, A., & Ross, J. (2006) PV system sizing using observed time series of solar radiation. Solar Energy, 80(1), 46–50.

Karaki, S., Chedid, R., & Ramadan R. (1999) Probabilistic performance assessment of autonomous solar-wind energy conversion systems. IEEE Transactions on Energy Conversion, 14(3), 766–772.

Tina, G., Gagliano, S., & Raiti, S. (2006). Hybrid solar/wind power system probabilistic modeling for long-term performance assessment. Solar Energy, 80 (5), 578–588.

Posadillo, R., & López Luque, R. (2008). Approaches for developing a sizing method for standalone PV systems with variable demand. Renewable Energy, 33(5), 1037–1048.

Bhuiyan, M., & Ali, A. M. (2003). Sizing of a stand-alone photovoltaic power system at Dhaka. Renewable Energy, 28(6), 929–938.

Bhandari, R., & Stadler, I. (2011). Electrification using solar photovoltaic systems in Nepal. Applied Energy, 88(2), 458–465.

Siddaiah, R., & Saini, R. (2016). A review on planning, configurations, modeling, and optimization techniques of hybrid renewable energy systems for off-grid applications. Renewable and Sustainable Energy Reviews, 58, 376–396.

Kellogg, W., Nehrir, M., Venkataramanan, G., & Gerez, V. (1996). Optimal unit sizing for a hybrid wind/photovoltaic generating system. Electric Power Systems Research, 39 (1), 35–38. Shahirinia, A. H., Tafreshi, S. M. M., Gastej, A. H., & Moghaddomjoo, A. R. (2005). Optimal sizing of hybrid power system using genetic algorithm. In: International Conference on Future Power Systems, Amsterdam, 1–6.

Hossam-Eldin, A., El-Nashar, A. M., & Ismaiel, A. (2012, August). Investigation into economical desalination using optimized hybrid renewable energy system. Electrical Power and Energy Systems, 43, 1393–1400.

Bai, Q. (2010, February). Analysis of particle swarm optimization algorithm. Computer and Information Science, 3(1), 180–184.

Huang Y., Xu Y., & Zhou X. (2011). Study on wind-solar hybrid generating system control strategy. In: International Conference on Multimedia Technology, Hangzhou, 773–776. Petalas, Y. G., Parsopoulos, K. E., Papageorgiou, E. I., Groumpos, P. P., & Vrahatis, M. N. (2007). Enhanced learning in fuzzy simulation models using memetic particle swarm optimization. In: Proceedings of the 2007 IEEE Swarm Intelligence Symposium, Honolulu, HI, 16–22.

Pranav, M. S., Karunanithi, K., Akhil, M., Saravanan, S., Afsal, V. M., & Krishnan, A. (2017). Hybrid renewable energy sources (HRES) - A review. In: International Conference on Intelligent Computing Instrumentation and Control Technologies (ICICICT), 162–165.

Fung, C. C., Ho, S. C. Y., & Nayar, C. V. (1993). Optimization of a hybrid energy system using simulated annealing technique. In: Proceedings of the 1993 IEEE Region 10 Conference on Computer, Communication, Control and Power Engineering, Beijing, 235–238.

Wu, Y., Lee, C., Liu, L., & Tsai, S. (2010). Study of reconfiguration for the distribution system with distributed generators. IEEE Transactions on Power Delivery, 25(3), 1678–1685.

Sunanda, S., & Chandel, S. S. (2015). Review of recent trends in optimization techniques for solar photovoltaic–wind based hybrid energy systems. Renewable and Sustainable Energy Reviews, 50, 755–769.

Katsigiannis, Y. A., Georgilakis, P. S., & Karapidakis, E. S. (2012). Hybrid simulated annealing Tabu search method for optimal sizing of autonomous power systems with renewable. IEEE Transactions on Sustainable Energy, 3(3), 330–338.

Khatib, T., Mohameda, A., & Sopian, K. (2012). Optimization of a PV/wind micro-grid for rural housing electrification using a hybrid iterative/genetic algorithm: Case study of Kuala Terengganu, Malaysia. Energy and Buildings, 47, 321–333.

Abbes, D., Martinez, A., & Champions, G. (2014). Life cycle cost embodied energy and loss of power supply probability for the optimal design of hybrid power systems. Mathematics and Computers in Simulation, 98, 46–62.

Anayochukwu, A. V., & Nnene E. A. (2013). Simulation and optimization of photovoltaic/diesel hybrid power generation systems for health service facilities in rural environments. Electronic Journal of Energy & Environment, 1(1), 57–70.

Lal, D. K., Dash, B. B., & Akella, A. (2011). Optimization of PV/wind/micro-hydro/diesel hybrid power system in Homer for the study area. International Journal on Electrical Engineering and Informatics, 3(3), 307–325.

Amer, M., Namaane, A., & M'Sirdi, N. K. (2013). Optimization of hybrid renewable energy systems (HRES) using PSO for cost reduction. Energy Procedia, 42, 318–327.

Bilal, B. O., Sambou, V., Kebe, C. M. F., Ndiaye, P. A., & Ndongo, M. (2012). Methodology to size an optimal stand-alone PV/wind/diesel/battery system minimizing the levelized cost of energy and the CO2 emissions. Energy Procedia, 14, 1636–1647.

Olatomiwa, L., Mekhilef, S., Huda, A. S. N., & Ohunakin, O. S. (2015). Economic evaluation of hybrid energy systems for rural electrification in six geopolitical zones of Nigeria. Renewable Energy, 83, 435–446.

Lujano-Rojas, J. M., Dufo-Lopez, R., & Bernal-Agustín, J. L. (2014). Technical and economic effects of charge controller operation and columbic efficiency on stand-alone hybrid power system. Energy Conversion and Management, 86, 709–716.

Hongxing, Y., Zhou, W., & Chengzhi, L. (2009) Optimal design and techno-economic analysis of a hybrid solar–wind power generation system. Applied Energy, 86, 163–169.

Kaabeche, A., Belhamel, M., & Ibtiouen, R. (2011) Sizing optimization of grid-independent hybrid photovoltaic/wind power generation system. Energy, 36, 1214–1222.

Shadmand, M. B., & Balog, R. S. (2014). Multi-objective optimization and design of photovoltaic wind hybrid system for community smart DC micro-grid. IEEE Transactions on Smart Grid, 5(5), 2635–2643.

Shivarama Krishna, K., & Sathish Kumar, K. (2015). A review on hybrid renewable energy systems. Renewable and Sustainable Energy Reviews, 52, 907–916.

Sinha, S., & Chandel, S. S. (2014). Review of software tools for hybrid renewable energy systems. Renewable and Sustainable Energy Reviews, 32, 192–205.

Kaur, D., & Cheema, P. S. (2017). Software tools for analyzing the hybrid renewable energy sources: a review. In: International Conference on Inventive Systems and Control (ICISC), 1–4.

Dynamic Modeling and Performance Analysis of Switched-Mode Controller for Hybrid Energy Systems

Abdallah, L., & El-Shennawy, T. (2013). Reducing carbon dioxide emissions from electricity sector using smart electric grid applications. Journal of Engineering. https://doi.org/10.1155/2013/845051

Abdelkader, H. I., Hatata, A. Y., & Hasan, M. S. (2015). Developing intelligent MPPT for PV systems based on ANN and P&O algorithms. International Journal of Scientific & Engineering Research, 6(2), 367–373.

Ahuja, R. K., & Kumar, R. (2014). Design and simulation of fuzzy logic controller based switched-mode power supply. IPASJ International Journal of Electrical Engineering, 2(5), 16–21.

Alward, Y. M. S., & Joshi, S. N. (2018). A review on control and automation based smart grid system and its impact on conventional grid. International Journal of Engineering Research & Technology, 7(04), 404–407.

Anoune, K., Bouya, M., Ghazouani, M., Astito, A., & Abdellah, A. B. (2016). Hybrid renewable energy system to maximize the electrical power production. In: 2016 International Renewable and Sustainable Energy Conference (IRSEC), Marrakech, pp. 533–539. https://doi.org/10.1109/IRSEC.2016.7983992

Bhandari, B., Lee, K. T., Lee, G. Y., Cho, Y. M., & Ahn, S. H. (2015). Optimization of hybrid renewable energy power systems: A review. International Journal of Precision Engineering and Manufacturing-Green Technology, 2(1), 99–112. https://doi.org/10.1007/s40684-015-0013-z Diaf, S., Notton, G., Belhamel, M., Haddadi, M., & Louche, A. (2008). Design and techno-economical optimization for hybrid PV/wind system under various meteorological conditions. Applied Energy, 85(10), 968–987. https://doi.org/10.1016/j.apenergy.2008.02.012

Naik, M. D., Sreenivasulu, R. D., & Devaraju, T. (2014). Dynamic modeling, control and simulation of a wind and PV hybrid system for grid connected application using MATLAB. International Journal of Engineering Research and Applications, 4(7), 131–139.

Meziane, S., Feddaoui, O., Toufouti, R., & Ahras, S. (2013). Modeling and simulation of hybrid wind-diesel power generation system. International Journal of Renewable Energy, 8(2), 49–58. https://doi.org/10.14456/iire.2013.11

Ezema, L. S., Peter, B. U., & Harris, O. O. (2012). Design of automatic change over switch with generator control mechanism. Part 1. Natural and Applied Sciences, 3(3), 125–130.

Vera,Y. E. G., Dufo-López,R., & Bernal-Augustin, J. L. (2019). Energy management in microgrids with renewable energy sources: A literature review. Applied Sciences, 9(18), 3854. Ramakumar, R. (2004). Role of renewable energy in the development and electrification of remote and rural areas. In: IEEE Power Engineering Society General Meeting, Denver, CO, pp. 2103–2105. https://doi.org/10.1109/pes.2004.1373253

Kaygusuz, K. (2009). Environmental impacts of the solar energy systems. Energy Sources, Part A: Recovery, Utilization, and Environmental Effects, 31(15), 1376–1386.

https://doi.org/10.1080/15567030802089664

Kermadi, M., & Berkouk, E. M. (2017). Artificial intelligence-based maximum power point tracking controllers for photovoltaic systems: Comparative study. Renewable and Sustainable Energy Reviews, 69, 369–386. https://doi.org/10.1016/j.rser.2016.11.125

Negi, S., & Mathew, L. (2014). Hybrid renewable energy system: A review. International Journal of Electronic and Electrical Engineering, 7(5), 535–542.

Prajapati, V., Patel, S., Thakor, P., & Chaudhary, T. (2018). Modelling and simulation of solar PV and wind hybrid power system using Matlab/Simulink. International Research Journal of Engineering and Technology, 5(4), 619–623.

Su, W., & Wang, J. (2012). Energy management systems in microgrid operations. Electricity Journal, 25(8), 45–60. https://doi.org/10.1016/j.tej.2012.09.010

Tous, Y. E., Hafith S. A., & Arabia, S. (2014). Photovoltaic/wind hybrid off-grid simulation model using Matlab/Simulink. International Journal of Latest Research in Science and Technology, 3(2), 167–173.

Allani, M. Y., Jomaa, M., Mezghani, D., & Mami, A. (2018). Modelling and simulation of the hybrid system PV-wind with MATLAB/SIMULINK. In: 2018 9th International Renewable Energy Congress (IREC), Hammamet, pp. 1–6. https://doi.org/10.1109/IREC.2018.8362514 Sajadian, S., & Ahmadi, R. (2016). High performance model predictive technique for MPPT of gird-tied photovoltaic system using impedance-source inverter. In: 2016 IEEE Power and

Energy Conference at Illinois (PECI), Urbana, IL, pp. 1–7.

https://doi.org/10.1109/PECI.2016.7459236

Bakić, V., Pezo, M., Stevanović, Ž., Živković, M., & Grubor, B. (2012). Dynamical simulation of PV/wind hybrid energy conversion system. Energy, 45, 324–328. https://doi.org/10.1016/j.energy.2011.11.063

Artificial Intelligence and Machine Learning Methods for Renewable Energy

Artificial intelligence in renewable energy. Available online at:

https://www.frontiersin.org/research-topics/12362/artificial-intelligence-in-renewable-energy The role of artificial intelligence in renewable energy. Available online at:

https://www.imaginovation.net/blog/artificial-intelligence-in-renewable-energy/

Meet A. Moradiya (2019). The challenges renewable energy face. Available online at: https://www.azocleantech.com/article.aspx?ArticleID=836

M. Hanan, X. Ai, M. Y. Javed, M. Majid Gulzar and S. Ahmad (2018). A two-stage algorithm to harvest maximum power from photovoltaic system. In: Energy Internet and Energy System Integration (EI2) IEEE Conference, pp. 1–6.

S. Suyanto, L. Mohammad, I. C. Setiadi and R. Roekmono (2019). Analysis and evaluation performance of MPPT algorithms: Perturb & observe (P&O), firefly, and flower pollination (FPA) in smart microgrid solar panel systems. In: 2019 International Conference on Technologies and Policies in Electric Power & Energy, pp. 1–6.

P. Dimitroulis and M. Alamaniotis (2020). Residential energy management system utilizing fuzzy based decision-making. In: 2020 IEEE Texas Power and Energy Conference (TPEC), pp. 1–6. Naveen and A. K. Dahiya (2018). Implementation and comparison of perturb & observe, ANN and ANFIS based MPPT techniques. In: Inventive Research in Computing Applications International Conference, pp. 1–5.

M. A. Samadhan and S. S. Kamble (2020). Introduction of different maximum power point tracking method using photovoltaic systems. In: Advanced Computing and Communication Systems International Conference (ICACCS), pp. 752–755.

E. H. M. Ndiaye, A. Ndiaye, M. A. Tankari and G. Lefebvre (2018). Adaptive neuro-fuzzy inference system application for the identification of a photovoltaic system and the forecasting of its maximum power point. In: Renewable Energy Research and Applications Conference, pp. 1061–1067.

P. Vinay and M. A. Mathews (2014). Modelling and analysis of artificial intelligence based MPPT techniques for PV applications. In: Advances in Green Energy International Conference, pp. 56–65.

M. Aurangzeb, X. Ai, M. Hanan, M. U. Jan, H. U. Rehman and S. Iqbal (2019). Single algorithm Mpso depend solar and wind MPPT control and integrated with fuzzy controller for grid integration. In: 2019 IEEE 3rd Conference on Energy Internet and Energy System Integration (El2), pp. 583–588.

S. B. Raha and D. Biswas (2020). Fuzzy controlled demand response energy management for economic micro grid planning. In: Power Electronics, Smart Grid and Renewable Energy IEEE Conference (PESGRE), pp. 1–6.

T. M. Sanjeev Kumar, S. Sharma, C. P. Kurian, S. M. Varghese and A. M. George (2020). Adaptive neuro-fuzzy control of solar-powered building integrated with daylight-artificial light system. In: 2020 IEEE International Conference on Power Electronics, Smart Grid and Renewable Energy Conference (PESGRE2020).

K. Geetha and B. Sangeetha (2013). Application of artificial intelligent technique to power quality issues in renewable energy sources. In: Renewable Energy and Sustainable Energy International Conference, pp. 233–237.

Renewable energy data set. Available online at:

http://s39624.mini.alsoenergy.com/Dashboard/2a5669735064572f4342554b772b71513d

Artificial Neural Network-Based Power Optimizer for Solar Photovoltaic System

Khalil, T. M. (1981). Comparative analysis of energy resources. International Journal of Production Research, *19*(4), 401–409.

Jin, T., Shi, T., & Park, T. (2018). The quest for carbon-neutral industrial operations: Renewable power purchase versus distributed generation. International Journal of Production Research, *56* (17), 5723–5735. DOI: 10.1080/00207543.2017.1394593

Sedaghati, F., Nahavandi, A., Badamchizadeh, M. A., Ghaemi, S., & Fallah, M. A. (2012). PV maximum power-point tracking by using artificial neural network. Mathematical Problems in Engineering. https://doi.org/10.1155/2012/506709.

Motahhir, S., El Ghzizal, A., Sebti, S., & Derouich, A. (2018). Modeling of photovoltaic system with modified incremental conductance algorithm for fast changes of irradiance. International Journal of Photoenergy. https://doi.org/10.1155/2018/3286479.

Christopher, I. W., & Ramesh, R. (2013). Comparative study of P&O and InC MPPT algorithms. American Journal of Engineering Research, *2*(12), 402–408.

Sreedhar, S., & Jagadeesh, D. (2016). A review on optimization algorithms for MPPT in solar PV system under partially shaded conditions. IOSR Journal of Electrical and Electronics Engineering, *1*, 23–32 (Two day National Conference on "SMart grid And Renewable Technologies" (SMART-2016).

Ezinwanne, O., Zhongwen, F., & Zhijun, L. (2017). Energy performance and cost comparison of MPPT techniques for photovoltaics and other applications. Energy Procedia, *107*, 297–303. https://doi.org/10.1016/j.egypro.2016.12.156 Babaie, M., Sebaaly, F., Sharifzadeh, M., Kannan, H. Y., & Al-Haddad, K. (2019). Design of an artificial neural network control based on Levenberg-Marquart algorithm for grid connected packed U-Cell Inverter. In: *2019 IEEE International Conference on Industrial Technology (ICIT)*, Melbourne, 1202–1207.

Rai, A. K., Kaushika, N. D., Singh, B., & Agarwal, N. (2011). Simulation model of ANN based maximum power point tracking controller for solar PV system. Solar Energy Materials and Solar Cells, *95*(2), 773–778. https://doi.org/10.1016/j.solmat.2010.10.022

Hadji, S., Gaubert, J. P., & Krim, F. (2015). Theoretical and experimental analysis of genetic algorithms based MPPT for PV systems. Energy Procedia, *74*, 772–787. https://doi.org/10.1016/j.egypro.2015.07.813

Bouselham, L., Hajji, M., Hajji, B., & Bouali, H. (2017). A new MPPT-based ANN for photovoltaic system under partial shading conditions. Energy Procedia, *111*, 924–933.

https://doi.org/10.1016/j.egypro.2017.03.255

Balasubramanian, G., & Singaravelu, S. (2012). Fuzzy logic controller for the maximum power point tracking in photovoltaic system. International Journal of Computer Applications, *41*(12), 22–28. https://doi.org/10.5120/5594-7840

Rezvani, A., Izadbakhsh, M., Gandomkar, M., & Vafaei, S. (2015). Implementing GA-ANFIS for maximum power point tracking in PV system. Indian Journal of Science and Technology, *8*(10), 982–991. https://doi.org/10.17485/ijst/2015/v8i10/51832

Paul, S., & Thomas, J. (2014). Comparison of MPPT using GA optimized ANN employing PI controller for solar PV system with MPPT using incremental conductance. In: *2014 International Conference on Power Signals Control and Computations, EPSCICON 2014*, Thrissur, 1–5. https://doi.org/10.1109/EPSCICON.2014.6887518

Sarenyadhevi, K., & Rajasekaran, N. (2015). An intelligent approach for MPP tracking in solar panel using artificial neural network and genetic algorithm. *International Journal of Research and Engineering*, *2*(4), 37–40.

Prasad, L. B., Sahu, S., Gupta, M., Srivastava, R., Mozhui, L., & Asthana, D. N. (2017). An improved method for MPPT using ANN and GA with maximum power comparison through perturb & observe technique. In: *2016 IEEE Uttar Pradesh Section International Conference on Electrical, Computer and Electronics Engineering, UPCON 2016*, Varanasi, 206–211. https://doi.org/10.1109/UPCON.2016.7894653

Predictive Maintenance

Yaguo Lei, Feng Jia, Jing Lin, Saibo Xing, Steven X. Ding. (2016). An intelligent fault diagnosis method using unsupervised feature learning towards mechanical big data. IEEE Transactions on Industrial Electronics, 63(5), 3137–3147.

Jia Huang, Zhaojun Li, Hu-Chen Liu. (2017). New approach for failure mode and effect analysis using linguistic distribution assessments and TODIM method. Reliability Engineering & System Safety, 167, 302–309.

K. D. Sharma, S. Srivastava. (2018). Failure mode and effect analysis (FMEA) implementation: A literature review. Journal of Advance Research in Aeronautics and Space Science, 5(1), 1–17.

Z. Bluvband, P. Grabov. (2009). Failure Analysis of FMEA. 2009 Annual Reliability and Maintainability Symposium, Fort Worth, Texas, 344–347. doi: 10.1109/RAMS.2009.4914700. Jui-Sheng Chou, Chien-Kuo Chiu, I-Kui Huang, Kai-Ning Chi. (2013), Failure analysis of wind turbine blade under critical wind loads. Engineering Failure Analysis, 27, 99–118.

Evrencan Özcan, Rabia Yumuşak, Tamer Eren. (2019). Risk based maintenance in the hydroelectric power plants. Energies, 12, 1502; doi:10.3390/en1208150.

S. Karthikeyan, M. Subramaniyam, A. Ghosh, M. Prashanth, R. Karunanithi. (2019). Micro hardness and corrosion properties of A390 alloy + x vol.% zirconium dioxide composites processed by P/M method. International Journal of Microstructure and Material Properties, 14(6), 511–523.

Manju Santhakumari, Netramani Sagar. (2019). A review of the environmental factors degrading the performance of silicon wafer-based photovoltaic modules: Failure detection methods and essential mitigation techniques. Renewable and Sustainable Energy Reviews, 110, 83–100.

Hamid Reza Feili, Navid Akar, Hossein Lotfizadeh, Mohammad Bairampour, Sina Nasiri. (2013). Risk analysis of geothermal power plants using failure modes and effects analysis (FMEA) technique. Energy Conversion and Management, 72, 69–76.

Shikun Cheng, Zifu Li, Heinz-Peter Mang, Kalidas Neupane, Marc Wauthelet, Elisabeth-Maria Huba. (2014). Application of fault tree approach for technical assessment of small-sized biogas systems in Nepal. Applied Energy, 113, 1372–1381.

E. Zio. (2012). Diagnostics and Prognostics of Engineering Systems: Methods and Techniques. IGI Global, Chap. 17.

Sule Selcuk. (2017). Predictive maintenance, its implementation and latest trends. Journal of Engineering Manufacture, 231(9), 1–10.

Thyago P. Carvalho, Fabrízzio A. A. M. N. Soaresa, Roberto Vita, Roberto da P. Francisco, João P. Basto, Symone G. S. Alcalá. (2019). A systematic literature review of machine learning methods applied to predictive maintenance. Computers & Industrial Engineering, 137, 106024. Alessandro Betti, Maria Luisa Lo Trovato, Fabio Salvatore Leonardi, Giuseppe Leotta, Fabrizio Ruffini, Ciro Lanzetta. (2017). Predictive Maintenance in Photovoltaic Plants with a Big Data Approach. 33rd European Photovoltaic Solar Energy Conference and Exhibition (EUPVSEC), 1895–1900.

Timo Huuhtanen, Alexander Jung. (2018). Predictive Maintenance of Photovoltaic Panels via Deep Learning. IEEE Data Science Workshop (DSW), Lausanne, 66–70. doi: 10.1109/DSW.2018.8439898.

Aicha Djalab, Ahmed Hafaifa, Mohamed Mounir Rezaoui, Ali Teta, Nassim Sabri. (2019). An Intelligent Faults Diagnosis and Detection Method Based an Artificial Neural Networks for Photovoltaic Array. The 3rd International Conference on Applied Automation and Industrial Diagnostics, Elazig, Turkey, 1–6.

Massimiliano De Benedetti, Fabio Leonardi, Fabrizio Messina, Corrado Santoro, Athanasios Vasilakos. (2018). Anomaly detection and predictive maintenance for photovoltaic systems. Neurocomputing, 310, 59–68.

Thomas Kenbeek, Stella Kapodistria, Alessandro Di Bucchianico. (2019). Data-Driven Online Monitoring of Wind Turbines. Proceedings of the 12th EAI International Conference on Performance Evaluation Methodologies and Tools, 143–150.

K. Kim, G. Parthasarathy, O. Uluyol, W. Foslien. (2011). Use of SCADA Data for Failure Detection in Wind Turbines. Energy Sustainability Conference and Fuel Cell Conference, 1–9. K. Wang, V. S. Sharma, Zhang, Z. (2014). SCADA data based condition monitoring of wind turbines. Advanced Manufacturing, 2, 61–69.

Mari Cruz Garcia, Miguel A. Sanz-Bobi, Javier del Pico. (2006). SIMAP: Intelligent system for predictive maintenance application to the health condition monitoring of a windturbine gearbox. Computers in Industries, 57(6), 552–568.

Bindi Chen, Peter J. Tavner, Yanhui Feng, William W. Song, Yingning Qiu. (2012). Bayesian Networks for Wind Turbine Fault Diagnosis. EWEA 2012, Copenhagen, 16–19 April. Bindi Chen, Peter C. Matthews, Peter J. Tavner. (2013). Wind turbine pitch faults prognosis using a-priori knowledge-based ANFIS. Expert Systems with Applications, 40, 6863–6876. Nassim Laouti, Nida Sheibat-Othman, Sami Othman. (2011). Support Vector Machines for Fault Detection in Wind Turbines. Proceedings of the 18th IFAC World Congress, Milan, 7067–7072. Z. Liu, S. Zou, L. Zhou. (2012). Condition Monitoring System for Hydro Turbines Based on LabVIEW. Asia-Pacific Power and Energy Engineering Conference, Shanghai, 1–4.

W. Jiang. (2008). Research on Predictive Maintenance for Hydropower Plant Based on MAS and NN. Third International Conference on Pervasive Computing and Applications, Alexandria, 604–609.

R. Yam, P. Tse, L. Li. (2001). Intelligent predictive decision support system for condition-based maintenance. International Journal of Advanced Manufacturing Technology, 17, 383–391.
E. J. Amaya, A. J. Alvares. (2010). SIMPREBAL: An expert system for real-time fault diagnosis of hydrogenerators machinery. IEEE 15th Conference on Emerging Technologies & Factory Automation, 1–8.

Xiaolian Liu, Yu Tian, Xiaohui Lei, Mei Liu, Xin Wen, Haocheng Huang, Hao Wang. (2019). Deep forest based intelligent fault diagnosis of hydraulic turbine. Journal of Mechanical Science and Technology, 33(5), 2049–2058.

L. Han, Z. Li. (2009). Research of Cloud Neural Network Based on Cloud Transformation and Its Application on Vibration Fault Diagnosis of Hydro-Turbine Generating Unit. Asia-Pacific Power and Energy Engineering Conference, 1–4.

Dong Liang, HuaShan Guo, ZeWei Guo, Tao Zheng. (2019). Fault diagnosis technique for hydroelectric generators using variational mode decomposition and power line communications. Journal of Physics: Conference Series, 1176(6), 1–6.

Man Zhou, Zhiyong Zou. (2018). Design of an Intelligent Control System for Rural Biogas Engineering. 2nd IEEE Advanced Information Management, Communicates, Electronic and Automation Control Conference (IMCEC), Xi'an, 1636–1639.

Djordje Djatkov, Mathias Effenberger, Milan Martinov. (2014). Method for assessing and improving the efficiency of agricultural biogas plants based on fuzzy logic and expert systems. Applied Energy, 134, 163–175.

XiaoJuan Wu, Qianwen Ye. (2016). Fault diagnosis and prognostic of solid oxide fuel cells. Journal of Power Sources, 321, 47–56.

Hector Alaiz-Moreton, Manuel Castejón-Limas, Jose-Luis Casteleiro-Roca, Esteban Jove, Laura Fernandez Robles, Jose Luis Calvo-Rolle. (2019). A fault detection system for a geothermal heat exchanger sensor based on intelligent techniques. Sensors, 19, 1–16.

M. Fast, T. Palme. (2010). Application of artificial neural networks to the condition monitoring and diagnosis of a combined heat and power plant. Energy, 35, 1114–1120.

José Luis Casteleiro-Roca, Héctor Quintián, José Luis Calvo-Rolle, Emilio Corchado, María del Carmen Meizoso-López, Andrés Piñón-Pazos. (2015). An intelligent fault detection system for a heat pump installation based on a geothermal heat exchanger. Journal of Applied Logic, 17, 36–47.

Zulkarnain, I. Surjandari, R. R. Bramasta, E. Laoh. (2019). Fault Detection System Using Machine Learning on Geothermal Power Plant. 16th International Conference on Service Systems and Service Management (ICSSSM), Shenzhen, 1–5.

S. Sharanya, Revathi Venkataraman. (2020). An intelligent context based multi-layered Bayesian inferential predictive analytic framework for classifying machine states. Journal of Ambient Intelligence and Humanized Computing. https://doi.org/10.1007/s12652-020-02411-2.

AI Techniques for the Challenges in Smart Energy Systems

Cheng, Lefeng, Yu, Tao. "A New Generation of AI: A Review and Perspective on Machine Learning Technologies Applied to Smart Energy and Electric Power Systems." International Journal of Energy Research 43 no. 6 (2019): 1928–1973.

Kalogirou, A. Soteris, Şencan, Arzu. "Artificial Intelligence Techniques in Solar Energy Applications." Solar Collectors and Panels, Theory and Applications (2010). DOI: 10.5772/10343.

Pearson, J. G. Peter, Foxon, J. Timothy. "A Low Carbon Industrial Revolution? Insights and Challenges from Past Technological and Economic Transformations." Energy Policy 50 (2012): 117–127.

Li, Fan, Peng, Xiaoqi, Wang, Zuo, Zhou, Yi Wang, Wu, Yuxia, Jiang, Minlin, Xu, Min. "Machine Learning (ML) – Assisted Design and Fabrication For Solar Cells." Energy & Environmental Materials 9 (2019): 280–291.

Lachmann, Malin, Maldonado, Jaime, Wiebke, Bergmann, Jung, Francesca, Weber, Markus, Büskens, Christof. "Self-Learning Data-Based Models as Basis of A Universally Applicable Energy Management System." Energies 13 no. 2084 (2020): 1–42.

Ascencio-Vàsquez, Juliàn, Bevc, Jakob, Reba, Kristjan, Brecl, Kristijan, Jankovec, Marko, Topič, Marko. "Advanced PV Performance Modelling Based on Different Levels of Irradiance Data Accuracy." Energies 13 no. 2166 (2020): 1–12.

Bose, B. K. "Artificial Intelligence Applications in Renewable Energy Systems and Smart Grid – Some Novel Applications." Power Electronics in Renewable Energy Systems and Smart Grid: Technology and Applications (2019): 625–675. DOI:

https://doi.org/10.1002/9781119515661.ch12.

Xu, Yueqiang, Ahokangas, Petri, Louis, Jean-Nicolas, Pongràcz, Eva. "Electricity Market Empowered by Artificial Intelligence: A Platform Approach." Energies 12 no. 4128 (2019): 1–21. Jang, Roger Jyh-Shing. "ANFIS: Adaptive-Network based Fuzzy Inference System." IEEE Transactions on Systems, Man, and Cybernetics 23 no. 3 (1993): 665–685. Bose, B. K. "Expert System, Fuzzy Logic, and Neural Network Applications in Power Electronics and Motion Control." 82 no. 8 (1994): 1303–1323.

Gündoğdu, Fatma Kutlu, Cengiz, Kahraman. "Spherical Fuzzy Sets and Spherical Fuzzy TOPSIS Method." Journal of Intelligent and Fuzzy Systems 36 no. 1 (2019): 337–352. Duojie, Jia-hua, Haidong, Zhang, Yanping, He. "Possibility Pythagorean Fuzzy Soft Set and Its

Application." Journal of Intelligent and Fuzzy Systems 36 no. 1 (2019): 413-421.

Kumanan, S., Jesuthanam, C. P., Ashok Kumar, R. "Application of Multiple Regression and Adaptive Neuro Fuzzy Inference System for the Prediction of Surface Roughness." International Journal of Advanced Manufacturing Technology 35 (2008): 778–788.

Wieprecht, Silke, Tolossa, Habtamu G., Yang, Chih Ted. "A Neuro-Fuzzy-Based Modelling Approach for Sediment Transport Computation." Hydrological Sciences Journal 58 no. 3 (2013): 587–599.

Savkovic, Borislav, Kovac, Pavel, Dudic, Branislav, Rodic, Dragan, Taric, Mirfad, Gregus, Michal. "Application of an Adaptive "Neuro-Fuzzy" Inference System in Modeling Cutting Temperature during Hard Turning." Applied Sciences 9 no. 3739 (2019): 1–13.

Takagi, T., Sugeno, M. "Fuzzy Identification of Systems and Its Applications to Modeling and Control." IEEE Transaction on Systems, Man, and Cybernetics 15 (1985): 116–132.

Toosi, Nadrajan Adel, Kahani, Mohsen. "A New Approach to Intrusion Detection Based on an Evolutionary Soft Computing Model Using Neuro-Fuzzy Classifiers." Computer Communications 30 (2007): 2201–2212.

Zadeh, L. A. "Calculus of Fuzzy Restrictions." Fuzzy Sets and Their Applications to Cognitive and Decision Process. Proceedings of The US-Japan Seminar on Fuzzy Sets and Their Applications held at The University of California, Berkeley, California, USA (1974, 1975): 1–39. Huynh, V. N., Ho, T. B., Nakamori, Y. "A Parametric Representation of Linguistic Hedges in Zadeh's Fuzzy Logic." International Journal of Approximate Reasoning 30 (2002): 203–223. Lo, S.-P. "An Adaptive-Network Based Fuzzy Inference System for Prediction of Workpiece Surface Roughness in End Milling." Journal of Materials Processing Technology 142 no. 10 (2003): 665–675.

Boyacioglu, M. A., Avci, D. "An Adaptive Network-based Fuzzy Inference System (ANFIS) for the Prediction of Stock Market Return: The Case of Istanbul Stock Exchange." Expert Systems with Applications 37 (2010): 7908–7912.

Cross, Valerie, Sudkamp, Thomas. "Patterns of Fuzzy Rule-Based Inference." International Journal of Approximate Reasoning 11 no. 3 (1994): 235–255.

Ishibuchi, H., Nakashima, T., Nii M. "Fuzzy If-Then Rules for Pattern Classification." In Ruan, D., Kerre E. E. (eds.) Fuzzy If-Then Rules in Computational Intelligence. (The Springer International Series in Engineering and Computer Science Springer) Springer, Berlin Heidelberg, p. 533 (2000): 267–295.

Mamdani, E. H., Assilian, S. "An Experiment in Linguistic Synthesis with a Fuzzy Logic Controller." International Journal of Man-Machine Studies 7 no. 1 (1975): 1–13.

Ying Cui, Qi Guo, Jacqueline P. Leighton, Man-Wai Chu. "Log Data Analysis with ANFIS: A Fuzzy Neural Network Approach." International Journal of Testing 20 no. 1 (2020): 78–96. Zhang, Guoqiang, Patuwo, Eddy B., Hu, Y. Michael. "Forecasting with Artificial Neural Networks: The State of the Art." International Journal of Forecasting 14 (1998): 35–62. Huang, Yiqun, Kangas, J. Lars, Rasco, A. Barbara. "Applications of Artificial Neural Networks (ANNs) in Food Science." Critical Reviews in Food Science and Nutrition 47 no. 2 (2007): 113–126.

Artificial Neural Networks. In Xingui, He, Shaohua, Xu, Zheng, Zibin, Lyu, R. Michael (eds.) Process Neural Networks. Advanced Topics in Science and Technology in China. Springer, Berlin Heidelberg (2009): 20–42.

Heyn, Jakob, Gümbel, Philip, Bobka, Paul, Dietrich, Franz, Dröder, Klaus. "Application of Artificial Neural Network in Force-Controlled Automated Assembly of Complex Shaped Deformable Components." Procedia CIRP 79 (2019): 131–136.

Srinivas, M., Patnaik, L. M. "Genetic Algorithms: A Survey." Computer 27 no. 6 (1994): 17–26. McCall, John. "Genetic Algorithms for Modelling and Optimisation." Journal of Computational and Applied Mathematics 184 no. 1 (2005): 205–222.

Chaudhry, S.S., Luo, W. "Application of Genetic Algorithms in Production and Operations Management: A Review." International Journal of Production Research 43 no. 19 (2005): 4083–4101.

Jennings, P.C., Lysgaard, S., Hummelshøj, J.S. "Genetic Algorithms for Computational Materials Discovery Accelerated by Machine Learning. npj Computational Materials 5 no. 46 (2019): 1–6.

Doganaksoy, N., Hahn, J. G. "Data Mining: A Gateway to Better Data Gathering." Statistical Analysis and Data Mining: The ASA Data Science 1 no. 4 (2009): 280–283.

Goodman, Arnold. "Emerging Topics and Challenges for Statistical Analysis and Data Mining." Statistical Analysis and Data Mining: The ASA Data Science 4 no. 1 (2011): 3–8.

Hassani, H., Huang, X., Silva, S. E., Ghodsi, M. "A Review of Data Mining Applications in Crime." Statistical Analysis and Data Mining: The ASA Data Science 9 no. 3 (2016): 139–154. Hoffmann, D., Apostolakis, J. "Crystal Structure Prediction by Data Mining". Journal of Molecular Structure 647 (2003): 17–39.

Salah, C. B., Chaabene, M., Ammar, M. B. "Multi-Criteria Fuzzy Algorithm for Energy Management of a Domestic Photovoltaic Panel." Renewable Energy 33 (2008): 993–1001. Altas, I. H., Sharaf, A. M. "A Novel Maximum Power Fuzzy Logic Controller for Photovoltaic Solar Energy Systems". Renewable Energy 33 (2008): 388–399.

Gomez, V., Casanovas, A. "Fuzzy Logic and Meteorological Variables: A Case Study of Solar Irradiance." Fuzzy Sets and Systems 126 (2002): 121–128.

Gomez, V., Casanovas, A. "Fuzzy Modelling of Solar Irradiance on Inclined Surfaces." Solar Energy 75 (2003): 307–315.

Paulescu, M., Gravila, P., Tulcan-Paulescu, E. "Fuzzy Logic Algorithms for Atmospheric Transmittances of Use in Solar Energy Estimation." Energy Conversion and Management 49 (2008): 3691–3697.

Şen, Z. "Fuzzy Algorithm for Estimation of Solar Irradiation from Sunshine Duration." Solar Energy 63 (1998): 39–49.

Gouda, M. M., Danaher, S., Underwood C. P. "Quasi-Adaptive Fuzzy Heating Control of Solar Buildings." Building and Environment 41 (2006): 1881–1891.

Lygouras, J. N., Botsaris, P. N., Vourvoulakis, J., Kodogiannis, V. "Fuzzy Logic Controller Implementation for a Solar Air-Conditioning System." Applied Energy 84 (2007): 1305–1318. Mellit, A., Kalogirou, S. A., Shaari, S., Salhi, H., Hadj Arab, A. "Methodology for Predicting Sequences of Mean Monthly Clearness Index and Daily Solar Radiation Data in Remote Areas: Application for Sizing A Stand-Alone PV System." Renewable Energy 33 (2008): 1570–1590. Rehman, S., Mohandes, M. "Artificial Neural Network Estimation of Global Solar Radiation Using Air Temperature and Relative Humidity." Energy Policy 36 (2008): 571–576.

Mellit, A., Pavan A. M. "A 24-h Forecast of Solar Irradiance Using Artificial Neural Network: Application for Performance Prediction of a Grid-Connected PV Plant at Trieste, Italy. Solar Energy 84 no. 5 (2010): 807–821.

Benghanem, M., Mellit, A., Alamri S. N. "ANN-Based Modelling and Estimation of Daily Global Solar Radiation Data: A Case Study." Energy Conversion and Management 50 no. 7 (2009): 1644–1655.

Mubiru, J., Banda, E. J. K. B. "Estimation of Monthly Average Daily Global Solar Irradiation Using Artificial Neural Networks." Solar Energy 82 (2008): 181–187.

Zagrouba, M., Sellami, A., Bouaicha, M., Ksouri, M. "Identification of PV Solar Cells and Modules Parameters Using the Genetic Algorithms: Application to Maximum Power Extraction." Solar Energy 84 no. 5 (2010): 860–866.

Loomans, M., Visser, H. "Application of the Genetic Algorithm for Optimisation of Large Solar Hot Water Systems." Solar Energy 72 (2002): 427–439.

Varun, S. "Thermal Performance Optimization of a Flat Plate Solar Air Heater Using Genetic Algorithm." Applied Energy 87 (2010): 1793–1799.

Gligor, A., Dumitru, C.-D., Grif, H.-S. "Artificial Intelligence Solution for Managing a Photovoltaic Energy Production Unit." Procedia Manufacturing 22 (2018): 626–633.

Kumaravel, S., Ashok, S. "Adapted Multilayer Feedforward ANN Based Power Management Control of Solar Photovoltaic and Wind Integrated Power System. In Innovative Smart Grid Technologies-India (ISGT India), 2011 IEEE PES (2011): 223–228.

Saberian, A., Hizam, H., Radzi, M. A. M., Ab Kadir, M. Z. A., Mirzaei, M. "Modelling and Prediction of Photovoltaic Power Output Using Artificial Neural Networks." International journal of Photoenergy 2014 (2014): 1–10.

Takagi, H., Hayashi, I. "NN-Driven Fuzzy Reasoning." International Journal of Approximate Reasoning 5 no. 3 (1991): 191–212.

Energy Efficiency

Diesendorf M. & Mark D. (2007) Greenhouse solutions with sustainable energy. University of New South Wales Press, Sydney, Australia.

Prindle B., Maggie E., Mike E. & Alyssa F. (2007). The twin pillars of sustainable energy: synergies between energy efficiency and renewable energy technology and policy. American Council for an Energy-Efficient Economy, Washington, DC.

Press Release. (2019). Vienna UN Conference Shows Consensus on Key Building Blocks for Effective International Response to Climate Change.

ISO 17742. (2015). Energy Efficiency and Savings Calculation for Countries, Regions and Cities. International Standards Association (ISO). Geneva, Switzerland. Retrieved 2016-11-11. Weinsziehr T. & Skumatz L. (2016). Evidence for Multiple Benefits or NEBs: Review on Progress and Gaps from the IEA Data and Measurement Subcommittee. In Proceedings of the International Energy Policy & Programme Evaluation Conference, Amsterdam, the Netherlands, 7–9.

International Energy Agency. (2014). Capturing the Multiple Benefits of Energy Efficiency. OECD, Paris.

International Energy Agency. (2007). Capturing the multiple benefits of energy efficiency. OECD, Paris.

https://cdn.pixabay.com/photo/2013/07/12/18/55/energy-efficiency-154006-960-720.png. McKinsey Global Institute. (2009). Pathways to a Low-Carbon Economy: Version 2 of the Global Greenhouse Gas Abatement Cost Curve. Retrieved February 16, 2016.

Environmental and Energy Study Institute. (2010). Energy-Efficient Buildings: Using Whole Building Design to Reduce Energy Consumption in Homes and Offices. Eesi.org. Retrieved 2010-07-16.

https://cdn.pixabay.com/photo/2019/04/03/12/04/home-4100193_960_720.jpg.

ENERGY STAR Buildings and Plants. (2019). Energystar.gov. Retrieved 2019-3-26.

What is artificial intelligence? (2016). Progressive Digital Media Technology News Retrieved from https://search.proquest.com/docview/1789725622?accountid=27468.

Elgendy N. & Ahmed E. (2014). "Big data analytics: a literature review paper." In Industrial conference on data mining, Springer, Cham, 214–227.

Emmanuel I. & Stanier C. (2016). Defining Big Data. In Proceedings of the International Conference on Big Data and Advanced Wireless Technologies (BDAW '16), Blagoevgrad, Bulgaria, 10–11 November 2016; ACM: New York, NY, USA, 5:1–5:6.

Moreno M. V., Dufour L., Skarmeta A. F., Jara A. J., Genoud D., Ladevie B. & Bezian J. J. (2016). Big data: the key to energy efficiency in smart buildings. Soft Computing, 20(5), 1749–1762.

https://cdn.pixabay.com/photo/2020/06/05/06/57/artificial-intelligence-5261742_960_720.jpg. O'Leary D. E. (2013). Artificial intelligence and big data. IEEE Intelligent Systems, 28(2), 96–99. Zheng N. N., Zi-Yi L., Peng-Ju R., Yong-Qiang M., Shi-Tao C., Si-Yu Y., Jian-Ru X., Ba-Dong C. & Fei-Yue W. (2017). Hybrid-augmented intelligence: collaboration and cognition. Frontiers of Information Technology & Electronic Engineering, 18(2), 153–179.

Louridas P. & Ebert C. (2016). Machine learning. IEEE Software, 10–115.

Goodfellow, Ian, Yoshua Bengio, & Aaron Courville. (2016). Deep learning. MIT Press, Cambridge, MA.

https://cdn.pixabay.com/photo/2018/04/07/09/48/hand-3298095_960_720.png

https://www.maximpact.com/artificial-intelligence-in-energy-efficiency/

https://cdn.pixabay.com/photo/2018/04/23/12/11/windmill-3344052_960_720.jpg

https://emerj.com/ai-sector-overviews/artificial-intelligence-for-energy-efficiency-and-renewable-energy/

Proctor, Cathy. (2018). Xcel's Forecast Rosy for Wind and Solar. Denver Business Journal, 4 May 2018, www.bizjournals.com/denver/news/2018/05/04/cover-storyxcel-s-forecast-rosy-forwind-and-solar.html

Verdigris. (2017). Fast Company. Retrieved 2017-3-14.

https://cdn.pixabay.com/photo/2019/05/03/11/40/smart-4175713_960_720.png Environmental and Energy Study Institute. (2015). Industrial Energy Efficiency: Using new technologies to reduce energy use in industry and manufacturing (PDF). Retrieved 2015-01-11. City of Edmonton (2019). Street Lighting. Edmonton.ca. Retrieved 2019-3-26. https://cdn.pixabay.com/photo/2020/06/01/13/23/workers-5246634_1280.jpg http://maximpactblog.com/ai-can-cut-energy-consumption/

Dorf R. C. (1981). Energy factbook. [includes glossary].

Höjer M. & Josefin W. (2015). "Smart sustainable cities: definition and challenges." In ICT innovations for sustainability, Springer, Cham, 333–349.

Sustainability & Energy Efficiency. (2011). Empire State Building, Esbnyc.com., 2011-06-16. Retrieved 2013-08-21.

Lovins A. B. (2012). A farewell to fossil fuels: answering the energy challenge. Foreign Affairs, 91, 134.

Tuominen P., Francesco R., Waled D., Bahaa E., Ghada E. & Abdelazim N. (2015). Economic appraisal of energy efficiency in buildings using cost-effectiveness assessment. Procedia Economics and Finance, 21, 422–430.

Heat Roadmap Europe. Heatroadmap.eu. Retrieved 2018-04-24.

Heinrich B. S. (2018). Energy atlas 2018. Figures and facts about renewables in Europe.

Renewable (Bio-Based) Energy from Natural Resources (Plant Biomass Matters)

Zafar, M. W., Shahbaz, M., Hou, F., & Sinha, A. (2019). From nonrenewable to renewable energy and its impact on economic growth: The role of research & development expenditures in Asia-Pacific Economic Cooperation countries. Journal of Cleaner Production, 212, 1166–1178. Sharif, A., Raza, S. A., Ozturk, I., & Afshan, S. (2019). The dynamic relationship of renewable and nonrenewable energy consumption with carbon emission: A global study with the application of heterogeneous panel estimations. Renewable Energy, 133, 685–691.

Wesseh, P. K., & Lin Jr., B. (2020). Energy substitution and technology costs in a transitional economy, Energy, 203, 117828.

Rudolfsson, M., Agar, D. A., Lestande, T. A., & Larsson, S. H. (2020). Energy savings through late-steam injection – A new technique for improving wood pellet production. Journal of Cleaner Production, 254, 120099.

Agyekum, E. B. (2020). Energy poverty in energy rich Ghana: A SWOT analytical approach for the development of Ghana's renewable energy. Sustainable Energy Technologies and Assessments, 40, 100760.

Atif, M., Hossain, M., Alam, M. S., & Goergen, M. (2021). Does board gender diversity affect renewable energy consumption? Journal of Corporate Finance, 66, 101665.

Kafka, J., & Miller, M. A. (2020). The dual angle solar harvest (DASH) method: An alternative method for organizing large solar panel arrays that optimizes incident solar energy in conjunction with land use. Renewable Energy, 155, 531–546.

Wang, Y., Fan, Y., Wang, D., Liu, Y., Qiu, Z., & Liu, J. (2020). Optimization of the areas of solar collectors and photovoltaic panels in liquid desiccant air-conditioning systems using solar energy in isolated low-latitude islands. Energy, 198, 117324.

Yamaki, A., Kanematsu, Y., & Kikuchi, Y. (2020). Lifecycle greenhouse gas emissions of thermal energy storage implemented in a paper mill for wind energy utilization. Energy (online 12 June 2020), 118056.

Caporale, D., Sangiorgio, V., Amodio, A., & Lucia, C.D. (2020). Multi-criteria and focus group analysis for social acceptance of wind energy. Energy Policy, 140, 111387.

Lafleur, C., Truelove, W. A., Cousine, J., Hiles, C. E., Buckham, B., & Crawford, C. (2020). A screening method to quantify the economic viability of off-grid in-stream tidal energy deployment. Renewable Energy (online 11 June 2020).

Longa, F. D., Nogueira, L. P., Limberger, J., van Wees, J. -D., & der Zwaan, B. (2020). Scenarios for geothermal energy deployment in Europe. Energy (online 10 June 2020), 118060. Yan. P., Xiao, C., Xu, L., Yu, G., Li, A., Piao, S., & He, N. (2020). Biomass energy in China's terrestrial ecosystems: Insights into the nation's sustainable energy supply. Renewable and Sustainable Energy Reviews, 127, 109857. Nassar, I. A., Hossam, K., & Abdella, M. M. (2019). Economic and environmental benefits of increasing the renewable energy sources in the power system. Energy Reports, 5, 1082–1088. Heng, Y., Lu, C.-L., Yu, L., & Gao, Z. (2020). The heterogeneous preferences for solar energy policies among US households. Energy Policy, 137, 111187.

Karytsas, S., & Theodoropoulou, H. (2014). Socioeconomic and demographic factors that influence publics' awareness on the different forms of renewable energy sources. Renewable Energy, 71, 480–485.

Ma, Z., Bao, H., & Roskilly, A. P. (2019). Seasonal solar thermal energy storage using thermochemical sorption in domestic dwellings in the UK. Energy, 166, 213–222.

Fthenakis, V., Mason, J. E., & Zweibel, K. (2009). The technical, geographical, and economic feasibility for solar energy to supply the energy needs of the US. Energy Policy, 37(2), 387–399. Almakrami, H., Wei, Z., Lin, G., Jin, X., Agar, E., & Liu, F. (2020). An integrated solar cell with built-in energy storage capability. Electrochimica Acta, 349, 136368.

Bahaja, A. B. S., Mahdy, M., Alghamdi, A. S., & Richards, D. J. (2020). New approach to determine the Importance Index for developing offshore wind energy potential sites: Supported by UK and Arabian Peninsula case studies. Renewable Energy, 152, 441–457.

Li, Y., Huang, X., Tee, K. F., Li, Q., & Wu, X.-P. (2020). Comparative study of onshore and offshore wind characteristics and wind energy potentials: A case study for southeast coastal region of China. Sustainable Energy Technologies and Assessments, 39, 100711.

Basheer, M., & Elagi, N. A. (2019). Temporal analysis of water-energy nexus indicators for hydropower generation and water pumping in the Lower Blue Nile Basin. Journal of Hydrology, 578, 124085.

Zhong, W., Guo, J., Chen, L., Zhou, J., & Zhang, J. (2020). Dangwei, future hydropower generation prediction of large-scale reservoirs in the upper Yangtze River basin under climate change. Journal of Hydrology, 588, 125013.

Avcıoğlu, A. O., Dayıoğlu, M. A., & Türker, U. (2019). Assessment of the energy potential of agricultural biomass residues in Turkey. Renewable Energy, 138, 610–619.

Zhang, Y., Qin, C., & Liu, Y. (2018). Effects of population density of a village and town system on the transportation cost for a biomass combined heat and power plant. Journal of Environmental Management, 223, 444–451.

Miao, Y., Bi, Q., Qin, H., Zhang, X., & Tan, N. (2020). Moderate drought followed by re-watering initiates beneficial changes in the photosynthesis, biomass production and Rubiaceae-type cyclopeptides (RAs) accumulation of *Rubia yunnanensis*. Industrial Crops and Products, 148, 112284.

Kumar, V., Kumar, P., Kumar, P., & Singh J., (2020). Anaerobic digestion of Azolla pinnata biomass grown in integrated industrial effluent for enhanced biogas production and COD reduction: Optimization and kinetics studies. Environmental Technology & Innovation, 17, 100627.

Iweh, N. S., Koyam, M., Akizuki, S., Ban, S., & Tod, T. (2020). Novel wet-solid states serial anaerobic digestion process for enhancing methane recovery of aquatic plant biomass. Science of the Total Environment, 730, 138993.

Kumari D., & Singh, R. (2018). Pretreatment of lignocellulosic wastes for biofuel production: A critical review. Renewable and Sustainable Energy Reviews, 90, 877–891.

Sewsynker-Sukai, Y., David, A. N., & Kan, E. B. G. (2020). Recent developments in the application of kraft pulping alkaline chemicals for lignocellulosic pretreatment: Potential beneficiation of green liquor dregs waste. Bioresource Technology, 306, 123225.

Javed, U., Ansari, A., Aman, A., & Qade, S. A. U. (2019). Fermentation and saccharification of agro-industrial wastes: A cost-effective approach for dual use of plant biomass wastes for xylose production. Biocatalysis and Agricultural Biotechnology, 21, 101341.

Fitria, Ruan, H., Fransen, S. C., Carter, A. H., Tao, H., & Yang, B. (2019). Selecting winter wheat straw for cellulosic ethanol production in the Pacific Northwest, U.S.A. Biomass and Bioenergy, 123, 59–69.

Li, J., Feng, P., Xiu, H., Li, J., Yang, X., Ma, F., Li, X., Zhang, X., Kozliak, E., & Ji, Y. (2019). Morphological changes of lignin during separation of wheat straw components by the hydrothermal-ethanol method. Bioresource Technology, 294, 122157.

Prasad, S., Malav, M. K., Kumar, S., Singh, A., Pant, D., & Radhakrishnand, S. (2018). Enhancement of bio-ethanol production potential of wheat straw by reducing furfural and 5hydroxymethylfurfural (HMF). Bioresource Technology Reports, 4, 50–56. Qi, G., Huang, D., Wang, J., Shen, Y., & Gao, X. (2019). Enhanced butanol production from ammonium sulfite pretreated wheat straw by separate hydrolysis and fermentation and simultaneous saccharification and fermentation. Sustainable Energy Technologies and Assessments, 36, 100549.

Ramirez, J. A., & Rainey, T. J. (2019). Comparative techno-economic analysis of biofuel production through gasification, thermal liquefaction and pyrolysis of sugarcane bagasse. Journal of Cleaner Production, 229, 513–527.

Pradana, Y. S., Daniyanto, Hartono, M., Prasakti, L., & Budiman, A. (2019). Effect of calcium and magnesium catalyst on pyrolysis kinetic of Indonesian sugarcane bagasse for biofuel production. Energy Procedia, 158, 431–439.

Gomes, A. C., Rodrigues, M. I., Passos, D. F., de Castro, A. M., Maria, L., Anna, M. S., & Pereira Jr., N., (2019). Acetone–butanol–ethanol fermentation from sugarcane bagasse hydrolysates: Utilization of C5 and C6 sugars. Electronic Journal of Biotechnology, 42, 16–22. Ranke, F. F. B., Shinya, T. Y., Figueiredo, F. C., Núñez, E. G. F., Cabral, H., & Neto, P. O. (2020). Ethanol from rice byproduct using amylases secreted by *Rhizopus microsporus* var. oligosporus. Enzyme partial purification and characterization. Journal of Environmental Management, 266, 110591.

Li, P., Cai, D., Luo, Z., Qin, P., Chen, C., Wang, Y., Zhang, C., Wang, Z., & Tan, T. (2016). Effect of acid pretreatment on different parts of corn stalk for second generation ethanol production. Bioresource Technology, 206, 86–92.

Kashif, M., Awan, M. B., Nawaz, S., Amjad, M., Talib, B., Farooq, M. Nizami, A. S., & Rehan, M. (2020). Untapped renewable energy potential of crop residues in Pakistan: Challenges and future directions. Journal of Environmental Management, 256, 109924.

Zhao, C., Zhang, Y., & Li, Y. (2019). Production of fuels and chemicals from renewable resources using engineered *Escherichia coli*. Biotechnology Advances, 37(7), 107402. Parascanu, M. M., Sánchez, P., Soreanu, G., Valverde, J. L., & Sanchez-Silva, L. (2019). Mexican biomasses valorization through pyrolysis process: Environmental and costs analysis. Waste Management, 95, 171–181.

Buonocore, E., Paletto, A., Russo, G. F., & Franzese, P. P. (2019).Indicators of environmental performance to assess wood-based bioenergy production: A case study in Northern Italy. Journal of Cleaner Production, 221, 242–248.

Ahmadi, L., Kannangar, M., & Bensebaa, F. (2020). Cost-effectiveness of small scale biomass supply chain and bioenergy production systems in carbon credit markets: A life cycle perspective. Sustainable Energy Technologies and Assessments, 37, 100627.

Zhang, L. Wang, Y.-Z., Zhao, T., & Xu, T. (2019). Hydrogen production from simultaneous saccharification and fermentation of lignocellulosic materials in a dual-chamber microbial electrolysis cell. International Journal of Hydrogen Energy, 44(57), 30024–30030.

Kumar, A. N., Chatterjee, S., Hemalatha, M., Althuri, A., Min, B., Kim, S.-H., & Mohan, S.V. (2020). Deoiled algal biomass derived renewable sugars for bioethanol and biopolymer production in biorefinery framework. Bioresource Technology, 296, 122315.

Fayiah, M., Dong, S., Li, Y., Xu, Y., Gao, X., Li, S., Shen, H., Xiao, J., Yang, Y., & Wesselld, K. (2019). The relationships between plant diversity, plant cover, plant biomass and soil fertility vary with grassland type on Qinghai-Tibetan Plateau. Agriculture, Ecosystems & Environment, 286, 106659.

Evolving Trends for Smart Grid Using Artificial Intelligent Techniques

International Energy Outlook 2019. (n.d.). Retrieved June 25, 2020, from https://www.eia.gov/outlooks/ieo/.

Ringel, M., & Knodt, M. (2018). The governance of the European Energy Union: Efficiency, effectiveness and acceptance of the Winter Package 2016. Energy Policy, *112*, 209–220. https://doi.org/10.1016/j.enpol.2017.09.047.

Outlook, E. (2016). About this review.

World Energy Outlook 2019 – Analysis – IEA. Retrieved June 25, 2020, from https://www.iea.org/reports/world-energy-outlook-2019.

Benysek, G. (2007). Improvement in the quality of delivery of electrical energy using power electronics systems. Springer-Verlag. https://www.springer.com/gp/book/9781846286483. ETIP SNET. (n.d.). *The European Technology and Innovation Platform Smart Networks for Energy Transition*. ETIP SNET. Retrieved June 25, 2020, from https://www.etip-snet.eu/. "GRID 2030" A National Vision for Electricity's Second 100 Years. (n.d.). Energy.gov. Retrieved June 25, 2020, from https://www.energy.gov/oe/downloads/grid-2030-national-vision-electricity-s-second-100-years.

Elias, S. (2013). *Reference Module in Earth Systems and Environmental Sciences*. Morren, J., de Haan, S. W. H., Kling, W. L., & Ferreira, J. A. (2006). Wind turbines emulating inertia and supporting primary frequency control. IEEE Transactions on Power Systems, *21*(1), 433–434. https://doi.org/10.1109/TPWRS.2005.861956.

Knopf, B., Nahmmacher, P., & Schmid, E. (2015). The European renewable energy target for 2030 – An impact assessment of the electricity sector. Energy Policy, *85*, 50–60. https://doi.org/10.1016/j.enpol.2015.05.010.

Wurster, S., & Hagemann, C. (2018). Two ways to success expansion of renewable energies in comparison between Germany's federal states. Energy Policy, *119*, 610–619. https://doi.org/10.1016/j.enpol.2018.04.059.

Akbari, H., Browne, M. C., Ortega, A., Huang, M. J., Hewitt, N. J., Norton, B., & McCormack, S. J. (2019). Efficient energy storage technologies for photovoltaic systems. Solar Energy, *192*, 144–168. https://doi.org/10.1016/j.solener.2018.03.052.

Benedek, J., Sebestyén, T.-T., & Bartók, B. (2018). Evaluation of renewable energy sources in peripheral areas and renewable energy-based rural development. Renewable and Sustainable Energy Reviews, *90*, 516–535. https://doi.org/10.1016/j.rser.2018.03.020.

Vega-Garita, V., Harsarapama, A. P., Ramirez-Elizondo, L., & Bauer, P. (2016). Physical integration of PV-battery system: Advantages, challenges, and thermal model. *2016 IEEE International Energy Conference (ENERGYCON)*, 1–6.

https://doi.org/10.1109/ENERGYCON.2016.7514038.

Bouhouras, A. S., Sgouras, K. I., Gkaidatzis, P. A., & Labridis, D. P. (2016). Optimal active and reactive nodal power requirements towards loss minimization under reverse power flow constraint defining DG type. International Journal of Electrical Power & Energy Systems, *78*, 445–454. https://doi.org/10.1016/j.ijepes.2015.12.014.

Nilsson, N. J. (1998). Artificial intelligence: A new synthesis. Morgan Kaufmann. Krishnamoorthy, C. S., Rajeev, S., & Rajeev, S. (2018). Artificial intelligence and expert systems for engineers. CRC Press. https://www.taylorfrancis.com/books/9781315137773. Shannon, C. E. (1950). XXII. Programming a computer for playing chess. The London, Edinburgh, and Dublin Philosophical Magazine and Journal of Science, *41*(314), 256–275.

https://doi.org/10.1080/14786445008521796.

Quan, X. I., & Sanderson, J. (2018). Understanding the artificial intelligence business ecosystem. IEEE Engineering Management Review, *46*(4), 22–25. https://doi.org/10.1109/EMR.2018.2882430.

Ahmad, T., & Chen, H. (2018). Utility companies strategy for short-term energy demand forecasting using machine learning based models. Sustainable Cities and Society, *39*, 401–417. https://doi.org/10.1016/j.scs.2018.03.002.

Zhang, D., Han, X., & Deng, C. (2018). Review on the research and practice of deep learning and reinforcement learning in smart grids. CSEE Journal of Power and Energy Systems, *4*(3), 362–370. https://doi.org/10.17775/CSEEJPES.2018.00520.

Bose, B. K. (1986). Power electronics and ac drives (p. 416). Englewood Cliffs, NJ: Prentice-Hall. http://adsabs.harvard.edu/abs/1986ph...book....B.

Konar, A. (2018). Artificial intelligence and soft computing: Behavioral and cognitive modeling of the human brain. CRC Press.

Fausett, L. (1994). Fundamentals of neural networks: architectures, algorithms, and applications. Prentice-Hall.

Ciabattoni, L., Ippoliti, G., Longhi, S., & Cavalletti, M. (2013). Online tuned neural networks for fuzzy supervisory control of PV-battery systems. *2013 IEEE PES Innovative Smart Grid Technologies Conference (ISGT)*, 1–6. https://doi.org/10.1109/ISGT.2013.6497901.

Rich, E., & Knight, K. (1990). Artificial intelligence (Subsequent edition). McGraw-Hill College. Senjyu, T., Hayashi, D., Yona, A., Urasaki, N., & Funabashi, T. (2007). Optimal configuration of power generating systems in isolated island with renewable energy. Renewable Energy, *32* (11), 1917–1933. https://doi.org/10.1016/j.renene.2006.09.003.

Hernández, J. C., Medina, A., & Jurado, F. (2007). Optimal allocation and sizing for profitability and voltage enhancement of PV systems on feeders. Renewable Energy. 32(10), 1768–1789. https://doi.org/10.1016/j.renene.2006.11.003.

Zadeh, L. A. (1965). Fuzzy sets. Information and Control. 8(3). 338–353.

https://doi.org/10.1016/S0019-9958(65)90241-X.

Fullér, R. (1995), Neural fuzzy systems. In Advances in soft computing series, Springer-Verlag, Berlin/Heidelberg.

Sen, Z. (1998). Fuzzy algorithm for estimation of solar irradiation from sunshine duration. Solar Energy, 63(1), 39-49. https://doi.org/10.1016/S0038-092X(98)00043-7.

Kalogirou, S. A. (2002). Design of a fuzzy single-axis sun tracking controller. https://ktisis.cut.ac.cv/handle/10488/17893.

Du, X., Cai, Y., Wang, S., & Zhang, L. (2016). Overview of deep learning. 2016 31st Youth Academic Annual Conference of Chinese Association of Automation (YAC), 159–164. https://doi.org/10.1109/YAC.2016.7804882.

Goodfellow, I., Bengio, Y., & Courville, A. (2016). Deep learning. MIT Press.

Sutton, R. S., & Barto, A. G. (n.d.). Reinforcement learning: An introduction (p. 352), MIT Press. Kalogirou, S. A. (2001). Artificial neural networks in renewable energy systems applications: A review. Renewable and Sustainable Energy Reviews, 5(4), 373–401. https://doi.org/10.1016/S1364-0321(01)00006-5.

Raza, M. O., & Khosravi, A. (2015). A review on artificial intelligence based load demand forecasting techniques for smart grid and buildings. Renewable and sustainable energy reviews, 50, 1352–1372. https://doi.org/10.1016/j.rser.2015.04.065.

Dong, X., Qian, L. & Huang, L. (2017). Short-term load forecasting in smart grid: A combined CNN and K-means clustering approach. 2017 IEEE International Conference on Big Data and *Smart Computing (BigComp)*, 119–125. https://doi.org/10.1109/BIGCOMP.2017.7881726. He, Y., Deng, J., & Li, H. (2017). Short-term power load forecasting with deep belief network and copula models, 2017 9th International Conference on Intelligent Human-Machine Systems

and Cybernetics (IHMSC), 1, 191–194. https://doi.org/10.1109/IHMSC.2017.50. Marino, D. L., Amarasinghe, K., & Manic, M. (2016). Building energy load forecasting using Deep Neural Networks, IECON 2016 – 42nd Annual Conference of the IEEE Industrial Electronics Society, 7046–7051. https://doi.org/10.1109/IECON.2016.7793413

Amarasinghe, K., Marino, D. L., & Manic, M. (2017). Deep neural networks for energy load forecasting, 2017 IEEE 26th International Symposium on Industrial Electronics (ISIE), (pp. 1483-1488). https://doi.org/10.1109/ISIE.2017.8001465.

Mocanu, D. C., Mocanu, E., Nguyen, P. H., Gibescu, M., & Liotta, A. (2016). Big IoT data mining for real-time energy disaggregation in buildings. 2016 IEEE International Conference on Systems, Man, and Cybernetics (SMC), 003765–003769.

https://doi.org/10.1109/SMC.2016.7844820.

Tornai, K., Oláh, A., Drenyovszki, R., Kovács, L., Pinté, I., & Levendovszky, J. (2017). Recurrent neural network based user classification for smart grids. 2017 IEEE Power Energy Society Innovative Smart Grid Technologies Conference (ISGT), 1–5. https://doi.org/10.1109/ISGT.2017.8086043.

Scerri, P., Pynadath, D. V., & Tambe, M. (2002). Towards adjustable autonomy for the real world. Journal of Artificial Intelligence Research, 17, 171–228. https://doi.org/10.1613/jair.1037. Zhao, Z., Xu, G., Qi, Y., Liu, N., & Zhang, T. (2016). Multi-patch deep features for power line insulator status classification from aerial images. 2016 International Joint Conference on Neural Networks (IJCNN), 3187-3194. https://doi.org/10.1109/IJCNN.2016.7727606.

Reddy, P. P., & Veloso, M. M. (2011, June 28). Strategy learning for autonomous agents in smart grid markets. Twenty-Second International Joint Conference on Artificial Intelligence. https://www.aaai.org/ocs/index.php/IJCAI/IJCAI11/paper/view/3346.

Introduction to AI Techniques for Photovoltaic Energy Conversion System

List of countries by electricity consumption

https://en.wikipedia.org/wiki/List_of_countries_by_electricity_consumption

Algore Documentary Inconvenient Truth, Documentary by Mr. Algore.

World Energy Outlook 2019, IEA, Paris https://www.iea.org

Masters G. (2013) Renewable and Efficient Electric Power Systems. WI Publication.

Patel M. R., (2006) Wind and Solar Power Systems: Design, Analysis, and Operation, Second Edition.

Mukerjee, A. K., and Nivedita Thakur. (2011). Photovoltaic Systems: Analysis and Design. PHI Learning Pvt. Ltd.

Kollimalla, S. K. & Mishra, M. K. (2013). Novel adaptive P&O MPPT algorithm for photovoltaic system considering sudden changes in weather condition. In 2013 International Conference on Clean Electrical Power (ICCEP), IEEE, pp. 653–658.

Kollimalla, S. K. & Mishra, M. K. (2014). A novel adaptive P&O MPPT algorithm considering sudden changes in the irradiance. IEEE Transactions on Energy conversion, 29(3), pp. 602–610.

Gules, R., Pacheco, J. D. P., Hey, H. L. & Imhoff, J. (2008). A maximum power point tracking system with parallel connection for PV stand-alone applications. IEEE Transactions on Industrial Electronics, 55(7), pp. 2674–2683.

Kollimalla, S. K. & Mishra, M. K. (2014). Variable perturbation size adaptive P&O MPPT algorithm for sudden changes in irradiance. IEEE Transactions on Sustainable Energy, 5(3), pp. 718–728.

Koutroulis, E. & Blaabjerg, F. (2015). Overview of maximum power point tracking techniques for photovoltaic energy production systems. Electric Power Components and Systems, 43(12), pp.1329–1351.

Subudhi, B. & Pradhan, R. (2012). A comparative study on maximum power point tracking techniques for photovoltaic power systems. IEEE Transactions on Sustainable Energy, 4(1), pp. 89–98.

Esram, T. & Chapman, P.L. (2007). Comparison of photovoltaic array maximum power point tracking techniques. IEEE Transactions on Energy Conversion, 22(2), pp. 439–449.

Salas, V., Olias, E., Barrado, A. & Lazaro, A. (2006). Review of the maximum power point tracking algorithms for stand-alone photovoltaic systems. Solar Energy Materials and Solar Cells, 90(11), pp. 1555–1578.

Elobaid, L. M., Abdelsalam, A. K. & Zakzouk, E. E. (2015). Artificial neural network-based photovoltaic maximum power point tracking techniques: A Survey. IET Renewable Power Generation, 9(8), pp.1043–1063.

Messalti, S. (2015). A new neural networks MPPT controller for PV systems. In IREC2015 the Sixth International Renewable Energy Congress, IEEE, pp. 1–6.

http://powersimtech.com/wp-content/uploads/2013/04/Tutorial-Solar-Module-physical-model.pdf Michael G. (2012). Design Calculations for Buck-Boost Converters. Texas Instruments Inc. Joshi, S., Pandya, V. & Dhandhia, A. (2017). Simulation on MPPT based solar PV standalone system. PDPU Journal of Energy & Management, November 2017, pp. 35–42.

Deep Learning-Based Fault Identification of Microgrid Transformers

Kevin Warwick, Arthur O. Ekwue & Raj Aggarwal. (1997). Artificial Intelligence Techniques in Power Systems. London: The Institution of Electrical Engineers.

G. Bretthauer, E. Handschin & W. Hoffmann. (1992). Expert systems application to power systems - State-of-the-art and future trends, IFAC Proceedings Volumes, 25(1), 463–468. I.H. Altas & J. Neyens. (2006). A fuzzy logic load-frequency controller for power systems, in: International Symposium on Mathematical Methods in Engineering, Cankaya University, Turkey, 1–10.

Biswarup Bhattacharya & Abhishek Sinha. (2017). Intelligent fault analysis in electrical power grids, in: IEEE 29th International Conference on Tools with Artificial Intelligence (ICTAI),

Boston, MA, 985–990, doi: 10.1109/ICTAI.2017.00151.

Anamika Yadav & Yajnaseni Dash. (2014). An overview of transmission line protection by artificial neural network: Fault detection, fault classification, fault location, and fault direction discrimination, Advances in Artificial Neural Systems, 2014(230382). http://dx.doi.org/10.1155/2014/230382

Nadil Amin, Aby Quaser Marowan & B. Chandra Ghosh. (2017). Application of genetic algorithm in power system optimization with multi-type FACTS, International Journal on Scientific and Research Publications, 5(5), 748–754.

Dongxia Zhang, Xiaoqing Han & Chunyu Deng. (2018). Review on the research and practice of deep learning and reinforcement learning in smart grids, CSEE Journal of Power and Energy Systems, 4(3), 362–369.

Nicholas S. Coleman, Christian Schegan & Karen N. Miu. (2015). A study of power distribution system fault classification with machine learning techniques, in: North American Power Symposium (NAPS), Charlotte, NC, 1–6, doi: 10.1109/NAPS.2015.7335264.

Zhi-Xin Yang, Xian-Bo Wang & Jian-Hua Zhong. (2016). Representational learning for fault diagnosis of wind turbine equipment: A multi-layered extreme learning machines approach, Energies, 9(379), 1–17. doi: 10.3390/en9060379.

Florian Rudin, Guo-Jie Li & Keyou Wang. (2017). An algorithm for power system fault analysis based on convolutional deep learning neural networks, International Journal of All Research Education and Scientific Methods, 5(9), 11–18.

Vidya Venkatesh. (2018). Fault classification and location identification on electrical transmission network based on machine learning method, MS thesis, Virginia Commonwealth University. https://scholarscompass.vcu.edu/etd/5582.

Shokoofeh Zare. (2018). Fault detection and diagnosis of electric drives using intelligent machine learning approaches, Electronic thesis and dissertations 7436, University of Windsor. https://scholar.uwindsor.ca/etd/7436.

Ian Goodfellow, Youshua Benigo & Aaron Courville. (2016). Deep Learning. Cambridge: MIT Press.

Sandro Skansi. (2018). Introduction to Deep Learning: From Logical Calculus to Artificial Intelligence. Cham: Springer International Publishing AG.

Y. Lecun, L. Bottou, Y. Bengio & P. Haffner. (1998). Gradient-based learning applied to document recognition, Proceedings of the IEEE, 86(11), 2278–2324.

Alex Krizhevsky, Ilya Sutskever & Geoffrey E. Hinton. (2012). ImageNet classification with deep convolutional neural networks, in: 25th International Conference on Neural Information Processing Systems – Volume 1 (NIPS'12), New York, NY, 1097–1105.

Christian Szegedy, Wei Liu, Yangqing Jia, Pierre Sermanet, Scott Reed, Dragomir Anguelov, Dumitru Erhan, Vincent Vannhoucke & Andrew Rabinovich. (2015). Going deeper with convolutions, in: IEEE Conference on Computer Vision and Pattern Recognition (CVPR), Boston, MA, 1–9, doi: 10.1109/CVPR.2015.7298594.

K. He, X. Zhang, S. Ren & Sun J. (2016). Deep residual learning for image recognition, in: IEEE Conference on Computer Vision and Pattern Recognition (CVPR), Las Vegas, NV, 770–778, doi: 10.1109/CVPR.2016.90.

https://www.mathworks.com/company/user_stories/hydro-qubec-models-wind-power-plant-performance.html

Power Quality Improvement for Grid-Integrated Renewable Energy Sources

Joseph, T., Ugalde-Loo, C. E., Liang, J., & Coventry, P. F. (2018). Asset management strategies for power electronic converters in transmission networks: Application to Hvdc and FACTS devices. IEEE Access, 6, 21084–21102

Cherian, E., Bindu, G. R., & Nair, P. S. C. (2016). Pollution impact of residential loads on distribution system and prospects of DC distribution. International Journal of Engineering Science and Technology, 19(4), 1655–1660.

Kazem, H. A. (2013). Harmonic mitigation techniques applied to power distribution networks. Advances in Power Electronics, 2013, 1–10

Grady, W. M., & Gilleskie, R. J. (1993). Harmonics and how they relate to power factor. In: Proceedings of the EPRI Power Quality Issues & Opportunities Conference (PQA'93), San Diego, CA.

Sharma, V. K. (2000). Power quality monitoring in industrial belt. In: Proceedings of the National Seminar (SASESC 2000), Agra, 103–107.

Al-Badi, A. H., Elmoudi, A., Metwally, I., Al-Wahaibi, A., Al-Ajmi, H., & Al Bulushi, M. (2011). Losses reduction in distribution transformers. In: International MultiConference of Engineers and Computer Scientists, Vol. II, Hong Kong.

Kung, P. (2018). Generator insulation-aging on-line monitoring technique based on fiber optic detecting technology. In: Sánchez, R. A. (Ed.), Simulation and Modeling of Electrical Insulation Weaknesses in Electrical Equipment. London: IntechOpen.

Owusu, P. A., & Asumadu-Sarkodie, S. (2016). A review of renewable energy sources, sustainability issues and climate change mitigation. Cogent Engineering, 3(1) https://doi.org/10.1080/23311916.2016.1167990.

Singh, M., Khadkikar, V., Chandra, A., & Varma, R. K. (2011). Grid interconnection of renewable energy sources at the distribution level with power-quality improvement features. IEEE Transaction on Power Delivery, 26(1), 307–315.

Bose, B. K. (2017). Power electronics, smart grid, and renewable energy systems. Proceedings of IEEE, 105(11), 2011–2018.

Elmoudi, A., Lehtonen, M., & Nordman, H. (2006). Effect of harmonics on transformers loss of life. In: Conference Record of the 1988 IEEE International Symposium on Electrical Insulation, 408–411.

Power Grid Corporation of India Limited. (2015). Swachh Power: A Glimpse of Power Quality in India. Gurugram: Power Grid Corporation of India.

Daut, I., Syafruddin, H. S., Rosnazri, A., & Zali, S. M. (2006). The effects of harmonic components on transformer losses of sinusoidal source supplying non-linear loads. American Journal of Applied Sciences, 3(12), 2131–2133.

El-Habrouk, M., Darwish, M., & Mehta, P. (2000). Active power filters: A review. IEE Proceedings – Electric Power Applications, 147(5), 403–413.

Stanciu, D., Teodorescu, M. A., Florescu, A., & Stoichescu. D. A. (2010). Single-phase active power filter with improved sliding mode control. In: Proceedings of the 17th IEEE International Conference on Automation, Quality and Testing, Robotics (AQTR), 15–19.

Patnaik, N., & Panda, A. K. (2014). Comparative analysis on a shunt active power filter with different control strategies for composite loads. In: Proceedings of the IEEE Region 10 Conference (TENCON 2014), 1–6.

Singh, B., Al-Haddad, K., & Chandra, A. (1999). Review of active filters for power quality improvement. IEEE Transactions on Industrial Electronics, 46(5), 960–971.

Newman, M. J., Holmes, G., Nielsen, J. G., & Blaabjerg, F. (2005). A dynamic voltage restorer (DVR) with selective harmonic compensation at medium voltage level. IEEE Transactions on Industry Applications, 41(6), 1744–1753.

Silva, S. A. O., & Campanhol, L. B. G. (2014). Application of shunt active power filter for harmonic reduction and reactive power compensation in three-phase four-wire systems. IET Power Electronics, 7(11), 2825–2836.

Fuchs, E. F., & Masoum, M. A. S. (2008). The roles of filters in power systems. In: Power Quality in Power Systems and Electrical Machines (359–395). Cambridge, MA: Academic Press.

Khadkikar, V., Chandra, A., Barry, A. O., & Nguyen, T. D. (2006). Conceptual study of unified power quality conditioner (UPQC). In: IEEE International Symposium on Industrial Electronics, Montreal, QC, 1088–1093.

Sant, A. V., Khadkikar, V., Weidong, X., & Zeineldin, H. (2013). Adaptive control of grid connected photovoltaic inverter for maximum VA utilization. In: Proceedings of the 39th Annual Conference of the IEEE Industrial Electronics Society (IECON 2013), Vienna, 388–393.

Fujita, H., & Akagi, H. (1998). The unified power quality conditioner: The integration of series active filters and shunt active filters. IEEE Transactions on Power Electronics, 13(2), 315–322. Khadkikar, V. (2012). Enhancing electric power quality using UPQC: A comprehensive overview. IEEE Transactions on Power Electronics, 27(5), 2284–2297.

Ramanaiah, M. L., & Reddy, M. D. (2018). Moth flame optimization method for unified power quality conditioner allocation. International Journal of Electrical and Computer Engineering, 8(1),

530-537.

Anjana, P., Gupta, V., & Tiwari. H. (2014). Reducing harmonics in micro grid distribution system using APF with PI controller. In: IEEE PES T&D Conference and Exposition, Chicago, IL, 1–5. Amirullah, Soeprijanto, A., Adiananda, & Penangsang, O. (2020). Power transfer analysis using UPQC-PV system under sag and interruption with variable irradiance. In: International Conference on Smart Technology and Applications (ICoSTA), Surabaya, Indonesia, 1–7. Mohammadi, H. R., Varjani, A. Y., & Mokhtari. H. (2009). Multiconverter Unified Power Conditioning System: MC-UPQC. IEEE Transaction on Power Delivery, 24(3), 1679–1686. Rahmani, S., & Al-Haddad, K. (2011). Filtering techniques for power quality improvement, IEEE Porto PowerTech, IEEE. doi:10.1201/b10643-45

Han, B., Bae, B., Kim, H., & Baek, S. (2006). Combined operation of unified power quality conditioner with distributed generation. IEEE Transaction on Power Delivery, 21(1), 330–338. Khadem, S. K., Basu, M., & Conlon, M. F. (2013). A new placement and integration method of UPQC to improve the power quality in DG network. In: 48th International Universities' Power Engineering Conference (UPEC), Dublin, 1–6.

Karelia, N., & Pandya, V. (2015). Distributed generation and its role of UPQC-DG in meeting power quality criteria. Procedia Technology, 21, 520–525.

Jindal, A. K., Ghosh, A., & Joshi. A. (2005). The protection of sensitive loads from interharmonic currents using shunt/series active filters. Electric Power System Research, 73(2), 187–196. Jindal, A. K., Ghosh, A., & Joshi, A. (2007). Interline unified power quality conditioner. IEEE Transaction on Power Delivery, 22(1), 364–372.

Karelia, N., Sant, A., & Pandya, V. (2019). Comparison of UPQC topologies for power quality enhancement in grid integrated renewable energy sources. In: IEEE 16th India Council International Conference (INDICON2019), Rajkot.

Akagi, H., Watanabe, E. H., & Aredes, M. (2007). Combined series and shunt power conditioners. In: Instantaneous Power Theory and Applications to Power Conditioning (265–374). Hoboken, NJ: John Wiley & Sons.

Jain, S. K., Agarwal, P., & Gupta, H. O. (2004). A control algorithm for compensation of customer generated harmonics and reactive power. IEEE Transactions on Power Delivery, 19(1), 357–360.

AI-Based Energy-Efficient Fault Mitigation Technique for Reliability Enhancement of Wireless Sensor Network

Abdul-Rahman, A., & Hailes, S. (2000, January). Supporting trust in virtual communities. In: *Proceedings of the 33rd Annual Hawaii International Conference on System Sciences* (9 pp.). Benini, L., Castelli, G., Macii, A., Macii, E., Poncino, M., & Scarsi, R. (2000, January). A discrete-time battery model for high-level power estimation. In: *Proceedings of the Conference on Design, Automation and Test in Europe Conference and Exhibition 2000 (Cat. No. PR00537)* (pp. 35–41).

Horst, R., Jewett, D., & Lenoski, D. (1993). The risk of data corruption in microprocessor-based systems. In: *The Twenty-Third International Symposium on Fault-Tolerant Computing (FTCS-23)* (pp. 576–585).

Siewiorek, D., & Swarz, R. (2017). Reliable Computer Systems: Design and Evaluation. Digital Press, Bedford, MA.

Barborak, M., Dahbura, A., & Malek, M. (1993). The consensus problem in fault-tolerant computing. ACM Computing Surveys, *25*(2), 171–220.

Zhao, S. Z., & Suganthan, P. N. (2011). Two-*lbests* based multi-objective particle swarm optimizer. Engineering Optimization, *43*(1), 1–17.

Dhillon, J., Parti, S. C., & Kothari, D. P. (1993). Stochastic economic emission load dispatch. Electric Power Systems Research, *26*(3), 179–186.

Syed, M., & Dubey, M. (2019, December). A novel adaptive neuro-fuzzy inference systemdifferential evolution (Anfis-DE) assisted software fault-tolerance methodology in wireless sensor network (WSN). In: *2019 International Conference on Computational Intelligence and Knowledge Economy (ICCIKE)* (pp. 736–741). Sohraby, K., Minoli, D., & Znati, T. (2007). Wireless Sensor Networks: Technology, Protocols, and Applications. John Wiley & Sons.

Dargie, W., & Poellabauer, C. (2010). Fundamentals of Wireless Sensor Networks: Theory and Practice. John Wiley & Sons.

Ni, K., Ramanathan, N., Chehade, M. N. H., Balzano, L., Nair, S., Zahedi, S., ... & Srivastava, M. (2009). Sensor network data fault types. ACM Transactions on Sensor Networks, *5*(3), 1–29. Jalote, P. (1994). Fault Tolerance in Distributed Systems. Prentice-Hall, Inc., Upper Saddle River, NJ,

Elhadef, M., Boukerche, A., & Elkadiki, H. (2008). A distributed fault identification protocol for wireless and mobile ad hoc networks. Journal of Parallel and Distributed Computing, *68*(3), 321–335.

Siewiorek, D. P., & Swarz, R. S. (1982). The Theory and Practice of Reliable System Design. Digital Press, Bedford, MA.

Gobriel, S., Khattab, S., Mossé, D., Brustoloni, J., & Melhem, R. (2006, September). Ridesharing: Fault tolerant aggregation in sensor networks using corrective actions. In: 2006 3rd Annual IEEE Communications Society on Sensor and Ad Hoc Communications and Networks (pp. 595–604).

Zhao, J., & Govindan, R. (2003, November). Understanding packet delivery performance in dense wireless sensor networks. In: *Proceedings of the 1st International Conference on Embedded Networked Sensor Systems* (pp. 1–13).

Babaie, S., Khadem-Zadeh, A., & Badie, K. (2012). Distributed fault detection method and diagnosis of fault type in clustered wireless sensor networks. Life Science Journal, *9*(4), 3410–3422.

Szewczyk, R., Osterweil, E., Polastre, J., Hamilton, M., Mainwaring, A., & Estrin, D. (2004). Habitat monitoring with sensor networks. Communications of the ACM, *47*(6), 34–40.

Ramanathan, N., Balzano, L., Burt, M., Estrin, D., Harmon, T., Harvey, C., ... & Srivastava, M. (2006). Rapid deployment with confidence: Calibration and fault detection in environmental sensor networks. UCLA: Center for Embedded Network Sensing. Available at: https://escholarship.org/uc/item/8v26b5qh

Li, H., Price, M. C., Stott, J., & Marshall, I. W. (2007, September). The development of a wireless sensor network sensing node utilising adaptive self-diagnostics. In: Hutchison, D., Katz, R. H. (eds) Self-Organizing Systems. IWSOS 2007. Lecture Notes in Computer Science, *vol 4725* (pp. 30–43). Springer, Berlin, Heidelberg.

Lee, M. H., & Choi, Y. H. (2008). Fault detection of wireless sensor networks. Computer Communications, *31*(14), 3469–3475.

Chessa, S., & Santi, P. (2002). Crash faults identification in wireless sensor networks. Computer Communications, *25*(14), 1273–1282.

Vuran, M. C., Akan, Ö. B., & Akyildiz, I. F. (2004). Spatio-temporal correlation: Theory and applications for wireless sensor networks. Computer Networks, *45*(3), 245–259.

Miao, X., Liu, K., He, Y., Papadias, D., Ma, Q., & Liu, Y. (2013). Agnostic diagnosis: Discovering silent in wireless sensor networks. IEEE Transactions on Wireless Communications, *12*(12), 6067–6075.

Luo, X., Dong, M., & Huang, Y. (2005). On distributed fault-tolerant detection in wireless sensor networks. IEEE Transactions on Computers, *55*(1), 58–70.

Guo, S., Zhong, Z., & He, T. (2009, November). FIND: faulty node detection for wireless sensor networks. In: *Proceedings of the 7th ACM Conference on Networked Sensor Systems* (pp. 253–266).

Gheorghe, L., Rughiniş, R., Deaconescu, R., & Țăpuş, N. (2010). Adaptive trust management protocol based on fault detection for wireless sensor networks. In: *Proceedings of the Second International Conferences on Advanced Service Computing* (pp. 216–221).

Koushanfar, F., Potkonjak, M., & Sangiovanni-Vincentell, A. (2002, June). Fault tolerance techniques for wireless ad hoc sensor networks. In: SENSORS, *2002 IEEE* (Vol. 2, pp. 1491–1496).

Luo, H., Wu, K., Guo, Z., Gu, L., & Ni, L. M.(2011). Ship detection with wireless sensor networks. IEEE Transactions on Parallel and Distributed Systems, 23(7), 1336–1343. Rakhamtov, D., & Vrudhula, S. (2001, August). Time-to-failure estimation for batteries in portable electronic systems. In: *Proceedings of the 2001 International Symposium on Low Power Electronics and Design* (pp. 88–91).

Wang, P., Zheng, J., & Li, C. (2007, November). An agreement-based fault detection mechanism for under water sensor networks. In: *IEEE GLOBECOM 2007-IEEE Global Telecommunications Conference* (pp. 1195–1200).

Sakib, K. (2012). "Asynchronous failed sensor node detection method for sensor networks. International Journal of Network Management, *22*(1), 27–49.

Barron, J. W., Moustapha, A. I., & Selmic, R. R. (2008, April). Real-time implementation of fault detection in wireless sensor networks using neural networks. In: *Fifth International Conference on Information Technology: New Generations (ITNG 2008)* (pp. 378–383).

O'Hare, G., O'Grady, M., & Marsh, D. (2006). Autonomic wireless sensor networks: Intelligent ubiquitous sensing. In: *Proceeding of ANIPLA 2006, International Congress on Methodologies for Emerging Technologies in Automation.*

AI Techniques Applied to Wind Energy

Fthenakis, V., and H. C. Kim. *Land use and electricity generation: A life-cycle analysis.* Renewable and Sustainable Energy Reviews, 2009. 13(6–7): pp. 1465–1474.

Leung, D. Y., and Y. Yang. *Wind energy development and its environmental impact: A review.* Renewable and Sustainable Energy Reviews, 2012. 16(1): pp. 1031–1039.

Premalatha, M., T. Abbasi, and S. A. Abbasi. *Wind energy: Increasing deployment, rising environmental concerns.* Renewable and Sustainable Energy Reviews, 2014. 31: pp. 270–288. De Souza Rabelo, D., et al. *Impedance-based structural health monitoring and statistical method for threshold-level determination applied to 2024-T3 aluminum panels under varying temperature.* Structural Health Monitoring, 2017. 16(4): pp. 365–381.

Ashley, F., et al. Bethany Wind Turbine Study Committee Report, 2007.

Hameed, Z., et al. *Condition monitoring and fault detection of wind turbines and related algorithms: A review.* Renewable and Sustainable Energy Reviews, 2009. 13(1): pp. 1–39. Germanischer Lloyd. *Wind energy, GL wind.* Possible Wind Turbine Damage, 2007. 30: p. 11. http://www.glgroup.com/industrial/glwind/3780.htm

Ciang, C.C., J.-R. Lee, and H.-J. Bang. *Structural health monitoring for a wind turbine system: A review of damage detection methods.* Measurement Science and Technology, 2008. 19(12): p. 122001.

Jureczko, M., M. Pawlak, and A. Mężyk. *Optimisation of wind turbine blades*. Journal of Materials Processing Technology, 2005. 167(2–3): pp. 463–471.

Sørensen, B.F., et al. Improved design of large wind turbine blade of fibre composites based on studies of scale effects (Phase 1). Summary report. Riso-R1390 (EN), Risø National Laboratory, Denmark, 2004.

Joosse, P., et al. *Acoustic emission monitoring of small wind turbine blades*. Journal of Solar Energy Engineering, 2002. 124(4): pp. 446–454.

Avdelidis, N., et al. Structural integrity assessment of materials by thermography. In: Conference on Damage in Composite Materials (CDCM), Stuttgart, Germany. 2006.

Raišutis, R., E. Jasiūnienė, and E. Žukauskas. *Ultrasonic NDT of wind turbine blades using guided waves*. Ultragarsas "Ultrasound", 2008. 63(1): pp. 7–11.

Schroeder, K., et al. A fibre Bragg grating sensor system monitors operational load in a wind turbine rotor blade. Measurement Science and Technology, 2006. 17(5): p. 1167.

Farrar, C. R., and K. Worden. Structural Health Monitoring: A Machine Learning Perspective. 2012. Hoboken, NJ: John Wiley & Sons.

Freudenthaler, B., et al. Case-based decision support for bridge monitoring. In: Third International Multi-Conference on Computing in the Global Information Technology (ICCGI 2008), Athens, 2008.

Worden, K., and G. Manson. *The application of machine learning to structural health monitoring*. Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences, 2007. 365(1851): pp. 515–537.

Singh, S., W. K. Seah, and B. Ng. Cluster-centric medium access control for WSNs in structural health monitoring. In: 13th International Symposium on Modeling and Optimization in Mobile, Ad Hoc, and Wireless Networks (WiOpt), Mumbai, 2015.

Stetco, A., et al. *Machine learning methods for wind turbine condition monitoring: A review.* Renewable Energy, 2019. 133: pp. 620–635.

Lapira, E., et al. *Wind turbine performance assessment using multi-regime modeling approach.* Renewable Energy, 2012. 45: pp. 86–95.

Wang, L., et al. *Wind turbine gearbox failure identification with deep neural networks*. IEEE Transactions on Industrial Informatics, 2016. 13(3): pp. 1360–1368.

Guo, P., D. Infield, and X. Yang. *Wind turbine generator condition-monitoring using temperature trend analysis.* IEEE Transactions on Sustainable Energy, 2011. 3(1): pp. 124–133.

Godwin, J. L., and P. Matthews. *Classification and detection of wind turbine pitch faults through SCADA data analysis*. International Journal of Prognostics and Health Management, 2013. 4: p. 90.

Jiang, G., et al. *Multiscale convolutional neural networks for fault diagnosis of wind turbine gearbox*. IEEE Transactions on Industrial Electronics, 2018. 66(4): pp. 3196–3207.

Leahy, K., et al. *Diagnosing and predicting wind turbine faults from SCADA data using support vector machines*. International Journal of Prognostics and Health Management, 2018. 9(1): pp. 1–11.

Yang, W., R. Court, and J. Jiang. *Wind turbine condition monitoring by the approach of SCADA data analysis.* Renewable Energy, 2013. 53: pp. 365–376.

Dao, P. B., et al. *Condition monitoring and fault detection in wind turbines based on cointegration analysis of SCADA data.* Renewable Energy, 2018. 116: pp. 107–122. Moness, M., and A. M. Moustafa. *A survey of cyber-physical advances and challenges of wind energy conversion systems: Prospects for internet of energy.* IEEE Internet of Things Journal, 2015. 3(2): pp. 134–145.

Tian, Z., et al. *Condition based maintenance optimization for wind power generation systems under continuous monitoring.* Renewable Energy, 2011. 36(5): pp. 1502–1509.

Nithya, M., S. Nagarajan, and P. Navaseelan. Fault detection of wind turbine system using neural networks. In: IEEE Technological Innovations in ICT for Agriculture and Rural Development (TIAR), Chennai, 2017.

Regan, T., et al. Wind turbine blade damage detection using various machine learning algorithms. In: International Design Engineering Technical Conferences and Computers and Information in Engineering Conference, Charlotte, NC, 2016.

Vitola, J., et al. Data-driven methodologies for structural damage detection based on machine learning applications. In: Ramakrishnan, S. (Ed.), *Pattern Recognition: Analysis and Applications*. 2016. London: IntechOpen.

Bulut, A., et al. Real-time nondestructive structural health monitoring using support vector machines and wavelets. In: Advanced Sensor Technologies for Nondestructive Evaluation and Structural Health Monitoring. 2005, 5770: pp. 180–189. International Society for Optics and Photonics.

Santos, P., et al. *An SVM-based solution for fault detection in wind turbines.* Sensors, 2015. 15(3): pp. 5627–5648.

Comparative Performance Analysis of Multi-Objective Metaheuristic Approaches for Parameter Identification of Three-Diode-Modeled Photovoltaic Cells

Yu, K., Liang, J. J., Qu, B. Y., Chen, X., & Wang, H. (2017). Parameters identification of photovoltaic models using an improved JAYA optimization algorithm. Energy Conversion and Management, *150*, 742–753.

Oliva, D., El Aziz, M. A., & Hassanien, A. E. (2017). Parameter estimation of photovoltaic cells using an improved chaotic whale optimization algorithm. Applied Energy, *200*, 141–154. Wu, L., Chen, Z., Long, C., Cheng, S., Lin, P., Chen, Y., & Chen, H. (2018). Parameter extraction of photovoltaic models from measured IV characteristics curves using a hybrid trust-region reflective algorithm. Applied Energy, *232*, 36–53.

Chellaswamy, C., & Ramesh, R. (2016). Parameter extraction of solar cell models based on adaptive differential evolution algorithm. Renewable Energy, *97*, 823–837.

Zhang, Y., Jin, Z., Zhao, X., & Yang, Q. (2020). Backtracking search algorithm with Lévy flight for estimating parameters of photovoltaic models. Energy Conversion and Management, *208*, 112615.

Jordehi, A. R. (2016). Parameter estimation of solar photovoltaic (PV) cells: A review. Renewable and Sustainable Energy Reviews, *61*, 354–371.

Gnetchejo, P. J., Essiane, S. N., Ele, P., Wamkeue, R., Wapet, D. M., & Ngoffe, S. P. (2019). Important notes on parameter estimation of solar photovoltaic cell. Energy Conversion and Management, *197*, 111870.

Abbassi, R., Abbassi, A., Jemli, M., & Chebbi, S. (2018). Identification of unknown parameters of solar cell models: A comprehensive overview of available approaches. Renewable and Sustainable Energy Reviews, *90*, 453–474.

Mehta, H. K., Warke, H., Kukadiya, K., & Panchal, A. K. (2019). Accurate Expressions for Single-Diode-Model Solar Cell Parameterization. *IEEE Journal of Photovoltaics*, *9*(3), 803–810. Qais, M. H., Hasanien, H. M., Alghuwainem, S., & Nouh, A. S. (2019). Coyote optimization algorithm for parameters extraction of three-diode photovoltaic models of photovoltaic modules. Energy, *187*, 116001.

Yousri, D., Thanikanti, S. B., Allam, D., Ramachandaramurthy, V. K., & Eteiba, M. B. (2020). Fractional chaotic ensemble particle swarm optimizer for identifying the single, double, and three diode photovoltaic models' parameters. Energy, *195*, 116979.

Elazab, O. S., Hasanien, H. M., Alsaidan, I., Abdelaziz, A. Y., & Muyeen, S. M. (2020). Parameter estimation of three diode photovoltaic model using grasshopper optimization algorithm. Energies, *13*(2), 497.

Qais, M. H., Hasanien, H. M., & Alghuwainem, S. (2020). Parameters extraction of three-diode photovoltaic model using computation and Harris Hawks optimization. Energy, *195*, 117040. Muangkote, N., Sunat, K., Chiewchanwattana, S., & Kaiwinit, S. (2019). An advanced onlooker-ranking-based adaptive differential evolution to extract the parameters of solar cell models. Renewable Energy, *134*, 1129–1147.

Jordehi, A. R. (2018). Enhanced leader particle swarm optimisation (ELPSO): An efficient algorithm for parameter estimation of photovoltaic (PV) cells and modules. Solar Energy, *159*, 78–87.

Yu, K., Qu, B., Yue, C., Ge, S., Chen, X., & Liang, J. (2019). A performance-guided JAYA algorithm for parameters identification of photovoltaic cell and module. Applied Energy, *237*, 241–257.

Qais, M. H., Hasanien, H. M., & Alghuwainem, S. (2019). Identification of electrical parameters for three-diode photovoltaic model using analytical and sunflower optimization algorithm. Applied Energy, *250*, 109–117.

Rawat, N., Thakur, P., & Jadli, U. (2019). Solar PV parameter estimation using multi-objective optimisation. Bulletin of Electrical Engineering and Informatics, *8*(4), 1198–1205.

Biswas, P. P., Suganthan, P. N., Wu, G., & Amaratunga, G. A. (2019). Parameter estimation of solar cells using datasheet information with the application of an adaptive differential evolution algorithm. Renewable Energy, *132*, 425–438.

Marler, R. T., & Arora, J. S. (2004). Survey of multi-objective optimization methods for engineering. Structural and Multidisciplinary Optimization, *26*(6), 369–395.

Branke, J., & Deb, K. (2005). Integrating user preferences into evolutionary multi-objective optimization. In: *Knowledge Incorporation in Evolutionary Computation* (pp. 461–477). Springer, Berlin, Heidelberg.

Deb, K. (2012, September). Advances in evolutionary multi-objective optimization. In: *International Symposium on Search Based Software Engineering* (pp. 1–26). Springer, Berlin, Heidelberg.

Mirjalili, S., Mirjalili, S. M., & Lewis, A. (2014). Grey wolf optimizer. Advances in Engineering Software, *69*, 46–61.

Mirjalili, S., Saremi, S., Mirjalili, S. M., & Coelho, L. D. S. (2016). Multi-objective grey wolf optimizer: A novel algorithm for multi-criterion optimization. Expert Systems with Applications, *47*, 106–119.

Saremi, S., Mirjalili, S., & Lewis, A. (2017). Grasshopper optimisation algorithm: Theory and application. Advances in Engineering Software, *105*, 30–47.

Mirjalili, S. Z., Mirjalili, S., Saremi, S., Faris, H., & Aljarah, I. (2018). Grasshopper optimization algorithm for multi-objective optimization problems. Applied Intelligence, *48*(4), 805–820.

Mirjalili, S., Gandomi, A. H., Mirjalili, S. Z., Saremi, S., Faris, H., & Mirjalili, S. M. (2017). Salp swarm algorithm: A bio-inspired optimizer for engineering design problems. Advances in Engineering Software, *114*, 163–191.

Mirjalili, S. (2016). Dragonfly algorithm: A new meta-heuristic optimization technique for solving single-objective, discrete, and multi-objective problems. Neural Computing and Applications, *27* (4), 1053–1073.

Mirjalili, S. (2015). The ant lion optimizer. Advances in Engineering Software, *83*, 80–98. Mirjalili, S., Jangir, P., & Saremi, S. (2017). Multi-objective ant lion optimizer: A multi-objective optimization algorithm for solving engineering problems. Applied Intelligence, *46*(1), 79–95.

Artificial Intelligence Techniques in Smart Grid

Bush S. F. (2014). Smart Grids. Piscataway, NJ: IEEE Press/Wiley.

Fang X., Misra S., Xue G., and Yang D. (2011). Smart grid — The new and improved power grid: A survey. IEEE Communication Surveys & Tutorials, 14(4), 944–980.

Simoes M., Roche R., Kyriakides E., Miraoui A., Blunier B., McBee K., Suryanarayanan S., Nguyen P., and Ribeiro P. (2011). Smart grid technologies and progress in Europe and the USA. In: IEEE Energy Conversion Congress and Exposition (ECCE 2011), 383–390.

Metz C. (2018). Big bets on AI open a new chip frontier for start up's too. *New York Times*, January 14.

Ferreira A., Leitao P., and Vrba P. (2014). Challenges of ICT and artificial intelligence in smart grids. In: IEEE International Workshop on Intelligent Energy System, 6–11.

Ipakchi A. and Albuyeh F. (2009). Grid of the future. IEEE Power and Energy Magazine, 7(2), 52–62.

Bose B. K. (2015). Energy environment and power electronics. In: IEEE Industrial Electronics Society Distinguished Lecture Presentation, Colorado School of Mines, Denver.

Chakraborty A. and Illic M. (2012). Power electronics and smart systems. In: Control and Optimization Methods for Electric Smart Grids. New York, NY: Springer.

Bojkovic Z. and Bakmaz B. (2012). Smart grid communications architecture: A survey and challenges. In: *Proceedings of the 11th International Conference on Applied Computer and Applied Computational Science (ACACOS)*, 83–89.

Gharavi H. and Ghafurian R. (2011). Smart grid: The electric energy system of the future. Proceedings of the IEEE, 99(6), 917–921

Bose B. K. (2017). Power electronics in smart grid and renewable energy systems.

Proceedings of the IEEE Power Electronics Magazine, 4(4), 1997–2285.

Abu-Rub H., Malinowski M., and Al Haddad K. (2014). Power Electronics for Renewable Energy Systems, Transportation and Industrial Applications. Piscataway, NJ: IEEE Press/John Wiley. Makala B. and Bakovic T. (2020). Artificial intelligence in the power sector. EMCompass No. 81. International Finance Corporation, Washington, DC. © International Finance Corporation. https://openknowledge.worldbank.org/handle/10986/34303.

Venayagamoorty G. K. (2009). Potentials and promises of computational intelligence for smart grid. In: *IEEE PES General Meeting Conference*, 1–6.

Greer C., Wollman D. A., Prochaska D. E., Boynton P. A., Mazer J. A., Nguyen C. T., Fitz Patrick G. J., Nelson T. L., Koepka G. H., Hefner Jr A. R., and Pillitteri V. Y. (2014). NIST framework and roadmap for smart grid interoperability standards. NIST Release 3.0, October 1. Bose B. K. (2019). Artificial intelligence techniques in power electronics and motor drives. In: IEEE Power Electronics and Industry Application Societies, Phoenix, Arizona, 625–675. Ramos C. and Liu C. C. (2011). Al in power systems and energy markets. IEEE Intelligent Systems, 26(2), 5–8.

Bose B. K. (2006). Power Electronics and Motor Drives: Advances and Trends. Burlington, MA: Academic Press Elsevier.

Vedder R. G. (1989). PC based expert system shells: some desirable and less desirable characteristics. Expert System, 6(1), 28–42.

Daoshen C. and Bose B. K. (1992). Expert system based automated selection of industrial ac drives. In: *IEEE/IAS Annual Meeting Conference Record*, 387–392.

Chhaya S. M. and Bose B. K. (1995). Expert system aided automated design, simulation and controller tuning of ac drive system. IEEE IECON Conference Record, 1, 712–718.

Takagi T. and Sugeno M. (1985). Fuzzy identification of a system and its applications to modeling and control. IEEE Transactions on Systems, Man, and Cybernetics, 1, 116–132. Jang S. R., Sun C. T., and Mizutani E. (1997). Neuro-Fuzzy and Soft Computing: A Computational Approach to Learning and Machine Intelligence. Upper Saddle River, NJ: Prentice Hall, 1482–1488.

Cirstea M. N., Dinu A., Khor J. G., and Cormick M. M. (2002). Neural and Fuzzy Logic Control of Drives and Power Systems. Burlington, MA: Newnes, Elsevier.

Radhakrishnan B. M., Srinivasan D., and Mehta R. (2016). Fuzzy-based multi-agent system for distributed energy management in smart grids. International Journal of Uncertainty, Fuzziness and Knowledge-Based Systems, 24(5), 781–803.

Munakata Y. and Yashvant J. (1994). Fuzzy system: An overview. *Communication of the ACM*, 37(3), 69–76.

Kariniotakis G., Pinson P., Siebert N., Giebal G., Barthelmie R. (2004). The state of the art in short term prediction of wind power - from an offshore perspective. In: *SeaTech Week - Ocean Energy Conference ADEME-IFREMER*, Brest, France

Bose B. K. (1998). A high performance inverter-fed drive system of an interior permanent magnet synchronous machine. IEEE Transactions on Industry Application, 24(6), 989–997. Xu R. I., Xu X., and Chen M. (2011). The application of genetic neural network on wind power prediction. ICICA Springer, CCIS 244, 2, 379–386.

Chang W. Y. (2014). A literature review of wind forecasting methods. Journal of Power and Energy Engineering, 2, 161–168.

Costa A., Crespo A., Navarro J., Lizcano G., Madsen H., and Feitosa E. (2008). A review on the young history of the wind power short term prediction. Renewable and Sustainable Energy Reviews, 12(6), 1725–1744.

Sideratos G. and Hatziargyriou N. D. (2007). An advanced statistical method for wind power forecasting. IEEE Transactions on Power System, 22(1), 258–265.

Lei M., Shiyan L., Chuanwen J., Hongling L., and Yan Z. (2009). A review on the forecasting of wind speed and generated power. Renewable Sustainable Energy Reviews, 13(4), 915–920. Li L., Wang M. H., Zhu F. F., and Wang C. S. (2009). Wind power forecasting based on time series and neural network. In: *Proceedings of the Second Symposium International Computer Science and Computational Technology*, 293–297.

Bose B. K. (1994). Expert system, fuzzy logic, and neural network applications in power electronics and motion control. Proceedings of the IEEE, 8(8), 1303–1323.

Principe C., Eulianoand N. R., and Lefebvre W. C. (2000). Neural and Adaptive Systems: Fundamentals through Simulations, Vol. 672. New York, NY: Wiley.

Haykin S. (1994). Neural Networks: A Comprehensive Foundation. New York, NY: Macmillan Publishing.

Bose B. K. (2007). Neural network applications in power electronics and motor drives-An introduction and perspective. IEEE Transactions on Industrial Electronics, 54(1), 14–33. Ciabattoni L., Ippoliti G., Longhi G., and Ceralietti W. M. (2013). Online tuned neural networks

for fuzzy supervisory control of PV-battery system. In: *IEEE PES Innovative Smart Grid Technologies Conference (ISGT)*, Washington, DC, 1–6.

Gomes P. and Castro R. (2012). Wind speed and wind power forecasting using statistical models: AutoRegressive moving average (ARMA) and artificial neural network (ANN). International Journal of Sustainable Energy Development. 10.20533/ijsed.2046.3707.2012.0007.

Lee J., Park G. L., Kim E. H., Kim Y., and Lee I. W. (2012). Wind speed modeling based on artificial neural networks for Jeju area. International Journal of Control and Automation, 5(2), 81–88.

Catalao J. B. S., Pousinho H. M. I., and Mendes V. M. F. (2009). An artificial neural network approach for short term wind power forecasting in Portugal. In: *International Conference on Intelligent System Applications to Power System*, 1–5.

Deshmukh M. K. and Moorthy C. B. (2010). Application of genetic algorithm to neural network model for estimation of wind power potential. Journal of Engineering, Science and Management Education, 2, 42–48.

Nair S. V., Kothari P., and Lodha K. (2014). ANN and statistical theory based forecasting and analysis of power system variables. International Journal of Emerging Technology and Advanced Engineering, 4(7), 753–758.

Bose B. K. (2007). Neural network applications in power electronics and motor drives-an introduction and perspective. IEEE Transactions on Industrial Electronics, 54(1), 12–13. Swami A., Mendel J. M., and Nikias C. L. (1998). Higher order spectral analysis toolbox. Math Works Inc. Neural Networks Toolbox User's Guide.

Catalao J. P. S., Mariano S. J. P. S., Mendes V. M. F., and Ferreira L. A. F. M. (2007). An artificial neural network approach for short-term electricity prices forecasting. In: International Conference on Intelligent Systems Applications to Power Systems, 15–23.

Zhang W. T., Xu F. Y., and Zhou L. (2010). Artificial neural network for load forecasting in smart grid. In: *International Conference Machine Learning and Cybernetics*, 3200–3206.

Bose B. K. (2017). Artificial intelligence techniques in smart grid and renewable energy systems: Some example applications. Proceedings of the IEEE, 105(11), 2262–2273. Jang J. S. R. (1993). ANFIS: Adaptive network-based fuzzy inference systems. IEEE Transactions on Systems, Man, and Cybernetics, 23(3), 665–685.

Kassa Y., Zhang J. H., Zheng D. H., and Wei D. (2016). Short term wind power prediction using ANFIS. In: *IEEE International Conference on Power and Renewable Energy*, 388–393. Catalao P. S., Pousinho H. M. I., and Mendes V. M. F. (2011). Hybrid wavelet-PSO-ANFIS approach for short-term wind power forecasting in Portugal. *IEEE Transactions on Sustainable Energy*, 2(1), 50–59.

Hmouz A., Shen J., Hmouz R., and Jun Y. (2012). Modeling and simulation of an adaptive neuro-fuzzy inference system (ANFIS) for mobile learning. IEEE Transactions on Learning Technologies, 5(3), 226–237.

Chen B., Matthews P. C., and Tavner P. J. (2013). Wind turbine pitch faults prognosis using aprior knowledge-based ANFIS. Expert Systems with Applications, 40(17), 6863–6876. Mohandes M., Rehman S., and Rahman S.M. (2012). Estimation of wind speed profile using adaptive neusssro fuzzy inference system ANFIS. Applied Energy, 88(11), 4024–4032. Catalao P. S., Pousinho H. M. I., and Mendes V. M. F. (2011). Hybrid wavelet-PSO-ANFIS approach for short-term wind power forecasting in Portugal. IEEE Transactions on Sustainable Energy, 2(1), 50–59.

Castellanos F., and James N. (2009). Average hourly wind speed forecasting with ANFIS II. In: 11th Americas Conference on Wind Engineering.

Parameter Identification of a New Reverse Two-Diode Model by Moth Flame Optimizer

Wu, L., Chen, Z., Long, C., Cheng, S., Lin, P., Chen, Y., & Chen, H. (2018). Parameter extraction of photovoltaic models from measured IV characteristics curves using a hybrid trust-region reflective algorithm. Applied Energy, *232*, 36–53.

Chellaswamy, C., & Ramesh, R. (2016). Parameter extraction of solar cell models based on adaptive differential evolution algorithm. Renewable Energy, *97*, 823–837.

Jordehi, A. R. (2016). Parameter estimation of solar photovoltaic (PV) cells: A

review. Renewable and Sustainable Energy Reviews, *61*, 354–371.

Abbassi, R., Abbassi, A., Jemli, M., & Chebbi, S. (2018). Identification of unknown parameters of solar cell models: A comprehensive overview of available approaches. Renewable and Sustainable Energy Reviews, *90*, 453–474.

Mehta, H. K., Warke, H., Kukadiya, K., & Panchal, A. K. (2019). Accurate expressions for single-diode-model solar cell parameterization. *IEEE Journal of Photovoltaics*, *9*(3), 803–810.

Qais, M. H., Hasanien, H. M., Alghuwainem, S., & Nouh, A. S. (2019). Coyote optimization algorithm for parameters extraction of three-diode photovoltaic models of photovoltaic modules. Energy, *187*, 116001.

Yousri, D., Thanikanti, S. B., Allam, D., Ramachandaramurthy, V. K., & Eteiba, M. B. (2020). Fractional chaotic ensemble particle swarm optimizer for identifying the single, double, and three diode photovoltaic models' parameters. Energy, *195*, 116979.

Nogueira, C. E. C., Bedin, J., Niedziałkoski, R. K., de Souza, S. N. M., & das Neves, J. C. M. (2015). Performance of monocrystalline and polycrystalline solar panels in a water pumping system in Brazil. Renewable and Sustainable Energy Reviews, *51*, 1610–1616.

Zhao, J., Li, Y., Yang, G., Jiang, K., Lin, H., Ade, H., ... & Yan, H. (2016). Efficient organic solar cells processed from hydrocarbon solvents. Nature Energy, *1*(2), 1–7.

Han, G., Zhang, S., Boix, P. P., Wong, L. H., Sun, L., & Lien, S. Y. (2017). Towards high efficiency thin film solar cells. Progress in Materials Science, *87*, 246–291.

Qais, M. H., Hasanien, H. M., & Alghuwainem, S. (2020). Parameters extraction of three-diode photovoltaic model using computation and Harris Hawks optimization. Energy, *195*, 117040. Jordehi, A. R. (2018). Enhanced leader particle swarm optimisation (ELPSO): An efficient algorithm for parameter estimation of photovoltaic (PV) cells and modules. Solar Energy, *159*, 78–87.

Elaziz, M. A., & Oliva, D. (2018). Parameter estimation of solar cells diode models by an improved opposition-based whale optimization algorithm. Energy Conversion and Management, *171*, 1843–1859.

Cuevas, E., Gálvez, J., & Avalos, O. (2020). Comparison of solar cells parameters estimation using several optimization algorithms. In: Recent Metaheuristics Algorithms for Parameter Identification (pp. 51–95). Springer, Cham.

Pillai, D. S., Sahoo, B., Ram, J. P., Laudani, A., Rajasekar, N., & Sudhakar, N. (2017). Modelling of organic photovoltaic cells based on an improved reverse double diode model. Energy Procedia, *117*, 1054–1061.

Muangkote, N., Sunat, K., Chiewchanwattana, S., & Kaiwinit, S. (2019). An advanced onlookerranking-based adaptive differential evolution to extract the parameters of solar cell models. Renewable Energy, *134*, 1129–1147.

Yu, K., Qu, B., Yue, C., Ge, S., Chen, X., & Liang, J. (2019). A performance-guided JAYA algorithm for parameters identification of photovoltaic cell and module. Applied Energy, *237*, 241–257.

Yu, K., Liang, J. J., Qu, B. Y., Chen, X., & Wang, H. (2017). Parameters identification of photovoltaic models using an improved JAYA optimization algorithm. Energy Conversion and Management, *150*, 742–753.

Derick, M., Rani, C., Rajesh, M., Farrag, M. E., Wang, Y., & Busawon, K. (2017). An improved optimization technique for estimation of solar photovoltaic parameters. Solar Energy, *157*, 116–124.

Beigi, A. M., & Maroosi, A. (2018). Parameter identification for solar cells and module using a hybrid firefly and pattern search algorithms. Solar Energy, *171*, 435–446.

Faramarzi, A., Heidarinejad, M., Stephens, B., & Mirjalili, S. (2020). Equilibrium optimizer: A novel optimization algorithm. Knowledge-Based Systems, *191*, 105190.

Heidari, A. A., Mirjalili, S., Faris, H., Aljarah, I., Mafarja, M., & Chen, H. (2019). Harris hawks optimization: Algorithm and applications. Future Generation Computer Systems, *97*, 849–872. Allam, D., Yousri, D. A., & Eteiba, M. B. (2016). Parameters extraction of the three diode model for the multi-crystalline solar cell/module using moth-flame optimization algorithm. Energy Conversion and Management, *123*, 535–548.

Abbassi, R., Abbassi, A., Heidari, A. A., & Mirjalili, S. (2019). An efficient salp swarm-inspired algorithm for parameters identification of photovoltaic cell models. Energy Conversion and Management, *179*, 362–372.

Biswas, P. P., Suganthan, P. N., Wu, G., & Amaratunga, G. A. (2019). Parameter estimation of solar cells using datasheet information with the application of an adaptive differential evolution algorithm. Renewable Energy, *132*, 425–438.

Mirjalili, S. (2015). Moth-flame optimization algorithm: A novel nature-inspired heuristic paradigm. Knowledge-Based Systems, *89*, 228–249.

Mirjalili, S., & Lewis, A. (2016). The whale optimization algorithm. Advances in Engineering Software, *95*, 51–67.

Mirjalili, S. (2016). SCA: A sine cosine algorithm for solving optimization problems. Knowledge-Based Systems, *96*, 120–133.

Mirjalili, S., Gandomi, A. H., Mirjalili, S. Z., Saremi, S., Faris, H., & Mirjalili, S. M. (2017). Salp swarm algorithm: A bio-inspired optimizer for engineering design problems. Advances in Engineering Software, *114*, 163–191.

Breslow, N. (1970). A generalized Kruskal-Wallis test for comparing K samples subject to unequal patterns of censorship. Biometrika, *57*(3), 579–594.

Wilcoxon, F., Katti, S. K., & Wilcox, R. A. (1970). Critical values and probability levels for the Wilcoxon rank sum test and the Wilcoxon signed rank test. In: Selected Tables in Mathematical Statistics, Vol. 1 (pp. 171–259). American Mathematical Society, Providence, RI. Hochberg, Y. (1988). A sharper Bonferroni procedure for multiple tests of significance. Biometrika, *75*(4), 800–802.

Time Series Energy Prediction and Improved Decision-Making

T. A. Napp, A. Gambhir, T. P. Hills, N. Florin, and P. S. Fennell, "A review of the technologies, economics and policy instruments for decarbonising energy-intensive manufacturing industries," Renewable and Sustainable Energy Reviews, vol. 30, pp. 616–640, 2014.

P. Thollander, M. Danestig, and P. Rohdin, "Energy policies for increased industrial energy efficiency: Evaluation of a local energy programme for manufacturing SMEs," Energy Policy, vol. 35, no. 11, pp. 5774–5783, 2007.

A. H. Kakaee, A. Paykani, and M. Ghajar, "The influence of fuel composition on the combustion and emission characteristics of natural gas fueled engines," Renewable and Sustainable Energy Reviews, vol. 38, pp. 64–78, 2014.

"India natural gas output grew 0.69% in FY19," 2020. [Online]. Available:

https://www.indoasiancommodities.com/2019/05/01/india-natural-gas-output-grew-0-69-in-fy19/ [accessed: 18 June 2020].

ET EnergyWorld, "India's natural gas production grew for second consecutive year in 2018-2019," 2020. [Online]. Available: https://energy.economictimes.indiatimes.com/news/oil-and-gas/indias-natural-gas-production-grew-for-second-consecutive-year-in-2018-2019/69108091 [accessed: 28 May 2020].

Reuters, "Global oil majors see surge in Indian demand for natural gas," 2019. [Online]. Available: https://www.reuters.com/article/us-energy-india-gas/global-oil-majors-see-surge-inindian-demand-for-natural-gas-idUSKBN1WT1QI [accessed: 18 June 2020].

K. Neuhoff and C. Von Hirschhausen, "Long-term vs. Short-term Contracts: A European Perspective on Natural Gas," Working Papers EPRG 0505, Energy Policy Research Group, Cambridge Judge Business School, University of Cambridge, 2006.

S. Makridakis, E. Spiliotis, and V. Assimakopoulos, "Statistical and Machine Learning forecasting methods: Concerns and ways forward," PLoS ONE, vol. 13, no. 3, pp. 1–26, 2018. Y. Zhang, F. Li, and B. Liu, "Research on the forecasting model of total human resource demand of large central enterprise group based on C-D production function," in: DEStech Transactions on Computer Science and Engineering, 2017. DOI: 10.12783/dtcse/itme2017/8008

D. Y. Wang and M. Lu, "Discussion on human resources management issues for SMEs — A case study of Qinzhouhuang Millet Co., Ltd.," in: Proceedings of the International Conference on Humanity and Social Science (ICHSS2016), pp. 22–27, 2017.

C. D. Ittner and J. Michels, "Risk-based forecasting and planning and management earnings forecasts," Review of Accounting Studies, vol. 22, no. 3, pp. 1005–1047, 2017.

V. A. Mabert, A. Soni, and M. A. Venkataramanan, "Enterprise resource planning: Managing the implementation process," European Journal of Operational Research, vol. 146, no. 2, pp. 302–314, 2003.

V. C. Sugiarto, R. Sarno, and D. Sunaryono, "Sales forecasting using Holt-Winters in enterprise resource planning at sales and distribution module," in: *Proceedings of the 2016 International Conference on Information and Communication Technology and Systems (ICTS 2016)*, pp. 8–13, 2017.

P. M. Catt, R. H. Barbour, and D. J. Robb, "Assessing forecast model performance in an ERP environment," Industrial Management & Data Systems, vol. 108, no. 5, pp. 677–697, 2008.
F. Gümrah, D. Katircioglu, Y. Aykan, S. Okumus, and N. Kilinçer, "Modeling of gas demand using degree-day concept: Case study for Ankara," Energy Sources, vol. 23, no. 2, pp. 101–114, 2001.

F. B. Gorucu and F. Gumrah, "Evaluation and forecasting of gas consumption by statistical analysis," Energy Sources, vol. 26, no. 3, pp. 267–276, 2004.

H. Aras and N. Aras, "Forecasting residential natural gas demand," Energy Sources, vol. 26, no. 5, pp. 463–472, 2004.

V. Ş. Ediger and S. Akar, "ARIMA forecasting of primary energy demand by fuel in Turkey," Energy Policy, vol. 35, no. 3, pp. 1701–1708, 2007.

F. Faisal, "Time series ARIMA forecasting of natural gas consumption in Bangladesh's power sector," Elixir, vol. 49, pp. 9985–9990, 2012.

M. Akpinar and N. Yumusak, "Year ahead demand forecast of city natural gas using seasonal time series methods," Energies, vol. 9, no. 9, p. 727, 2016.

M. Akpinar and N. Yumusak, "Forecasting household natural gas consumption with ARIMA model: A case study of removing cycle," in: *AICT 2013 – 7th International Conference on Application of Information and Communication Technologies, Conference Proceedings*, 2013.

C. V. Cardoso and G. L. Cruz, "Forecasting natural gas consumption using ARIMA models and artificial neural networks," IEEE Latin America Transactions, vol. 14, no. 5, pp. 2233–2238, 2016.

Z. Zhao, C. Fu, C. Wang, and C. J. Miller, "Improvement to the prediction of fuel cost distributions using ARIMA model," in: *IEEE Power and Energy Society General Meeting*, 2018.
O. A. Karabiber and G. Xydis, "Forecasting day-ahead natural gas demand in Denmark," Journal of Natural Gas Science and Engineering, vol. 76, p. 103193, 2020.

S. Chaturvedi and D. Chakrabarti, "Internal purchase process : A study for setting improvement target," International Journal of Procurement Management, vol. 8, no. 6, pp. 753–768, 2015. J. J. Sikorski, O. R. Inderwildi, M. Q. Lim, S. S. Garud, J. Neukäufer, and M. Kraft, "Enhanced procurement and production strategies for chemical plants: Utilizing real-time financial data and advanced algorithms," Industrial & Engineering Chemistry Research, vol. 58, no. 8, pp. 3072–3081, 2019.

F. Rocha, "Procurement as innovation policy and its distinguishing effects on innovative efforts of the Brazilian oil and gas suppliers," Economics of Innovation and New Technology, vol. 27, no. 9, pp. 1–20, 2017.

G. Salema and A. Buvik, "Buyer-supplier integration and logistics performance in healthcare facilities in Tanzania : The moderating effect of centralised decision control," International Journal of Procurement Management, vol. 11, no. 2, pp. 250–265, 2018.

C. Mosácula, J. P. Chaves-Ávila, and J. Reneses, "Designing natural gas network charges: A proposed methodology and critical review of the Spanish case," Utilities Policy, vol. 54, pp. 22–36, 2018.

M. Silva and P. S. Figueiredo, "Supplier selection: A proposed framework for decision making," International Journal of Procurement Management, vol. 11, no. 2, pp. 233–249, 2018.

J. Villada and Y. Olaya, "A simulation approach for analysis of short-term security of natural gas supply in Colombia," Energy Policy, vol. 53, pp. 11–26, 2013.

L. Contesse, J. C. Ferrer, and S. Maturana, "A mixed-integer programming model for gas purchase and transportation," Annals of Operations Research, vol. 139, no. 1, pp. 39–63, 2005. T. Li, M. Eremia, and M. Shahidehpour, "Interdependency of natural gas network and power system security," IEEE Transactions on Power Systems, vol. 23, no. 4, pp. 1817–1824, 2008. C. M. Correa-Posada and P. Sanchez-Martin, "Integrated power and natural gas model for energy adequacy in short-term operation," IEEE Transactions on Power Systems, vol. 30, no. 6, pp. 3347–3355, 2015.

P. F. Kaming, A. Koesmargono, and B. W. Aji, "Delay model for Engineering Procurement Construction (EPC): A case of Liquefied Natural Gas (LNG) projects in Indonesia," in: The 2nd Conference for Civil Engineering Research Networks (ConCERN-2 2018), MATEC Web Conference, vol. 270, 2019.

M. Shahrukh, R. Srinivasan, and I. A. Karimi, "Evaluating the benefits of LNG procurement through spot market purchase," in: Computer Aided Chemical Engineering, Vol. 46 (pp. 1723–1728). Elsevier B.V., Amsterdam, 2019.

V. Sillanpää, J. Liesiö, and A. Käki, "Procurement decisions over multiple periods under piecewise-linear shortage costs and fixed capacity commitments," Omega (United Kingdom),, vol. 100, p. 102207, 2021.

M. Kovačič and B. Šarler, "Genetic programming prediction of the natural gas consumption in a steel plant," Energy, vol. 66, pp. 273–284, 2014.

A. S. Weigend, Time Series Prediction. Routledge, London, 2018.

R. J. Hyndman and G. Athanasopoulos, Forecasting: Principles and Practice, 2nd edition. OTexts, Melbourne, Australia, 2018.

Liu, Y. and Tajbakhsh, S.D., 2020. Fitting ARMA Time Series Models without Identification: A Proximal Approach. arXiv preprint arXiv:2002.06777.

M. Gerolimetto, "ARIMA and SARIMA Models," Ca' Foscari University of Venice, Italy, 2010. G. Box, G. Jenkins, G. Reinsel, and G. Ljung, Time Series Analysis: Forecasting and Control. John Wiley & Sons, Hoboken, NJ, 2008.

R. Hyndman and Y. Khandakar, "Automatic time series for forecasting: The forecast package for R," Journal of Statistical Software, vol. 27, no. 3, 2007.

K. Aho, D. Derryberry, and T. Peterson, "Model selection for ecologists: The worldviews of AIC and BIC," Ecology, vol. 95, no. 3, pp. 631–636, 2014.

Ministry of Petroleum and Natural Gas, Government of India, "LPG-policies and guidelines." [Online]. Available: http://petroleum.nic.in/marketing/policies-and-guidelines/lpg-policies-and-guidelines [accessed: 19 May 2020].

"RLNG, B I D Agreement, Gail," 2016. [Online]. Available:

https://powermin.gov.in/sites/default/files/uploads/E-bid_RLNG_Agreement-for_2nd-phase.pdf

Machine Learning-Enabled Cyber Security in Smart Grids

J. M. Carrasco, L. G. Franquelo, J. T. Bialasiewicz, E. Galván, R. C. Portillo, M. A. Martín Prats, J. E. León, and N. Moreno-Alfonso, "Power-electronic "systems for the grid integration of renewable energy sources: A survey," IEEE Transaction on Industrial Electronics, vol. 53, no. 4, pp. 1002–1016, 2006.

R Gupta, S. N. Singh, and S. K. Singal, "Automation of small hydropower station," in:

International Conference on Small Hydropower - Hydro Sri Lanka, October 22–24, 2007.

L. M. Camarinha-Matos, "Collaborative smart grids – a survey on trends," Renewable and Sustainable Energy Reviews, vol. 65, pp. 283–294, 2016.

NIST, Introduction to NISTIR 7628 Guidelines for Smart Grid Cyber Security.

http://www.nist.gov/smartgrid/upload/nistir-7628_total.pdf

R. Apel, "Smart grid architecture model: Methodology and practical application," in: Workshop of Electrical Power Control Centers, 2013.

M. Miller, M. Johns, E. Sortomme, and S. Venkata, "Advanced integration of distributed energy resources" in: Power and Energy Society General Meeting, pp. 1–2, July 2012.

R. Morales Gonzalez, B. Asare-Bediako, J. Cobben, W. Kling, G. Scharrenberg, and D. Dijkstra, "Distributed energy resources for a zero-energy neighborhood," in: 3rd IEEE PES International Conf. and Exhibition on Innovative Smart Grid Technologies, pp.1–8, 2012.

A. Sanjab, W. Saad, I. Guvenc, A. Sarwat, and S. Biswas, "Smart grid security: Threats, challenges, and solutions." arXiv:1606.06992. 2016

W. Wang and Z. Lu, "Cyber security in the smart grid: Survey and challenges," Computer Networks, vol. 57, no. 7, pp. 1344–1371, 2013.

E. De Santis, A. Rizzi, and A. Sadeghian, "A learning intelligent system for classification and characterization of localized faults in smart grids," paper presented at the 2017 IEEE Congress on Evolutionary Computation (CEC), pp. 2669–2676, 2017.

T. M. Lawrence, R. T. Watson, M.-C. Boudreau, and J. Mohammadpour, "Data flow requirements for integrating smart buildings and a smart grid through model predictive control," Procedia Engineering, vol. 180, pp. 1402–1412, 2017.

M. Moretti, S. N. Djomo, H. Azadi, K. May, K. De Vos, S. Van Passel, and N. Witters, "A systematic review of environmental and economic impacts of smart grids," Renewable and Sustainable Energy Reviews, vol. 68, pp. 888–898, 2017.

T. Abdulrahman, O. Isiwekpeni, N. Surajudeen-Bakinde, and A. Otuoze, "Design, specification and implementation of a distributed home automation system," Procedia Computer Science, vol. 94, pp. 473–478, 2016.

L. Zhou, Y. Chen, K. Guo, and F. Jia, "New approach for MPPT control of photovoltaic system with mutative-scale dual-carrier chaotic search," IEEE Transactions on Power Electronics, Vol. 26, no. 4, April 2011.

J. Hieb, "Security hardened remote terminal units for SCADA networks," Electronic Theses and Dissertations, Paper 615, University of Louisville, 2008.

G. Lu, D. De, and W.-Z. Song, "Smart grid lab: A laboratory-based smart grid testbed," in: Proceedings of the IEEE Conference on Smart Grid Communications, 2010.

M. S. Thomas and J. D. Mc Donald, Power System SCADA and Smart Grids. Oxfordshire: Taylor & Francis Group, LLC; 2015

The Smart Grid Interoperability Panel - Smart Grid Cybersecurity Working Group, "Smart grid cyber security strategy and requirements," NIST IR-7628, February 2010.

S. Tan, D. De, W.-Z. Song, J. Yang, and S. K. Das, "Survey of security advances in smart grid: A data driven approach," IEEE Communications Surveys and Tutorials, vol. 19, no. 1, pp. 397–422, 2017.

S. Paul and Z. Ni, "Vulnerability analysis for simultaneous attack in smart grid security," in: Eighth Conference on Innovative Smart Grid Technologies, pp. 1–5, April 2017.

S. Ruj and A. Pal, "Analyzing cascading failures in smart grids under random and targeted attacks," paper presented at the 28th International Conference on Advanced Information Networking and Applications (AINA), pp. 226–233, 2014

X. Yu and Y. Xue, "Smart grids: A cyber-physical systems perspective," Proceedings of the IEEE, vol. 104, no. 5, pp. 1058–1070, 2016.

K. Moslehi and R. Kumar, "A reliability perspective of the smart grid," IEEE Transactions on Smart Grid, vol. 1, no. 1, pp. 57–64, 2010.

K. Wang et al., "Wireless big data computing in smart grid," IEEE Wireless Communications, vol. 24, no. 2, pp. 58–64, 2017.

I. Esnaola, S.M. Perlaza, H.V. Poor, and O. Kosut, "Maximum distortion attacks in electricity grids," IEEE Transactions on Smart Grid, vol. 7, no. 4, pp. 2007–2015, 2016.

V. Aravinthan, V. Namboodiri, S. Sunku, and W. Jewell, "Wireless AMI application and security for controlled home area networks," in: Proceedings of the IEEE Power and Energy Society General Meeting, pp. 1–8, July 2011.

A. Singh, J. Fernandes, M. Yadav, N. Patil and M. M. Charadva, "Wireless multiuser real time LCD display using short message service basedon GSM technology," International Journal of Advanced Foundation and Research in Science and Engineering, vol. 2, pp. 173–177, 2016 D. Wei et al. "Protecting smart grid automation systems against cyberattacks," IEEE Transactions on Smart Grid, vol. 2, no. 4, pp. 782–795, 2011.

Y. Xu, C. C. Liu, K. Schneider, F. Tuffner, and D. Ton, "Microgrids for service restoration to critical load in a resilient distribution system," IEEE Transactions Smart Grid, vol. 1, no. 99, 2016.

Y. Wang, M. M. Amin, J. Fu, and H. B. Moussa, "A novel data analytical approach for false data injection cyber-physical attack mitigation in smart grids," IEEE Access, vol. 5, pp. 26022–26033, 2017.

M. B. Line, I. A. Tondel, and M. G. Jaatun, "Cyber security challenges in smart grids," in: 2nd IEEE PES International Conference and Exhibition, Innovative Smart Grid Technologies (ISGT Europe), Manchester, 2011.

F. Qiu, J. Wang, C. Chen, and J. Tong, "Optimal black start resource allocation," IEEE Transactions on Power Systems, vol. 31, no. 3, pp. 2493–2494, 2016.

X. Liu, M. Shahidehpour, Z. Li, X. Liu, Y. Cao, and Z. Bie, "Microgrids for enhancing the power grid resilience in extreme conditions," IEEE Transactions on Smart Grid, vol. 8, no. 2, no. 3, pp. 589–597, 2016. http://dx.doi.org/10.1109/TSG.2016.2579999.

M. Kammerstetter, "Architecture-driven smart grid security management," in: ACM Workshop on Information Hiding and Multimedia Security, 2014.

S. Ahmed, Y. Lee, S. Hyun, and I. Koo, "Feature selection–based detection of covert cyber deception assaults in smart grid communications networks using machine learning," IEEE Access, vol. 6, pp. 27518–27529, 2018.